

The influence of social values on consumer perceptions of food risks

GIOVE, Samantha

Available from the Sheffield Hallam University Research Archive (SHURA) at:

http://shura.shu.ac.uk/17099/

A Sheffield Hallam University thesis

This thesis is protected by copyright which belongs to the author.

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author.

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given.

Please visit http://shura.shu.ac.uk/17099/ and http://shura.shu.ac.uk/information.html for further details about copyright and re-use permissions.

SHEFFIELD HALLAM UNIVERSITY LEARNING CENTRE
CITY CAMPIUS, HOWARD STREET
SHEFFIELD S1 1WB

101 624 179 8

Fines are charged at 50p per hou

REFERENCE

The Influence of Social Values on Consumer Perceptions of Food Risks

Samantha Giove

A thesis submitted in partial fulfilment of the requirements of Sheffield Hallam University for the degree of Doctor of Philosophy

November 2001



ABSTRACT

Consumers in the twenty-first century present many challenges for managers within the food industry. Consumers are everyday bombarded with choices, making decisions in respect of these involves an element of risk taking. Whilst the consequences of 'bad' choices infrequently prove to be life threatening, the values which collectively constitute a consumer's lifestyle, are potentially threatened by these day to day decisions. Consumers can both support and reinforce their lifestyle and values through selection and rejection of the products and services offered by the food industry.

This research recognises that consumers' lifestyles and values differ and investigates the implications of these differences for consumer perceptions of food risk and risk relieving / avoiding strategies for a homogeneous sample of 215 women. An eclectic approach was adopted using both quantitative and qualitative research methods.

Taylor Nelson's British based Social Value Groups as reported by MacNulty (1985) were selected as the most appropriate framework for value segmentation. A screening instrument was constructed and used to identify members of the Sustenance driven, Outer directed and Inner directed social value groups described by MacNulty (1985). The relationships between these identified social values and perceptions of food risk and risk relieving / avoiding strategies were investigated.

The findings support the view that consumers' food risk perceptions and risk avoiding strategies are influenced by their social values. Each of the three groups of consumers presented a specific portfolio of food risk perception and risk avoiding strategies. The perceptions of all three groups are important to understanding consumers' food choices. The Inner directed consumers do however present a set of values and perceptions predicted to increase. Furthermore the Inner directed values question the likely future success of traditional management and marketing strategies in the food industry.

This research is the first to provide an insight into how social values influence food risk perceptions and risk avoiding strategies. It provides a platform for beginning to develop strategies for the management of consumers' food risk perceptions. The findings identify the need for further research into Inner directed values, particularly in the context of their implications for the management of food risk perceptions in the future. Similarly findings in respect of risk avoiding strategies need to be explored further.

A significantly funded repeat of the research would be necessary to confirm the generalisability of the findings. However from a theoretical point of view it is likely that identified relationships and differences would be apparent in other sectors of the British population.

LIST OF CONTENTS

LIST OF TABLES	I
LIST OF FIGURES	IV
ACKNOWLEDGEMENTS	v
CHAPTER 1 INTRODUCTION	1
1.1 Aims and Objectives of the Research	2
1.2 Structure of the Thesis	3
1.3 Research Process	5
CHAPTER 2 LITERATURE REVIEW	6
2.1 Social Values, Variables Influencing Consumer Perceptions	6
2.2 Social Value Segmentation	8
2.3 Taylor and Nelson's Social Value Groups	12
2.3.1 Social Value Group's Elements of Social Life	15
2.3.1.1 Family	15
2.3.1.2 Work and Leisure	16
2.3.1.3 Attitudes Towards Authority	17
2.4 Consumer Perceptions	18
2.5 Risk and Risk Perception	19
2.5.1 Definitions of Risk	19
2.5.2 Risk, 'Types' and Contexts	20
2.5.3 Risk Perception Consumer Selection	22
2.6 Risk Relieving Strategies	24
2.7 Risk and Trust	25
2.8 Food Risk, Perceived Food Risk and Risk Relieving Strategies:-	27
Research and Limitations	
2.8.1 Food Risk Research	27
2.8.2 Perceived Food Risk Research	28
2.8.3 Food Risk Relieving Strategy Research	33

2.9 Summary	35
CHAPTER 3 RESEARCH METHODOLOGY AND DESIGN	37
3.1 Part One Methodological Review	37
	37
3.1.1 Research Paradigms	
3.1.2 The Qualitative Paradigm	38
3.1.2.1 Qualitative Methodology	38
3.1.2.2 Qualitative Methods	39
3.1.2.2.1 In-Depth Interviews 3.1.2.3 Strengths and Weaknesses of Qualitative Methods	39 41
3.1.3 The Quantitative Paradigm	41
	41
3.1.3.1 Quantitative Methodology3.1.3.2 Quantitative Methods	41
3.1.3.2.1 Questionnaires	42
3.1.3.2.2 Response Format - Forced Choice Using Attitude Scales	44
3.1.3.3 Strengths and Weaknesses of Quantitative Methods	45
3.1.4 Quantitative Vs. Qualitative Paradigms	46
3.1.5 Triangulation	48
3.2 Part Two Research Design and Process	50
3.2.1 Social Value Group (SVG) Screening Instrument (Part Two of the Questionnaire)	52
3.2.1.1 Design of the Screening Instrument	52
3.2.1.2 Use of the Screening Instrument to Categorise Respondents into Social Value Groups	53
3.2.2 Interview One	54
3.2.3 Results of Piloting the (SVG) Screening Instrument (Part Two of the Questionnaire) and Interview One	56
3.2.4 Food Risk Perception and Risk Relieving Strategy	57
Questionnaires (Part Three and Part Four of the Questionnaire)	
	58
3.2.5 Design of Food Risk Perception Questionnaire (Part Three of the Questionnaire)	
3.2.6 Design of Risk Relieving Strategy Questionnaire (Part Four of the Questionnaire)	59
3 2 7 Interview Two	60

3.2.8 Results of Piloting Part Three and Part Four of the	61
Questionnaire and Interview Two	
3.2.9 Main Study	62
3.2.9.1 The Sample	62
3.2.9.2 Distribution of the Questionnaire and Covering Letter	64
3.2.9.3 Questionnaire Response Rate	65
3.2.9.4 Results of the Main Study	65
3.2.10 Interview Three	66
CHAPTER 4 QUANTITATIVE ANALYSIS	68
4.1 Techniques of Analysis	69
4.1.1 Pearson Correlation	69
4.1.2 Analysis of Variance	69
4.1.3 Significance Levels	70
4.1.4 Factor Analysis	70
4.2 Analysis of the Screening Instrument (Part Two of the Questionnaire)	71
4.2.1 Pearson Correlation Results	71
4.2.1.1 Sustenance Driven Social Value Variables Significant Correlations (215 respondents)	72
4.2.1.2 Outer Directed Social Value Variables Significant Correlations	74
4.2.1.3 Inner Directed Social Value Variables Significant Correlations	77
4.2.1.4 Summary of Analysis of the Screening Instrument	80
4.2.2 Generation of Social Value Score Profiles for Homogeneous	80
Sample	
4.2.2.1 Conditions for the Identification of Dominantly Sustenance Driven, Outer Directed and Inner Directed Respondents	82
4.3 Analysis of Food Risk Perception Questionnaire (Part Three of the	85
Questionnaire)	
4.3.1 Examination of Food Risk Perception Mean Scores	85
4.3.2 One Way Analysis of Variance (Independent Groups)	87
4.3.3 Factor Analysis	90
4.3.3.1 Results of the Factor Analysis	90
4.3.3.2 Results of the Analysis of Variance for Factor Scores for the	92
Dominant Sample of 54 (18 Sustenance Driven, 18 Outer Directed and 18 Inner Directed)	

4.3.4 Investigation of the Relationship Between Social Value	93
Variables and Food Risk Perception Variables	
4.3.4.1 Investigation of the Relationship Between Social Value Group	97
Scores and Food Risk Perceptions Variables (215 respondents)	
4.4 Analysis Risk Relieving Strategies Questionnaire (Part Four of the	99
Questionnaire)	
4.4.1 Examination of Means	99
4.4.2 One Way Analysis of Variance (Independent Groups)	101
4.4.3 Factor Analysis	103
4.4.3.1 Results of the Factor Analysis	103
4.4.3.2 Results of the Analysis of Variance for Factor Scores for theDominant Sample of 54 (18 Sustenance Driven, 18 Outer Directed and 18 Inner Directed)	105
4.4.4 Investigation of the Relationship Between Social Value	106
Variables and Risk Relieving Strategy Variables	
4.4.4.1 Investigation of the Relationship Between Social Value Group	109
Scores and Risk Relieving Strategy Variables (215 respondents)	
4.5 Summary	111
CHAPTER 5 QUALITATIVE ANALYSIS	114
5.1 Analysis of Qualitative Interviews	115
5.2 Interview One Social Values and Social Life / Lifestyle	117
5.2.1 Family	117
5.2.1.1 Sustenance Driven Respondents	118
5.2.1.2 Outer Directed Respondents	119
5.2.1.3 Inner Directed Respondents	120
5.2.2 Work and Leisure	121
5.2.2.1 Sustenance Driven Respondents	121
5.2.2.2 Outer Directed Respondents	122
5.2.2.3 Inner Directed Respondents	123
5.2.3 Attitudes Towards Authority, Experts and Convention	124
Explored Through an Examination of Health	
5.2.3.1 Sustenance Driven Respondents	124
5.2.3.2 Outer Directed Respondents	125
5.2.3.3 Inner Directed Respondents	126

5.2.4 Organisational Structures (Working Environment)	127
5.2.4.1 Sustenance Driven Respondents	128
5.2.4.2 Outer Directed Respondents	129
5.2.4.3 Inner Directed Respondents	130
5.2.5 What Might Improve Their Lifestyle	131
5.2.5.1 Sustenance Driven Respondents	131
5.2.5.2 Outer Directed Respondents	132
5.2.5.3 Inner Directed Respondents	132
5.2.6 What Makes Them Feel Good About Themselves	132
5.2.6.1 Sustenance Driven Respondents	133
5.2.6.2 Outer Directed Respondents	133
5.2.6.3 Inner Directed Respondents	133
5.3 Interview Responses to Food Risk Perception Questionnaire	134
5.3.1 Technology as a Perceived Risk	135
5.3.1.1 Sustenance Driven Respondents	135
5.3.1.2 Outer Directed Respondents	136
5.3.1.3 Inner Directed Respondents	136
5.3.2 High Cost as a Risk	138
5.3.2.1 Sustenance Driven Respondents	138
5.3.2.2 Outer Directed Respondents	139
5.3.2.3 Inner Directed Respondents	140
5.3.3 Nutrition as a Risk	140
5.3.3.1 Sustenance Driven Respondents	140
5.3.3.2 Outer Directed Respondents	141
5.3.3.3 Inner Directed Respondents	141
5.3.4 Misleading Information	142
5.3.4.1 Sustenance Driven Respondents	142
5.3.4.2 Outer Directed Respondents	142
5.3.4.3 Inner Directed Respondents	143
5.3.5 Cheap / Poor Image as a Risk	143
5.3.5.1 Sustenance Driven Respondents	143
5.3.5.2 Outer Directed Respondents	144
5.3.5.3 Inner Directed Respondents	145
5.3.6 Ecological / Ethical Risks	146
5.3.6.1 Sustenance Driven Respondents	146
5.3.6.2 Outer Directed Respondents	147
5.3.6.3 Inner Directed Respondents	148

5.3.7 Unfamiliar as a Risk	149
5.3.7.1 Sustenance Driven Respondents	149
5.3.7.2 Outer Directed Respondents	150
5.3.7.3 Inner Directed Respondents	150
5.3.8 Non Organic	151
5.3.8.1 Sustenance Driven Respondents	151
5.3.8.2 Outer Directed Respondents	152
5.3.8.3 Inner Directed Respondents	152
5.4 Interview Responses to Food Risk Relieving Strategies	153
Questionnaire	
5.4.1 Societal Acceptance / Endorsement	154
5.4.1.1 Sustenance Driven Respondents	154
5.4.1.2 Outer Directed Respondents	155
5.4.1.3 Inner Directed Respondents	155
5.4.2 Low Cost	156
5.4.2.1 Sustenance Driven Respondents	156
5.4.2.2 Outer Directed Respondents	156
5.4.2.3 Inner Directed Respondents	157
5.4.3 Branding (Product Orientation / Confidence)	157
5.4.3.1 Sustenance Driven Respondents	157
5.4.3.2 Outer Directed Respondents	158
5.4.3.3 Inner Directed Respondents	159
5.4.4 High Cost / Expensive and Image Association	159
5.4.4.1 Sustenance Driven Respondents	159
5.4.4.2 Outer Directed Respondents	159
5.4.4.3 Inner Directed Respondents	160
5.4.5 Authoritarian Reassurance	160
5.4.5.1 Sustenance Driven Respondents	160
5.4.5.2 Outer Directed Respondents	161
5.4.5.3 Inner Directed Respondents	162
5.5 Qualitative Analysis Summary	163
CHAPTER 6 DISCUSSION	164
6.1 Discussion of the Research Methodology	164
6.1.1 Using Quantitative and Qualitative Methods	164
6.1.2 Methodological Limitations	165

6.1.2.1 A Convenience Sample	165
6.1.2.2 Linguistic Ambiguity	166
6.1.2.3 Selection of Parametric Tests for Quantitative Analysis	166
6.1.2.4 Social Value Group Membership	167
6.1.3 Market / Consumer Segmentation	167
6.1.3.1 Social Value Group Segmentation	168
6.2 Summary of Social Value Groups General World View	169
6.2.1 Sustenance Driven Consumers	169
6.2.2 Outer Directed Consumers	169
6.2.3 Inner Directed Consumers	169
6.3 Social Values and their Correlation with Risk Relieving Strategies	170
and Risk Perceptions	
6.3.1 Sustenance Driven Consumers	170
6.3.2 Outer Directed Consumers	171
6.3.3 Inner Directed Consumers	171
6.3.4 Summary	172
6.4 Consumer Strategies for Limiting Exposure to Perceived Risk	173
(Risk Avoiding Strategies)	
6.5 Risk Perception and Risk Avoiding Strategy Portfolios	175
6.6 Trust and Responsibility	175
6.6.1 Social Value Groups - Who do they 'trust'?	180
6.6.1.1 Sustenance Driven and Outer Directed Consumers	180
6.6.1.2 Inner Directed Consumers	180
6.7 Recommendations to the Food Industry in Light of the Research	181
Findings	
CHAPTER 7 CONCLUSION	184
7.1 Findings of the Research	184
7.2 Implications and Recommendations for Further Research	186
BIBLIOGRAPHY	188

APPENDICES

- APPENDIX 1. Consumer Perceptions of Food Risk Questionnaire and Variable Codes
- APPENDIX 2. Correlation Matrix
 (Dominant Sample of 54)
- APPENDIX 3. Correlation Matrix

 (Total Homogeneous Sample of 215)
- **APPENDIX 4.** Interview Guides
- APPENDIX 5. Social Value Score Profiles for Total Homogeneous
 Sample of 215
- APPENDIX 6. Tables to Illustrate the Relationships between Social

 Value Variables and Food Risk Perception Variables
 for the Total Homogeneous Sample of 215
- APPENDIX 7. Tables to Illustrate the Relationships between Social
 Value Variables and Risk Relieving Strategy
 Variables for the Total Homogeneous Sample of 215
- APPENDIX 8. Sample for Qualitative Interviews
- APPENDIX 9. Qualitative Interview Transcripts : Selection of Quotes

LIST OF TABLES

Table 2.1	Summary of Social Value Group Value's	14
	(Adapted from MacNulty (1985: 332-347))	
Table 2.2	Mitchell and McGoldrick (1996) Risk Relieving Strategies	33
Table 4.1	Sustenance Driven value variable "I am primarily motivated by	72
	a need to feel secure"	
Table 4.2	Sustenance Driven value variable "I am keen not to seek	72
	change"	
Table 4.3	Sustenance Driven value variable "I tend to be set in my ways"	73
Table 4.4	Sustenance Driven value variable "I believe that the breakdown	73
	of tradition causes instability"	
Table 4.5	Sustenance Driven value variable "I tend to hold onto my	73
	existing lifestyle"	•
Table 4.6	Sustenance Driven value variable "I invest relatively little time	73
	in making new contacts / friends"	
Table 4.7	Outer Directed value variable "I am motivated by the desire to	75
	have the best"	
Table 4.8	Outer Directed value variable "Winning is a priority for me"	75
Table 4.9	Outer Directed value variable "I am strongly motivated to get	75
	ahead of others"	
Table 4.10	Outer Directed value variable "I'm motivated by a desire to	75
	improve my status"	
Table 4.11	Outer Directed value variable "I take pride in keeping up	76
	appearances"	
Table 4.12	Outer Directed value variable "I tend to make sure that other	76
	people know all about my successes"	
Table 4.13	Inner Directed value variable "I am primarily motivated by a	77
	need for self discovery"	
Table 4.14	Inner Directed value variable "I am largely unconcerned about	· 78
	the opinion of me held by the world at large"	

Table 4.15	Inner Directed value variable "I would describe myself as fairly	78
	unconventional"	
Table 4.16	Inner Directed value variable "I voice my opinions even if it	78
	means disagreeing with others or being unpopular"	
Table 4.17	Inner Directed value variable "I work most effectively where	78
	there are few rules and little structure"	
Table 4.18	Inner Directed value variable "I believe groups with alternative	78
	values and lifestyles are a positive force in society"	
Table 4.19	Means and Standard Deviations for Social Group Profile Scores	82
	for Homogeneous Group of 215	
Table 4.20	Social Value Score (SVG) Profiles for Dominant Groups	84
	(Sample of (54) 18 Sustenance driven, 18 Outer directed, 18 Inner directed)	
Table 4.21	Food Risk Perception Variable Mean Scores	85
Table 4.22	Results of Analysis of Variance for Food Risk Perception	87
	Questionnaire	
Table 4.23	Summary Table: Results of the Analysis of Variance for Food	89
	Risk Perception Questionnaire. Significant Differences Only	
	(Data extracted from Table 4.22)	
Table 4.24	Nine Factor Solution for Food Risk Perception Variables	91
Table 4.25	Results of the Analysis of Variance for Factor Scores for the	92
	Dominant Sample of 54	
	(18 Sustenance driven, 18 Outer directed, 18 Inner directed)	
Table 4.26	Factor 1 Technology (man made additions to food stuffs) Risks	94
Table 4.27	Factor 2 High Cost as a Risk	94
Table 4.28	Factor 3 Nutritional Risks	95
Table 4.29	Factor 4 Cheap (Image Associated) Risks	95
Table 4.30	Factor 5 Ecological (conservationalist / environmental) Risks	96
Table 4.31	Factor6/7 Unfamiliarity as Risk	96
Table 4.32	Results of Pearson Correlation for Social Value Group (SVG)	98
	Score and Food Risk Variable Score for sample of 215	
Table 4.33	Risk Relieving Strategy Variable Mean Scores	100
Table 4.34	Results of the Analysis of Variance for food Risk Relieving	101
	Strategies	

Table 4.35	Summary of Results for the Analysis of Variance for Food Risk	102
	Relieving Strategies (Significant differences only)	
Table 4.36	Five Factor Solution for Food Risk Relieving Strategy	104
	Variables	
Table 4.37	Results of the Analysis of Variance for the Factor Scores for	105
	the Dominant Sample of 54	
	(18 Sustenance driven, 18 Outer directed, 18 Inner directed)	
Table 4.38	Factor 1 Social Acceptance / Endorsement	107
Table 4.39	Factor 2 Low Cost	107
Table 4.40	Factor 3 Branding (Product Orientation)	108
Table 4.41	Factor 4 High Cost (Image)	108
Table 4.42	Factor 5 Authoritarian Reassurance	109
Table 4.43	Results of Pearson Correlation for Social Value Group (SVG)	110
	Score and Risk Relieving Strategy Variable Score for sample of	
	215 (Analysis of the sample of 54 generated comparative results)	

LIST OF FIGURES

Figure 1

Research Process

5

ACKNOWLEDGEMENTS

I would like to express my thanks to the following people without whom the task of completing this thesis would have been considerably more difficult.

I would firstly like to thank my director of studies, Ms Lorna Daly for her support and guidance throughout the duration of my PhD. In particular, I would like to thank her for her unstinting enthusiasm and patience. I would also like to thank Ms Kate Drew for her inspiration at the outset of the research.

The support I have received from colleagues at Sheffield Hallam University has been immense. Firstly I would like to thank all 1216 members of staff who completed and returned my questionnaire, particularly those who further contributed by participating in interviews.

Secondly and on a more personal level I would like to say how much I have appreciated the sustained support of the many colleagues and friends I have, within the School of Sport and Leisure Management.

Finally, I would like to thank my family and friends for their tremendous support throughout. Particularly I would like to thank the following people, firstly Karen, Larissa and Rachel for helping me to keep calm and laugh a lot. Secondly, my sister and my mother for their support in so many ways. My sincerest thanks must go to my husband Vincenzo, who has never complained, despite the pressures I have put on him. Lastly I want to thank my beautiful son Marcello who always keeps me smiling.

CHAPTER ONE

INTRODUCTION

The past fifteen years have presented many challenges for the British Food Industry.

Understanding consumers' risk perceptions are amongst the most fundamental challenges of all.

The need to explain consumer risk perceptions has been recognised by many, including more recently the Institute of Food Research Reading, MAFF (1997), the Department of Health (1998) and the Food Standards Agency (2000) not least as a means of finding ways to alleviate the fears that consumers exhibit and which ultimately cause turbulence in food markets.

These most recent attempts at explaining food risk perceptions have several limitations. Firstly like those preceding them, they focus for the most part on food safety issues, as such they fail to demonstrate the diversity of consumer food risk perceptions. Secondly they fail to investigate the role that risk relieving strategies play in risk perception, furthermore a sequential ordering of risk perception and risk relief is assumed. Thirdly they are limited in their ability to address the different food risk perceptions of groups of consumers with different values and lifestyles.

The potential of these insights for exploring food risk perceptions can be seen in other and related academic fields. Within the fields of marketing and consumer behaviour Mitchell and Boustani (1992), Foxall and Goldsmith (1994) support the belief that consumers perceive a variety of risks associated with the purchasing and consuming of food. Consumers recognise that a risk may challenge not only their physical well being, but just as importantly their psychological, economic or spiritual well being.

Mitchell and McGoldrick (1996: 1), in the context of retailing, identify that the area of risk relieving strategies had been given "little attention" and describe it as a neglected area in need of further investigation.

From a background of Anthropology, Douglas teams up with Wildavsky (1982a, 1982b) suggesting that people do not focus on particular risks simply in order to protect health or safety. Fundamentally they suggest that consumers' choices also reflect beliefs and values. Douglas and Wildavsky (1982a) proposed that individuals choose what to fear and how much to fear it, in order to support and reinforce their way of life.

An examination of the literature highlights that despite its cost there is an increased enthusiasm within marketing and related fields for the use of value and lifestyle segmentation (Gunter and Furnham 1992). Recognising that no one strategy will serve all consumers, marketers acknowledge the diversity of consumers and the relationship between consumers personal characteristics and their perceptions and behaviour. The value and importance of value segmentation for investigating and increasing understanding of consumer perceptions and behaviour has few critics in marketing. Despite this, researchers of risk have not yet engaged in its use.

Where value and lifestyle segmentation has been used by researchers in a British context the tendency has been to use American value typologies to do so despite there being a British alternative in Taylor Nelson's Social Value Groups. Taylor Nelson's Social Value Group typologies offer an alternative that are based on a longitudinal empirical study of British consumers' values and lifestyles.

This piece of research proposes to address these gaps and limitations.

1.1 AIMS AND OBJECTIVES OF THE RESEARCH

The overall aim of the research is to investigate and clarify how social values, as identified by Taylor Nelson and reported by MacNulty (1985), influence consumer perceptions of food risks and risk relieving strategies.

More specifically the research has the following objectives:-

- 1. to critically evaluate existing theoretical and empirical research, in the areas of perception of risk in general and more specifically food risk;
- 2. to identify members of Taylor Nelson's three Social Value Groups, 'Sustenance driven', 'Outer directed' and 'Inner directed';
- 3. to investigate the food risk perceptions of 'Sustenance driven', 'Outer directed' and 'Inner directed' consumers;
- 4. to investigate the risk relieving strategies of 'Sustenance driven', 'Outer directed' and 'Inner directed' consumers;
- 5. to make recommendations to the food industry, in light of the research findings.

This research aimed to be the first to use Taylor Nelson's Social Value Group typologies, as reported by MacNulty (1985), to investigate the relationship between and differences in consumers' social values and their food risk perception and food risk avoiding strategies.

1.2 STRUCTURE OF THESIS

The thesis consists of seven chapters. Chapter One provides an introduction to the research. Chapter Two presents a critical review of existing theoretical and empirical research relating to consumers' social values and consumer perceptions of food risk. The review develops an integrated knowledge and understanding of the research area in question. Gaps in and limitations of the published literature are identified, a theoretical framework is set, boundaries are defined and fundamentally a frame of reference for identifying questions to be addressed in the proposed research is provided.

Chapter Three is in two parts. Part One is a methodological review focusing predominantly on those broad approaches, methods and techniques adopted in the research. Part Two details the research design and process for both the pilot and the main study (see Fig 1 for an overview of the Research Process). Chapters Four and Five present the results of the research. In Chapter Four, statistical analysis of consumers' responses to the survey questionnaire is used to test the relationships between respondent's social values, food risk perceptions and risk avoiding strategies and identifies differences between consumers with differing social value group membership. In Chapter Five analysis of the qualitative interviews provides an in-depth understanding of the findings of the quantitative results to include the motivational factors involved. Chapter Six presents an in-depth discussion of the findings in the context of the literature reviewed in Chapter Two. Limitations of the research are also reviewed in this chapter. Chapter Seven concludes the thesis by discussing the findings in the context of the objectives set in Chapter One and outlining a number of recommendations for further research.

1.3 RESEARCH PROCESS

(Fig 1)

CHAPTER

(Fig I)		,11/11
Pilot	Design, Construct and Administer (pilot sample of 19). Social Value Group Screening Instrument, (based on the findings of Taylor Nelson's 'Monitor' as reported by MacNulty (1985). Latter forming Part Two of the main study questionnaire	3,
The Screening Instrument	Interview One (sample same 19 as above) Qualitative interviews to validate the Screening Instrument and develop an indepth understanding of the Social Value Groups.	3, 5*
	Design Construct and Administrat (complete a chave)	
Pilot	Design, Construct and Administer (sample as above)	
Food Risk	Food Risk Questionnaire (Part Three of the Main Study Questionnaire). Food Risk Relieving Questionnaire (Part Four of the Main Study Questionnaire).	3:
Perception and Risk Relieving	Additional sections were piloted at this stage (but not taken forward into main study).	
Strategy Questionnaires	Interview Two (sample comprised 9 from the pilot sample:- 3 Inner directed, 3 Outer directed, 3 Sustenance driven)	3: 5*
	To review questionnaire responses with respect to the relationships between social values and food risk perceptions.	
To Obtain Sample for Main Study	University. 1216 returned and completed questionaires (62%). Part One of the questionnaire was used to select members of an homogenous group. 215 respondents meet the criteria for the homogeneous sample. Female, 25 - 45 years of age, in full time employment, household income >£12,000pa, no dependent children	3 3
	Analysis of results of Part Two of the Questionnaire Quantitative analysis of the screening instrument (Part Two of the questionnaire). The results of this analysis determined the categorisation of the respondents into Sustenance driven, Outer directed, Inner directed and non dominant. The analysis also provided a test of the robustness of the screening instrument.	4*
Main Study	Analysis of results of Part Three of the Questionnaire Quantitative analysis of the relationships and differences between the food risk perceptions of Sustenance driven, Outer directed and Inner directed respondents.	4*
Undertaken on the Homogeneous Sample of 215 respondents	Analysis of results of Part Four of the Questionnaire Quantitative analysis of the relationships and differences between the risk relieving strategies of Sustenance driven, Outer directed and Inner directed respondents.	4*
respondents	Interview Three (sample comprised 14 from homogeneous group, 6 Sustenance driven, 4 Inner directed, 4 Outer directed) Qualitative Interviews focused on Part Three and Part Four of the questionnaire.	5*
		met

results

CHAPTER TWO

LITERATURE REVIEW

The following chapter critically reviews both theoretical and empirical research relating to consumers' social values and consumers' perceptions of food risk. The review seeks to develop an integrated knowledge and understanding of the research area, and to generate a frame of reference for identifying the questions to be addressed in the proposed research.

Gaps and limitations in the research literature suggest that the influence of social values on the consumer perception of food risk is an area in need of further exploration, investigation and development. This is in the context of the following questions, what food risks are perceived, by whom and why, and finally how do consumers manage those risk perceptions.

The review focuses on the following key areas. Firstly the influence of social values on consumers' perceptions is examined, with an in-depth exploration of Taylor Nelson's Social Value Group typologies. The review continues with an examination of the concept of risk. Risk perceptions and risk relieving strategies are then explored as an integral part of consumer's decision making processes. This exploration is initially non specific before focusing on food.

2.1 SOCIAL VALUES, VARIABLES INFLUENCING CONSUMER PERCEPTIONS

The disciplines, 'Consumer Behaviour' and 'Marketing,' express considerable interest in using the behavioural sciences, particularly social psychology and sociology, to understand consumers. Consumers' demographic, socio-economic, behavioural,

psychographic (values and lifestyles) and geographic profiles have been examined to ascertain how they may contribute to our understanding of consumer perceptions, behaviour and decision making processes.

The intangible subjectivity of social values and attitudes means that they are open to interpretation and are much more complex to use as independent variables in investigation and statistical analysis than more tangible data such as age and gender. Social values are however increasingly recognised as fundamental to understanding consumers.

Mitchell (1983: vii) believes that values and lifestyles help to explain practical and diverse questions such as, "why we support some issues and oppose others; why some people are strong leaders and others weak; why we trust some people and are suspicious of others and why some products attract us and others don't".

"By the term 'values' one mean's the entire constellation of a person's attitudes, beliefs, opinions, hopes, fears, prejudices, needs, desires and aspirations that, taken together govern how one behaves. One's interior set of values – numerous, complex, overlapping, and contradictory though they are – finds holistic expression in a lifestyle" (Mitchell 1983: vii).

All definitions of value derive from the Latin word, 'valere', meaning 'to be worth' or 'to be strong'. Social scientists' concept of value most closely allies itself with the worth conception, a principle, standard, course of action or quality considered useful or worthwhile. Values are central to people's lives in that they relate highly to what they prize, hold in esteem and nurture, because of this importance, "values also influence behaviour" (Kahle 1983: 14). Rokeach (1973: 79) has provided one of the most influential definitions by a social scientist, "value is an enduring belief that a specific mode of conduct or end state of existence is personally or socially preferable to an opposite or converse mode of conduct or end state of existence. A value system is an enduring organisation of beliefs concerning preferable modes of conduct or end - states of existence along a continuum of relative importance."

Values are a type of social cognition. That is, values help us to know and understand our interpersonal worlds. "As primarily learned or acquired conceptions, a social cognition tells a great deal about the people that hold them" (Kahle 1983: 14).

Rokeach (1973: 76) defined values as, "desirable and enduring end states that influence both perceptions and behaviour. As desired ends, values transcend specific situations" this is supported by Kahle (1983). They provide a structural framework for existing knowledge, a basis for interpreting and incorporating new information and an abstract set of guiding principles.

There is a substantial body of theory that predicts that values shape perceptions and behaviours. The link between values and attitudes, and wants and needs has been recognised Rose et al (1994: 1504) recognised that, "social values influence the need for affiliation, which ultimately influences the relative importance attached to the display dimensions of clothing (e.g. style, brand name), versus the utilitarian properties of a garment (e.g. ease of care, durability)".

Consumers' values vary according to many factors, these include; their dominant cultural beliefs, personal experiences, integration into social organisations / institutions. Values are the result of a multitude of variables combining fundamental elements of the consumer's life. As a result they provide a base from which to develop an understanding of consumer perceptions. Any attempt to understand consumers without exploring their values would be to ignore a fundamental element of what motivates their decision making and their lifestyle.

2.2 SOCIAL VALUE SEGMENTATION

Demographic and socio-economic descriptors have been commonly used in research as a way of categorising groups of consumers. This adoption is largely as a result of their presumed stability. They are also widely and easily understood, consequently they are administered and diagnosed with relative ease (this is often believed vital, particularly for

large scale consumer surveys). Despite this Bowles (1987) highlights that an increasing dissatisfaction with demographic classifications, as the sole way of describing target audiences, increasingly exists, and that their still dominant use is due to their commonality rather that their value.

Yankelovich (1964: 83); a respected author in the field of market segmentation, stated "differences in buyer attitudes, motivations, and values are more crucial to marketing objectives than demographics". Velzhoffer and Ascheberg (1999: 47) claimed that "in everyday life the same socio-economic life conditions evidently produce unequal worlds" questioning the reliability of socio-economic segmentation, this is supported by Brown (1993).

Psychographic segmentation, broadly defined as the study of values and lifestyles, is increasingly recognised as an important tool when attempting to understand, predict and even modify consumers perceptions and behaviour. Hatton (2000: 2) claims that "today, psychographic and lifestyle segmentation based approaches, are more likely to yield segments with a unique common need, which organisations can work towards satisfying". Psychographic variables are frequently used alongside more traditional segmentation variables such as demographics in attempts to adopt a multi-segmentation approach. Psychographic variables seek to describe the characteristics of consumers such as self concept, attitudes, interests, opinions, beliefs and values that may influence the manner in which consumers react to situations, products and even advertising / communication efforts (Perri 1990).

Much literature and research supports segmentation based on consumer values (Scott and Lamont 1970, Cunningham and Crissy 1972, Vinston et al 1977a, Vinston at al 1977b, Henry 1976, Munson 1979, Atlas 1984, Leach 1987, Goldsmith et al 1997, Bainbridge 1999, Kahle et al 1999, Dolliver 2000). Its use and application however still remains limited, this is despite its potential to offer much in the way of an explanation and understanding of consumer perceptions and decision making processes.

There have been several attempts to produce psychological typologies of values and lifestyles, which cover the entire active consumer population. "These systems are

purported to yield enduring psychological constructs which define the broadest consumer populations, but which also predict idiosyncratic behaviour" (Gunter and Furnham 1992: 26).

Theories, methods and typologies of values and lifestyles used in consumer research include Riesman's 'Theory on Social Character', 'The Value and Lifestyle Survey' (VALS) / Mitchell's 'The Nine American Lifestyles', The List of Values (LOV) developed in Michigan and Taylor Nelson's 'Social Value Groups'.

Riesman (1950) presents a theory of 'social character' which has had a great impact on value and lifestyle research. The theory asserts that, in general, human beings can be grouped into three major types of social character, 'Tradition-directed'; 'Inner-directed'; and Other-directed'.

Mitchell's 'The Nine American Lifestyles' (1983), developed at Stanford Research Institute, California was strongly influenced by Riesman's work. Mitchell used a survey to analyse and systematise the values and lifestyles (VALS) of Americans in such a way as to yield insights into why people believe and act as they do. VALS attempts to provide a way of looking at the dynamics of individual and societal change. The system is foremost a conceptual scheme based on the findings of developmental psychology, but, as established in field research, the approach has practical applicability. The VALS typology comprises four groups that are sub-divided into nine lifestyles. The four groups were labelled 'Need- Driven', 'Outer- Directed', 'Inner Directed' and 'Combined Outer and Inner Directed'. The nine lifestyles were given the labels; 'Survivor', 'Sustainer', 'Belonger', 'Emulator', 'Achiever', 'I Am Me', 'Experiential', 'Societally Conscious', 'Integrated'.

The List of Values (LOV) was developed by researchers at the University of Michigan Survey Research Centre (Kahle 1983; Veroff et al 1981). LOV was developed from the theoretical base of Feather's (1975), Maslow's (1970) and Rokeach's (1973) work on values in order to assess adaptation to various roles through value fulfilment and includes nine basic values of:- 'self respect', 'sense of accomplishment', 'being well respected', 'security', 'warm relationships with others', 'sense of belonging', 'fun and enjoyment in life', 'self fulfilment' and 'excitement'. When using this system Kahle (1983)

subdivided values into two dimensions: Internal and External. This division was theoretically based on the work of Rotter (1966).

With the exception of Taylor Nelson's Social Value Groups these typologies have all been designed and developed according to the values of American consumers living in the USA. In contrast Taylor Nelson Ltd, a British based company, used empirical research to identify Social Value Groups based on the contrasting values and attitudes of British consumers.

Taylor Nelson's Social Value Groups do however bear a strong resemblance to the 'Social Characters' described by Riesman in 'The Lonely Crowd' (1950), in so far as they comprise three Social Value Groups, 'Sustenance driven', 'Inner directed' and 'Outer directed'. The characteristics of Taylor Nelson's Social Value Groups are detailed in Table 2.1.

Fundamental to all of these value typologies is Motivation theory. Acknowledged in all of these typologies is Maslow's 'Hierarchy of Needs' (Maslow 1970). As a fundamental theory of motivation, Maslow's 'Hierarchy of needs' has been extensively cited and used as a frame of theoretical reference for value segmentation.

Maslow (1970) proposed that the individual has seven levels of need, he/she is motivated by these needs and must satisfy the more basic needs before he/she can give his/her attention to those higher in the hierarchy. According to Maslow, when struggling for bare survival the individual has a narrow perspective; his/her demands are for food and shelter. With industrial wealth guaranteeing economic well-being, the individual is motivated to satisfy their need for personal expression and personal freedom. At a more developed stage of the economy, the individual can afford the luxury of a social conscience; at this point altruistic concerns come to the surface.

Using Maslow's Hierarchy of Needs, MacNulty (1985) positions Taylor Nelson's three Social Value Groups, Sustenance driven, Outer directed and Inner Directed at ascending stages of the hierarchy. However, whilst Maslow's hierarchy implies that individuals only proceed to the next stage once they have satisfied needs at the lower level, Taylor Nelson's Social Value Groups often overlap thus generating groups with a combination

of values from more than one group. Despite this, Taylor Nelson's Social Value Groups clearly share Maslow's belief in the fundamental importance of motivation in understanding consumer decision making and behaviour.

MacNulty (1985: 335) reviewing Taylor Nelson's Social Value Groups stresses the importance of motivation. He suggests that consumers in different social value groups might behave in a similar 'fashion', for example "choosing to eat less", however, fundamentally their motivation is very different. The Sustenance driven consumer for example eats less because "food is so expensive", the Outer directed consumer in contrast eats less because they "want to look good" and the Inner directed consumer eats less "to feel better". This insight underlines the importance not only of exploring the perceptions of different groups but also of exploring the motivations that lie beneath them.

Taylor Nelson's typology of Social Value Groups has already been recognised as a useful conceptual tool in understanding consumer behaviour in the fields of leisure (Gratton and Taylor 1991) and Tourism (Dalen 1989). Other areas of application have included product innovation, purchasing, branding, advertising, food consumption (Nelson 1986, Skelly and Nelson 1966, Caulkin 1987, Laurence 1989), new product development (Nelson 1986), management strategy (Tame 1993), retailing (Powderly and MacNulty 1990) and organisational structures / dynamics (MacNulty 1985).

2.3 TAYLOR NELSON'S SOCIAL VALUE GROUPS

By the beginning of the 1970's it was clear to Taylor Nelson Associates, a British based International Market Research Group, that their clients required additional information beyond traditional economic, technological and market research data. They needed to know how people behaved in the market place and work place, but also and perhaps most fundamentally what motivated them to make decisions and behave in the way they do. This information was increasingly recognised as fundamental to long term strategic planning for organisations.

In 1973, with an aim of meeting this need, Taylor Nelson Ltd started 'The Monitor', a longitudinal survey aimed at investigating the structure and dynamics of British society via 15,000 interviews. The programme was based on an annual survey of a national, stratified, random sample of the UK population. The survey contained questions relating to the choices and concerns of the respondents. Subsequent analysis of the data enabled the examination of attitudes and values of the population in terms of 36 trends derived from more than 160 variables which have been identified in British society (MacNulty 1985). One objective was that the Monitor system would help companies respond to changes in social values taking place across Britain at any given period of time.

The results of the monitor indicate that the British population holds values and attitudes which divide into seven groups. These groups were given the labels; 'Aimless'; 'Survivors'; 'Belongers'; 'Conspicuous Consumers'; 'Social Resisters', 'Experimentalists'; 'Self Explorer', these in turn were aggregated into three major classifications labelled as: 'Sustenance driven consumers'; 'Outer directed consumers' and 'Inner directed consumers' (MacNulty 1985). Fundamentally members of every socio-economic level were found within each Social Value Group. After examining the results of the Monitor, Laurence (1989: 123) observed that "there is almost no correlation between social value groups and the social class groups traditionally used in market research. The social value groups are equally represented in each of the A,B,C1,C2,D,E categories". MacNulty (1985) described each of the three Social Value Groups by highlighting their values and characteristic traits. These are detailed in Table 2.1.

Table 2.1 Summary of Social Value Group Value's (Adapted from MacNulty(1985:

Sustenance Driven

- Clannish

332-347))

- Motivated by the need for security
- Cling to an existing lifestyle, set in their ways, resistant to change
- Have a great respect for tradition
- Concerned to maintain the status quo, looking for tradition and stability
- Perceive change as a threat
- Lead fearful and constrained lives, risk aversive
- Class conscious
- Relatively limited in the scope of their behaviour
- Tend to have somewhat narrow horizons both mentally and physically

Outer Directed

- Motivated by the search for esteem and status
- The criteria by which they measure their success are external to themselves
- They want to be seen to live in the "right part of town", drive the "right car", etc.
- Concerned about their appearance and position
- Materialistic
- Seek to improve their position in financial and social terms
- Motivated by acquisition, competition and getting ahead, being / having the best
- Traditional
- Pro-authority, support law and order
- Pushy
- Require social mobility to enjoy their lifestyles

Inner Directed

- Motivated by self realisation / actualisation
- Individualistic
- Largely unconcerned about the opinion of them held by the world at large
- Their criteria for success and the standards of their behaviour are within themselves
- They usually have a broad horizon,
- A good understanding of the world's events
- Altruistic
- Concerned with social issues such as peace, ecology, social responsibility, and democratic processes which include everyone
- Likely to be concerned with spiritual values, but in a more or less orthodox way
- Caring (motives are fairness and a good quality of life at the societal level)
- Autonomous behaviour

MacNulty (1985) illustrated the contrasting values of the members of the Social Value Groups further by examining fundamental elements of their social life, these included the family, work, leisure and attitudes towards authority.

2.3.1 SOCIAL VALUE GROUP'S ELEMENTS OF SOCIAL LIFE

2.3.1.1 FAMILY

MacNulty (1985) identified Sustenance driven consumers as

"conservative, traditional, conventional, pragmatic and devoted to his family, which he places above other things and for which he makes substantial self sacrifice" (: 333).

"He tends to be against change and his interest in the future is to ensure his family's security" (: 333).

MacNulty suggested that because

"horizons are constrained by orientation to the family, one also tends to be worried, pedantic and rule following" (: 333).

They believe the family unit supports their need for security and stability. Resistant to change he describes them as "clannish" (: 332). In contrast the *Outer directed* consumer's priority is that the family be "a credit to them" (: 333).

"They are interested in their family's appearance and behaviour..." (: 333).

MacNulty (1985) suggested that Inner directed consumers are

"highly individualistic" (: 334,342)

"autonomous behaviour will be a principal factor. Individual family members will tend to develop their own interests and friends outside the rather loosely defined family context. Friendship and common interests will be considered more important than (or certainly equally important as) blood relationships" (: 342) "They reject traditional authority and established procedures, and are a powerful force for individual autonomy and the use of new relationships" (: 334)

2.3.1.2 WORK AND LEISURE

well as leisure.

MacNulty (1985) suggested that Sustenance driven consumers

"are motivated by the need for security" (: 332)

they were frequently economically disadvantaged but even when this was not the case the Sustenance driven consumer would still remain thrifty. There is a suggestion that they perceive work as a necessity and leisure very much a luxury. MacNulty (1985) suggests that the Sustenance driven consumer will commonly,

"follow the Protestant work ethic and are deeply concerned about unemployment" (: 334)

"motive is to get along and their attitudes are those which we call 'working class values'"(: 332)

"They believe that society should provide employment" (: 336).

"They associate their employment with the notion of producing goods" (: 336).

Outer directed consumers are described by MacNulty as

"materialistic, pushy and motivated by seeking to improve their position in financial and social terms" (: 333).

This being the case, both work and leisure time would be fundamental to their self identity. In sharp contrast MacNulty (1987) suggested that the *Inner directed* consumers had a very different work/ leisure ethic, emphasising the attitudes of

"caring, autonomous behaviour and self realisation(: 334)"

"individualistic, empathetic, tolerant and non-exploitative" (: 342)

Work perhaps a mechanism through which they can self express and self realise rather than just economic / material necessity or status.

"The economy will be designed to support a society which places value on......work as a means of personal fulfilment (rather than for economic necessity), and the creative use of leisure time for individual growth" (: 341)

More emphasis is placed on the experience and less on status. Leisure emphasises self development and individual growth which may take the form of educational activities as

2.3.1.3 ATTITUDES TOWARDS AUTHORITY

MacNulty (1985) suggested that Sustenance driven consumers are

"conservative, traditional and conventional" (: 333)

MacNulty suggested that we might expect this group

"to be relatively limited in the scope of their behaviour" (: 333).

"Alternative courses of action and imaginative responses to new situations will be difficult" (: 333).

Outer directed consumers are described as respecting and trusting experts and authority but being motivated by

"status and acquisition" (: 333).

MacNulty (1985) indicates that *Inner directed* consumers exhibit autonomous behaviour. They are further described as "less orthodox"(: 334), as having "little trust for traditional authority and importantly, very likely to be concerned with spiritual values".

MacNulty (1985) described the Inner directed consumers as

"has broad horizons" (: 333).

"...likely to reject doctrinaire solutions in favour of the moral holistic view that change is an organic process" (: 334).

"broadest horizons, the highest tolerances and the largest propensity to solve problems on a global scale. They also have a tendency to hold spiritual values" (: 334).

MacNulty (1985) suggests that up until the 1970's the two classifications Sustenance driven and Outer directed represented a fair picture of British society. Believing that the major institutions in UK society are based on this kind of model, MacNulty (1985) presumes that there were members of the Inner directed group about, but that they were probably so few as to be statistically undetectable. That situation has now changed and the Inner directed group is regarded as being of significant importance. Taylor Nelson's findings revealed the increasing existence of 'Inner directed' consumers. In fact Inner directedness was identified as the set of social values set most likely to be dominant in the twenty first century.

The social values of the Inner directed group can also be identified in the work of Giddens (1991). Giddens (1991: 109) describes a 'universe of high modernity', in which many of the values described as 'Inner directed' by MacNulty (1985) are identified. The highly reflexive Inner directed consumer is evident in Giddens (1991) work, actively 'reskilling' i.e. reacquiring both knowledge and skills. For these individuals the 'reflexive project of the self generates programmes of actualisation and mastery, typically aspirations of the Inner directed consumer.

2.4 CONSUMER PERCEPTIONS

As stated earlier in this review, Rose et al (1994) reports that there is a substantial body of theory that predicts that values shape perceptions. Consumers perceptions in turn are fundamental to understanding consumers' attitudes and behaviour.

However, despite this Foxall and Goldsmith (1994: 50) suggest that

"as with so many of the terms encountered in the social sciences, perception is

used rather vaguely in everyday discourse, but must be understood more

precisely if it is to be useful in explaining aspects of human behaviour".

Harrell (1986: 7029) understands perception to mean "the process of recognising, selecting, organising, and interpreting stimuli, in order to make sense of the world around us". "Perception of goods and services depend in part, on the stimuli to which consumers are exposed, and in part on the ways these stimuli are given meaning by consumers". Foxall and Goldsmith (1994: 50) proposed that "Our different perceptions of products can account for different attitudes and behaviours towards products".

Foxall and Goldsmith (1994: 51) suggest that

"two facets of perception are of special interest. First, people become aware of their environment through the five senses (see, hear, touch, taste, smell) and therefore sensation is the process with which perception begins. Equally important, and this is the second facet, is the process of interpretation which

depends on the socio-psychological meaning the individual attaches to the object perceived"

The perceptual process of food choice is complex, personal and highly selective. Foxall and Goldsmith (1994: 50) recognise this process as selective and suggest that "consumers pay attention to and interpret stimuli that reinforce and enhance their views of their world, of themselves and the goods and services that they buy". This understanding of the way in which consumers construct and reinforce 'world views' through the selective perception of risk is discussed and explored further, later in this chapter.

2.5 RISK AND RISK PERCEPTION

2.5.1 **DEFINITIONS OF RISK**

A number of authors have highlighted the centrality of risk to modern life. Lupton (1999: 3) argues that "In contemporary western societies, where control over one's life has become increasingly viewed as important, the concept of 'risk' is now widely used to explain deviations from the norm, misfortune and frightening events"

Giddens (1991: 3) proposed that risk is a fundamental element in modern society. "The concept of risk becomes fundamental to the way both lay actors and technical specialists organise their social world". "To live in the universe of high modernity is to live in an environment of chance and risk". Giddens (1991: 324) highlights a characteristic of high modernity as plurality of choice, he suggests that, "to act in and to engage with a world of plural choices is to opt for alternatives. This system inevitably presents an increased number of decisions to be made and consequently potential risks".

Numerous attempts have been made to define risk. Beck (1992: 21) defines risk as "a systematic way of dealing with hazards and insecurities induced and introduced by modernisation itself". Beck sees 'risk society' as a catastrophic society, he contends that

this factor has political potential since "averting and managing these can include a reorganisation of power and authority".

Other authors including Short (1984: 712) suggest that "risk is the probability of some future event" similarly Lowrance (1976: 7027) defined risk as, "the likelihood, or probability, of some adverse effect of a hazard". Interestingly Hohenemser *et al* (1983: 378) suggested that "hazards are a special kind of environmental event that pose threats to humans and to the things that humans value".

2.5.2 RISK, 'TYPES' AND CONTEXTS

From the broadest of contexts, it is fair to suggest that when a consumer perceives that something they value is threatened they perceive risk. This is supported by Kates and Kasperson (1983: 7027) when they suggest that "hazards are threats to people and what they value and risks are measures of hazards". Risk is about perceived loss which may fall into any of the categories, "'physical', 'psychological', 'social', 'finance' or 'performance'" (Kaplan *et al* 1974: 289). Mitchell and Boustani (1992: 17) suggest that there are "several different types of risk which might affect consumers' purchasing habits. These include physical risk, financial risk, social risk, psychological risk and time risk". Foxall and Goldsmith (1994: 57) present 'performance /functional', 'financial', 'physical', 'psychological', 'social' and 'time' as types of perceived risk.

Consumer perception of risk has been investigated and reported extensively in the academic literature. Contributions to the understanding of risk perception have been informed by theoretical insights from one or more paradigms including, Sociology (Giddens 1990, 1991, Beck 1992, Lupton 1999); Anthropology (Douglas and Wildavsky 1982a 1982b); Psychology (Weinstein 1983, 1984, 1987, McKenna 1993, Hoorens and Bunk 1993, Langer 1975); Political Science (Wildavsky and Dake 1990, Dake 1991); Information and Decision Sciences (Slovic 1993, Curley 1992); Consumer Behaviour (Bauer 1967).

Areas of application have included:- nuclear power, industrial chemicals, natural disasters, medical and biological intervention and health (Slovic, et al 1979, Sparks et al 1992, Baxter 1990). Within this body of work there is a dominant interest in technological hazards and physical risk, on the whole these are analysed objectively with an aim to their quantification. By and large the 'victims' of such risks appear to be largely observed as an homogeneous 'public'. Such an approach for the most part ignores the fundamental need to recognise differences amongst the masses.

Decision theorists define 'risk' as the situation where a decision maker has a prior knowledge of both the consequences of alternatives and their probabilities of occurrence. The concept of perceived risk used by consumer researchers, however, bears a closer relationship to the concept of partial ignorance (where neither the consequences of alternatives nor their probabilities of occurrence are accurately known). In this view partial ignorance may led to dread or/and hysteria (Slovic et al 1982, 1979) on one hand or optimistic bias (Weinstein et al 1988) on the other.

Bauer (1960: 389), proposed consumer behaviour as an, "instance of risk taking", he suggested that:

"consumer behaviour involves risk in the sense that, any action of a consumer will produce consequences which he cannot anticipate with anything approximating certainty, and some of which at least, are likely to be unpleasant" (Bauer 1960: 390).

Understanding perceived risk is at the heart of understanding consumers' behaviour, Bauer (1967), strongly emphasised that he, and consumer behaviour research, is concerned only with the subjective (perceived) risk and not 'real world' or 'objective risk'. This approach is particularly appropriate for the investigation of food risk perceptions, where an intimate and deeply personal relationship exists between the consumer and food choice. Mitchell (1998) suggests that perceived risk is not only present in the highly visible food scares, but also motivates and helps to explain consumers' daily and weekly food purchasing. Consumer perceptions of food risks influence consumer intention and behaviour, regardless of whether the risk is 'real' and

justified, or not. The importance of understanding why consumers perceive what they perceive cannot be understated.

2.5.3 RISK PERCEPTION CONSUMER SELECTION

Risk perception is increasingly recognised as an integral part of a consumer's social world. In this context Mary Douglas, a British anthropologist, and Aarron Wildavsky, an American political scientist (1982a, 1982b) examined the question: 'Why do people emphasise some risks while ignoring others?' They focused on technological risks such as the perils of nuclear waste and carcinogenic chemicals. Douglas and Wildavsky believe that societies selectively choose particular risks for attention. They suggest that societies have their own distinctive portfolio of risks, equally they suggest that each society institutionalises means for controlling / managing some risks and not others.

Douglas and Wildavsky (1982a, 1982b) claim that people do not focus on particular risks simply in order to protect health, safety or the environment. "Their choice also reflects their beliefs, values, social institutions, nature, and moral behaviour. Risks are exaggerated or minimised according to the social, cultural, and moral acceptability of the underlying activity" (Douglas and Wildavsky (1982b: 12). They believed that what societies chose to call 'risky' is largely determined by social and cultural factors, not nature.

Viewing individuals as the active organisers of their own perceptions, Wildavsky and Dake (1990, 1991), as other cultural theorists have, proposed that individuals choose what to fear and how much to fear it, in order to support their way of life. In this perspective,

"selective attention to risk, and preferences among different types of risk taking or avoiding, correspond to cultural biases, that is, to world views or ideologies entailing deeply held values and beliefs defending different patterns of social relations. Social relations are defined in their theory in terms of a small number of distinctive patterns of relationships:- 'hierarchical', 'hermit', 'fatalist', 'egalitarian', or 'individualist'" (Wildavsky and Dake 1990: 44,45)

Combinations of cultural biases and social relations are referred to in this theory as 'ways of life'. More specifically, then, 'hierarchical', 'hermit', 'fatalist', 'egalitarian', or 'individualist' forms of social relations, together with the cultural biases that justify them, are each hypothesised to engender distinctive representations of what constitutes a hazard and what does not. Among all possible risks, those selected for worry or dismissal are functional, in the sense that they strengthen one of these 'ways of life', and weaken the others. Wildavsky and Dake (1990: 44) suggest that "since cultural biases are forms of ideology, there should be high correlations between certain biases and corresponding ideologies".

Accordingly, from the perspective of cultural theory, risk is not an objective reality; instead, "the perception of risk is a social process" (Douglas and Wildavsky 1982a: 50, 1982b). Each culture, each set of shared values and supporting social institutions, is biased toward highlighting certain risks and downplaying others.

Douglas and Wildavsky focus largely on differences between societies and not on the differences between groups within those societies, Wildavsky and Dake (1990) however designed a questionnaire to focus on differences between groups within society and more specifically within American society. From a cultural theory perspective, and in relation to risks associated with, social deviance, nature and technology Wildavsky and Dake (1990) and Dake (1991) have successfully used this approach, as a basis for understanding risk perception within the United States of America. Crucially as part of this research Wildavsky and Dake (1990) tested five risk perception theories on their ability to predict and explain what kinds of people will perceive which sorts of hazards, and how dangerous they might be. The theories examined were: 'knowledge theory'; 'personality theory', 'economic theory'; 'political theory' and 'cultural theory'. Their results confirmed that whilst there is value in all the five theories, cultural theory provides the best predictions of a broad range of perceived risk.

Wildavsky and Dake (1990) present enlightening and inspirational research. However, there are questions as to whether or not this approach can readily be transposed, to investigate consumer perceptions of food risks in a British society. These concerns are detailed later in section 2.82.

2.6 RISK RELIEVING STRATEGIES

Consumers are frequently presented with situations where they are required to make assessments, judgements and ultimately decisions. Giddens (1991) suggests that members of today's society will have an increasing number of decisions to make.

Consumers will inevitably adopt strategies aimed at effectively managing their exposure to potential risks. Bauer (1960: 390) suggests that

"consumers characteristically develop decision strategies and ways of reducing risk, this enables them to act with relative confidence and ease in situations where their information is inadequate and the consequences of their actions are in some meaningful sense incalculable".

Knowledge of consumer's perceived risks and effective risk relieving strategy is particularly valuable in the food industry, where perceived risk can lead at worst to the failure of a product / service, and at best to the decline in sales or a damaged brand / company reputation.

"Buyers often face the dilemma of wanting to purchase a product, and yet they hesitate to buy, because it involves taking the risk of suffering some type of loss", (Roselius 1971: 56). When a buyer perceives risk in a purchase situation Roselius (1971) believes he / she will pursue one of four different strategies of risk resolution,

"reduce perceived risk by either decreasing the probability that the purchase will fail, or by reducing the severity of real or imagined loss suffered if the purchase does fail; shift from one type of loss to one for which he has tolerance; postpone the purchase, in which case he would be shifting from one general risk type to another; make the purchase and absorb the unresolved risk" (Roselius 1971: 56).

Mitchell and McGoldrick (1996: 1-3) as a result of an intensive review of the published research into consumers' risk reduction strategies, note that much research has focused on the concept of risk and its measurement. They suggest that both practitioners and academic researchers should show equal or greater interest in "how consumers 'handle'

risk and the strategies they employ to reduce it". They believe that this area has been given "little attention" and describe it as "neglected".

The orientation of published research into risk reliving strategies is on the whole one dimensional. As the words 'risk relieving / reducing strategy' suggest the assumption all too often being made in the literature is that a strategy is developed as the consequence of a risk being perceived. Furthermore, the literature on risk relieving strategies totally ignores the area of social value segmentation.

2.7 RISK AND TRUST

Trust has been recognised by many authors as a fundamental issue when we are talking about risk and risk relief (O'Riordan 1982, 1983, Wildavsky 1979, Wildavsky and Dake 1990, Slovic 1993, Short 1984, Frewer et al 1995, Frewer 2000, Fife – Shaw 1995, Giddens 1991).

Giddens (1991: 3) believes that risk and trust are inextricably linked. He suggests that "in circumstances of uncertainty and multiple choice the notions of trust and risk have particular application". He suggests that "trust presumes a leap to commitment, a quality of 'faith' related to absence in time, space as well as ignorance".

Distrust can both result in or be the result of perceived risk. Understanding in whom and what consumers trust is vital if organisations are to be successful in the design, development and marketing of goods and services. Equally trust is fundamental to effective communication.

Barber (1983: 7) suggests that "trust has much to do with expectation". He selects three kinds of expectation that involve some of the fundamental meaning of trust. The most general is "expectation of the persistence and fulfilment of the natural and the moral social order. Second is expectation of technically competent role performance from those involved with us in social relationships and systems. And the third is expectation

that partners in interaction will carry out their duties in certain situations to place others interests before their own".

In its most general sense, trust means the expectations which all humans in society internalise, that the natural order both physical and biological and the moral social order will persist and be more or less realised.

Trust has been most vividly described by Luhman (1980: 4)

"Trust, in the broadest sense of confidence in one's expectations, is a basic fact of social life. In many situations, of course, man can choose in certain respects whether or not to bestow trust. But a complete absence of trust would prevent him even from getting up in the morning. He would be prey to a vague sense of dread, to paralysing fears. He would not even be capable of formulating distrust and making that a basis for precautionary measures, since this would presuppose trust in other directions. Anything and everything would be possible. Such abrupt confrontation with the complexity of the world at its most extreme is beyond human endurance".

Trust appears to be linked to perceptions of accuracy, knowledge and concern with public welfare. Distrust is associated with perception of deliberate distortion of information, bias and having been proven wrong in the past. Sources that are perceived to be over-accountable, or protecting a vested interest, are unlikely to be trusted.

Wildavsky and Dake (1990: 55) highlight the integral role of trust in risk perceptions. They propose that "it is not only that the facts cannot by themselves convince doubters, but that behind one set of facts are always others relating to whether business and government can be trusted".

Currently in the UK, trust in the government's ability to manage risk appears to be at an all time low. The usefulness of value segmentation in developing a more in-depth understanding of this phenomenon needs to be explored in the broader context to include risk relieving strategies.

2.8 FOOD RISK, PERCEIVED FOOD RISK AND RISK RELIEVING STRATEGIES: - RESEARCH AND LIMITATIONS

2.8.1 FOOD RISK RESEARCH

Much of the research in the area of food risk perceptions has focused on public resistance to new technologies and / or the impact of food scares on food markets. The focus has been on 'real', highly visible and physical food risks such as Salmonella in eggs, Listeria in pate and certain soft cheeses, Bovine Spongiform Encephalopathy (BSE) in 'meat products', Escherichia coli (E.coli) in beef, Alar in apples, benzene in mineral water and other pathogens.

Consumer perceptions of risk in relation to food products and processes has had devastating effects on sales and marketing and on consumer confidence in the food industry in recent years. In the food products arena, BSE is the model exemplar of the impact of risk perception. At the time of the BSE outbreak evidence presented and communicated by the government denied any link between BSE and new variant CJD (Creutzfeldt Jakob Disease). Despite all of these claims, sales of beef slumped. On the 19th of December 1995 'The Guardian' newspaper reported that figures from the Meat and Livestock Commission suggested that sales of beef were down 15%. Bovril were forced to deny that its beef drink uses materials from British cows and more than 1,150 schools deleted beef from lunch menus (Arthur and Wilkie 1995). The handling of the BSE crisis resulted in low levels of consumer trust in both the Government and its agencies. This reaction forced the government to pay greater attention to the importance of risk communication strategies and over time the crisis has precipitated the creation of the Food Standards Agency.

Food Irradiation is a key example of a process which is also perceived as high risk.

Extensive research prior to the legalisation of Food Irradiation, included three National

Surveys conducted by the Consumers Association (1987), the British Marketing Research Bureau (1986) and Marplan (1987). Research into the commercial feasibility of the process has also been undertaken. Findings of this research indicate that at best only 20 per cent of consumers in the UK would be willing to buy Irradiated food. Since Irradiation of food became lawful on the 1st January 1991, only one application has been made for an irradiation license. It was approved. Many manufactures and retailers of food, have recognised the negative consumer perception of irradiated food and consequently have avoided the use of the technology, despite its recognised advantages in the food safety arena.

Bord and O'Conner cited in and supported by Wildavsky and Dake (1990: 54) propose that "knowledge is inversely related to fear of technology: "having accurate knowledge about the food irradiation process translates to greater acceptance". They do add however, that fundamentally "it is not knowledge per se, but confidence in institutions and the credibility of the information that is the issue".

2.8.2 PERCEIVED FOOD RISK RESEARCH

An investigation of consumer perceptions of food risk requires an acknowledgement of the broad scope and complexity of issues and consequent perspectives relating to food.

A diverse and comprehensive range of factors are acknowledged as influencing food preferences / choices. These include socio economic, educational, cultural, religious and regional, biological, physiological and psychological factors. Personal factors include familiarity, influence of other persons and meanings attached to foods. Intrinsic factors such as the organoleptic attributes of food and extrinsic factors to include advertising and merchandising. Authors such as Shepherd and Sparks (1994) acknowledge these factors as critical when attempting to develop an understanding of consumer food choices. Similarly marketers acknowledge such factors, though presented as demographic, socio-economic, geographic, behavioural and psychographic variables, all are acknowledged as potentially critical to understanding the food choices consumers

make. Similarly, and most importantly for this research, these factors influence the rejection of food by consumers.

Conner (1993: 28) recognises food choice as a complex human behaviour and suggests that "most attitudes towards food are derived from socially transmitted information". Similarly Wright (2001: 348) suggests that "taste in food reflects, in part, the consumers social and cultural origins and social ambitions".

Literature which examines the social significance of food is clearly longstanding. Murcott (1983: 1) suggested however that "the social significance of food and eating has been left to social anthropologists, social historians, social nutritionists so-called, and other social commentators". Murcott (1983: 46) suggests that "detailed sociological investigation of popular belief systems in the face of the British 'scares' is lacking". Whilst it has been suggested that distrust of food is widespread Murcott (1983) identified that attention has tended to focus on the role of the mass media in heightening public anxiety (Beardsworth 1990; Gofton, 1990; Smith, 1991 cited in Murcott 1983). In a similar vain Murcott (1998) whilst examining the area of food choice, food scares and health, as previously, focuses on the role of the media.

Murcott (1983: 1) also points out that "sociologists, especially in Britain, have paid virtually no sustained attention" to the significance of food. Mennell (1992: 1) likewise recognised the seemingly limited attention "to food and eating as topics of serious intellectual interest". Whilst Murcott (1983, 1998), Mennell et al (1992) do go some way to address the gap, none of this is explicit to consumers perceptions of food risk and / or risk relieving strategies or consumers social values as a perspective for understanding such perceptions.

Mary Douglas, however, presents an exception, acknowledging food risk perceptions as reflections of a society's values and attitudes. A close examination of her perspective is developed in this chapter.

Whilst acknowledging the broad scope of perspectives, this literature review focuses on social values as a perspective for food risk perceptions and risk relieving strategies. A clear justification is made for this throughout this chapter.

Research into the consumer perceptions of food risks still remains limited in both scope and approach. As with research into consumer perception of risk in general, food risk research has all too often simply described different topical risks and then reported consumer views and reactions in relation to these. On the whole such research, in seeking to explain these views has failed to recognise the influence of values and lifestyles on consumer perception. This method of reporting fails to provide the food industry with the information needed to understand why consumers perceive food risks in the ways they do. If the food industry understands why some consumers perceive one food product/process as a risk and another not, strategies can be developed that aim to manage those perceptions effectively.

Research by Ford and Rennie (1987) is one example of research which simply investigates consumer views in an undifferentiated way. They investigated consumer understanding of food irradiation in 1987. The results of the survey suggested that consumers are concerned about the application of radiation to foodstuffs and in particular the potential adverse affects on health. Seventy per cent of the sample said they were not willing to purchase irradiated food, amounting to an outright rejection of the process by the vast majority of shoppers. As with the majority of this type of research, the results are expressed as the opinion of the population as a whole, thus failing to acknowledge the differences between the kind and severity of risk perceived by different groups of respondents. Ford and Rennie also make no attempt to recommend how the results of their survey may be used.

More ambitious food risk research maps a wide range of food risk perceptions using factor and cluster analysis. Mutsaers and Shepherd (1994) conducted a study using both qualitative and quantitative methodologies. They aimed to investigate which risk characteristics are important to consumers in judging food poisoning hazards in relation to other food hazards, and to investigate which consumer characteristics are crucial in risk perception of food poisoning. They concluded that

"the dimensions 'knowledge' and 'concern' were the most important risk characteristics, incorporating concerns about 'seriousness', 'probability of occurrence', 'delayed effects', 'impact on future generations', 'dread',

'voluntariness', 'controllability', 'clarity of benefits', 'familiarity'". (Mutsaers and Shepherd (1994: 362))

These findings mirrored those of Slovic et al (1987). Mutsaers and Shepherd recommended that knowledge and concern should be included in risk communication messages. The number of people under six years old in a household was the consumer characteristic that they believed to play a significant role in the perception of food poisoning risk. Whilst Mutsaers and Shepherd quantitatively illustrated the differences between food risks on the basis of the amount of knowledge and concern people held in relation to them, they did not differentiate between consumers, neither did they investigate why consumers have these perceptions.

There is very little food risk research that investigates how social values may influence consumer perception of food risks. British researchers Frewer et al (1994) did however modify Wildavsky and Dake's 'cultural biases questionnaire' to investigate how they may inform attitudes toward the use of biotechnology in the production of food. This research, administered by the Institute of Food Research Reading was conducted as part of a project funded by the Ministry of Agriculture, Fisheries and Food in the UK. This level of support is an illustration and an indication of the level of importance placed upon this area of research by both leading academics and industry.

In using the work of Wildavsky and Dake (1990, 1991) to inform their research, Frewer et al (1994) do highlight the value of using value segmentation. In investigating food risk their research is an example of the kind of research needed in this area, it is however subject to criticism. The research is limited in its scope, like a large majority of risk research the dominant interest is technological risk.

A criticism of using Wildavsky and Dake's 'cultural biases questionnaire' (1990) in the context of research into British consumers is firstly that it has its origins based on American society, to assume that the 'cultural biases' described, exist in similar form and context in Britain is to ignore the cultural differences between British and American societies. Secondly, Wildavsky and Dake concentrate on technological and environmental risks, however, many food risks do not fall into these two categories and

furthermore many food risks are intrinsically subjective as a consequence of the unique intimate relationship that exists between the consumer and his/her food. After careful consideration of the literature a third concern has emerged. It may be argued that the 'cultural biases' described by Wildavsky and Dake are a reflection of more fundamental and deeply rooted attitudes and values. These attitudes and values in turn may manifest themselves in ways other than 'cultural biases' and more specifically in ways that are much more appropriate to developing an understanding of British consumer perceptions of food risk. Finally and perhaps most important is the existence of Taylor Nelson's Social Value Groups as an alternative. Taylor Nelson's 'Monitor' presents empirical results which identify social value groups in Britain "The groups exist empirically, they are not invented by the researchers" (Laurence 1989: 123). In addition these groups have already been used successfully to investigate consumer behaviour in a number of areas to include food.

Furthermore Nelson (1986) whilst reporting the findings of 'The Monitor' has demonstrated the usefulness of using the social value groups identified as a framework for new food product development strategies. She goes on to suggest how social value group influences purchase criteria, specifically indicating the importance of branding and advertising in these processes.

Goldsmith et al (1997: 353) similarly recognise the impact of social values on food related attitudes albeit in the USA. They state that "social values represent desired end states of being or desired behaviours such as self respect, warm relations with others, or excitement. These abstract social cognitions help shape product attitudes and through them guide overt behaviour". They suggest that in the past empirical research has supported the theoretical structure of values - attitudes - behaviour, but only for a relatively few high involvement products. Their research concluded that social values were related to the consumption of snack foods independent of the effects of demographic variables, thus supporting the general thrust of Taylor Nelson's findings in the UK.

2.8.3 FOOD RISK RELIEVING STRATEGY RESEARCH

Mitchell and McGoldrick (1996) reviewed the literature in the area of 'Consumer risk reduction strategies. This valuable piece of work concluded that this area of research had been "neglected". Mitchell and McGoldrick (1996) identify two generic approaches to risk reduction. "One is to increase the certainty that the purchase will not fail, the other is to reduce the consequences of it failing". They suggest that by far the most common approach is to increase certainty. Mitchell and McGoldrick (1996) during their extensive review identified thirty seven risk relieving strategies from the literature. Many of these were not consumer orientated and therefore not within the scope of this research. Ten strategies were described as accounting for over 60 per cent of the strategies used in studies to date.

Table 2.2 details fifteen strategies identified by Mitchell and McGoldrick as pertinent for food retailing.

Table 2.2 Mitchell and McGoldrick (1996) Risk Relieving Strategies

Family and friends word of	Price Information	Store Image (recognised as
mouth	·	important when purchasing
		non food)
Well known brands	Packaging (information on)	Warranties/ money back
		guarantees
TV commercials	Trials and free samples	Shopping around
Print advertisements	Salesperson's advice	Brand loyalty
Consumer reports	Past experience	Retailer visits

(Adapted from Mitchell and McGoldrick (1996))

Roselius (1971) in a much cited piece of research, identified eleven methods for reducing / relieving different types of perceived risk, these mirrored those strategies identified and synthesised by Mitchell and McGoldrick (1996). Consumers' preferences for the eleven different methods of reducing the perceived risk were investigated via a quantitative questionnaire, the methods investigated included:- 'endorsements', 'brand loyalty', 'private

testing', 'store image', 'free sample', 'money back guarantee', 'government testing', 'shopping around', 'expensive model', 'word of mouth'. Derbaix (1983) later added sales person advice to this list. Brand and store loyalty have been the most thoroughly investigated of these strategies (Hisrich et al 1972, Cunningham 1964, Taylor and Rao 1982) and established as amongst the most effective strategies for relieving perceived risk. Word of mouth has also been recognised and investigated as a very effective risk relieving strategy (Arndt 1967, Perry & Hamm 1969).

Roselius (1971) suggests that the degree of relief anticipated from each of the relievers is influenced by the kind of loss involved and the type of consumer. He identified four types of loss as:- time, hazard, ego, money, and categorised consumers as:- high perceivers of risk, medium perceivers of risk or low perceivers of risk.

One interesting piece of research in the area of risk relieving / reducing strategy specific to food was undertaken by Mitchell and Boustani (1992). Mitchell and Boustani's research used both qualitative and quantitative methodologies. The results of their study suggested that there are five different types of risk; physical, financial, social, psychological time which affect consumers' purchasing habits. Using Roselius's risk relieving / reducing strategies, Mitchell and Boustani focused on breakfast cereals to illustrate that all products are potentially perceived to some degree as 'risky' and as a consequence consumers hunt out methods of relieving or reducing that perceived risk.

Although authors in this area of risk relieving / reducing strategies implicitly recognise that preferred risk relieving / reducing strategy is strongly influenced by the buyer's social values and attitudes, this has not been investigated explicitly or in any depth. This is an area that this research intends to explore in more depth.

2.9 SUMMARY

Consumer perception of food risk is now an area of research recognised by the Government and leading academics as important, but lacking both in terms of the quality and the quantity of research undertaken to date.

Many surveys have used demographic segmentation in their attempts to investigate consumer perceptions of food risk. Simply reporting generalised public perceptions, such surveys fail to analyse the reasons for these perceptions and as a result strategies aimed at managing consumer perceptions of food risk are all too often ineffective and inefficient.

Value and lifestyle variables are increasingly recognised as a valuable means by which to increase our understanding of consumers. Yet research in this area is limited in both scope and depth. To date only one attempt has been made in Britain to investigate how social values and related attitudes influence food risk perceptions. This research by Frewer et al (1994) used a modified version of Wildavsky and Dake's 'cultural biases questionnaire' (1990), (devised originally for use on American consumers) to investigate UK consumer perceptions of the risks associated with biotechnology.

In the UK, over a period of twenty five years Taylor Nelson Ltd, a British based market research company, have continuously monitored the social values and attitudes held by British consumers. Analysis of the information obtained from this twenty five year longitudinal survey, has revealed the existence in British society of seven distinct Social Value Groups, which were aggregated by Taylor Nelson into three major classifications: 'Sustenance driven consumers', 'Outer directed consumers' and 'Inner directed consumers'. Members of these Social Value Groups were found to hold distinctive sets of values, beliefs and motivations which underpin everything they do. These values and beliefs have been described in the literature.

This typology of Social Value Groups ('Sustenance driven', 'Outer directed' and 'Inner directed') has already been recognised as a useful conceptual tool in understanding

consumer behaviour in the fields of leisure (Gratton and Taylor 1991) and Tourism (Dalen 1989). However, it has only been alluded to briefly in the context of food (Nelson 1986, Dade 1988); as such this typology presents a novel and valuable opportunity to investigate the impact of social values on consumer perception of food risk in the UK. It is the stated intention of this research to do so.

It is recognised that the study of 'risk relieving strategies' is a neglected area. This research therefore also aims to investigate how Taylor Nelson's social values may influence these.

CHAPTER THREE

RESEARCH METHODOLOGY AND DESIGN

The following chapter is in two parts. Part One is a methodological review, focusing predominantly on the broad approaches, methods and techniques used in this research. Part Two provides detail of the research design and process for both the pilot and the main study. (Also see Fig 1 for details of the research process).

3.1 PART ONE. METHODOLOGICAL REVIEW

3.1.1 RESEARCH PARADIGMS

Risking distortion which is frequently borne of simplification, it is possible to identify two main research traditions within the social sciences. They have variously been labelled positivistic versus humanistic (Hughes 1976), positivistic versus interpretative (Giddens 1976), scientific versus humanistic (Marindale 1974) and naturalistic versus humanistic (Walker 1985) or more simply quantitative versus qualitative approaches.

Adoption of a paradigm is based on philosophical assumptions. Assumptions about Ontology, the nature of reality whether it be believed to be objective or subjective. Epistemology, the relationship of the researcher to the researched, whether they are independent or interactive. Axiological, whether there is an imposition of values, is the researcher value free and unbiased or value laden and biased. Rhetorical, what kind of language is used, spoken and written is it formal or informal. These philosophical assumptions provide the context for the methodology and the design process for the research.

In theory, there is an assumption made that researchers should adopt the philosophy of one of the two paradigms. However increasingly it is accepted that a wealth of rich data can be obtained when a combination of the two paradigms is adopted.

3.1.2 THE QUALITATIVE PARADIGM

3.1.2.1 QUALITATIVE METHODOLOGY

Bryman (1993: 131) describes the qualitative approach as, "the study of the social world which seeks to describe and analyse the culture and behaviour of humans from the point of view of those being studied". Such an approach requires the ability and preparedness of the researcher to empathise with the mind set of those being studied.

Similarly Hakim (1987: 27) describes qualitative research as:-

"being concerned with individuals own accounts of their attitudes, motivation and behaviour. It offers richly descriptive reports of individuals perceptions, attitudes, beliefs, views and feelings, the meanings and interpretations given to events and things, as well as their experiences and illuminates the motivations which connect attitudes and behaviour, the discontinuities or even contradictions between attitudes and behaviour, or how conflicting attitudes and motivations are resolved in particular choices made".

Qualitative research was described by the Research and Development Committee of the Market Research Society (1979) as, in its very essence, the discovery of basic information about the nature and elements of new or familiar or changing consumer markets that can only be provided by means of exploration and discovery.

Peterson (1994), cited in Veal (1997: 130), examined qualitative research from a marketing perspective. Peterson outlines nine purposes of qualitative research:-

- to develop hypotheses concerning relevant behaviour and attitudes;
- to identify the full range of issues, views and attitudes which should be pursued in large scale research;
- 3 to suggest methods for quantitative enquiry for example in terms of deciding who should be included in interview surveys;
- 4 to identify language used to address relevant issues;
- to understand how a buying decision is made. Questionnaire surveys are not very good at exploring processes;
- 6 to develop new product, service or marketing strategy ideas;
- 7 to provide an initial screening of new product, service or strategy ideas;
- to learn how communication received what is understood and how processes e.g. buying decisions;
- 9 to develop a forum for brainstorming ideas, attitudes, opinions.

3.1.2.2 QUALITATIVE METHODS

The three main qualitative methods available to researchers are often recognised as participant observation, (advocated by Malinowski soon after the turn of the century), individual in-depth interviews (which may be unstructured or semi-structured, in each case the interviewee is given a much freer rein than in the survey interview) and group discussions (where the group leader guides the conversation of a small group / focus group). In all cases the researcher provides minimal guidance and allows considerable latitude for interviewees.

3.1.2.2.1 IN-DEPTH INTERVIEWS

In the social sciences one of the most commonly used qualitative methods is the in-depth interview, which is unstructured, of very variable length and may be extended into repeat interviews at later dates. Although the interviewer guides the discussion, often with the help of a list of topics to ensure coverage of the area of interest, the in-depth interview

provides enough freedom for responses to steer the conversation, often highlighting issues that the interviewer may not have covered using a more structured method. The informant's world of meaning is encouraged.

The questioning should be as open ended as possible to encourage respondents to talk spontaneously about their behaviour and opinions, the questioning technique should also encourage respondents to communicate their underlying attitudes, beliefs and values.

Hoinville et al (1983: 9) suggested that "Frank discussion can be impeded in the following ways, by the respondent attempting to rationalise, a lack of awareness, fear of being shown up and over politeness". To break down these barriers the interviewer must be able to put respondents at ease and make them feel that all views and all types of behaviour, attitude, opinion are acceptable and interesting.

A relaxed and unselfconscious interviewer who is not obviously from any particular socio-economic group is most likely to succeed in establishing the necessary rapport. Interviewers must ensure they are neither condescending nor deferential, that they look at respondents often enough to convey interest without appearing dominating or intrusive, and that they are tactful in their attempts to get at the meaning of vague or evasive statements.

In-depth interviews like group discussions should be tape recorded and transcribed, so that direct quotations from the respondents can be referenced to later in results. When the focus if fairly specific in-depth interviews with 20 - 25 people would be used (Hakim 1987). The more diverse and diffused the subject the larger the number required to identify reliable clusters or patterns of attitudes and related behaviours.

3.1.2.3 STRENGTHS AND WEAKNESSES OF QUALITATIVE METHODS

The great strength of qualitative research is the validity of the data obtained. Individuals are interviewed in sufficient detail for the results to be taken as true, correct, complete and believable reports of their views and experiences. In addition it provides a means of exploring the informant's world of meaning and motivation.

A weakness is that it is often expensive and as a consequence the small numbers of respondents investigated cannot be taken as representative, even if great care is taken to choose a fair cross section of the type of people who are the subject of the study. Additionally because of the non - standardised way the data are collected, metrical analysis may be difficult or impossible. Dean et al (1967: 275) suggest that another major limitation is "the researchers use of impressionistic interpretation of the data. It contains the possibility of bias and prejudice".

3.1.3 QUANTITATIVE PARADIGM

3.1.3.1 QUANTITATIVE METHODOLOGY

Quantitative methods are deeply embedded in the tradition of positivism viewing the study of society and human behaviour as scientific in the sense of the natural sciences. Positivism reveals itself in quantitative research through its particular emphasis on facts which are the products of observation, either direct or indirect. The stress on devising valid and reliable measurement procedures is especially redolent of positivism. The aim to generalise is also indicative of this approach.

Another fundamental belief shared by positivists is the view that the social and natural worlds conform to certain fixed and unalterable laws in an endless chain of causation. This view is shared by Labowitz and Hagedorn (1971). They believe that the major goal of scientific research is to establish causal laws that enable us to predict and explain

specific phenomena. To establish these laws a science must have reliable information or facts.

Mayntz et al (1976) suggest that the requirement to establish causal laws necessitates the formulation and empirical testing of explanatory theories which in turn 'tend towards a theory which is a deductively, indeed axiomatically structured system of empirically verified statements. Similarly positivism instils the requirement of reliable 'facts' leading in turn to an emphasis on rigour, objectivity and measurement.

3.1.3.2 QUANTITATIVE METHODS

Typically quantitative methods are taken to be exemplified by the survey questionnaire and experimental investigation. However, others include the analysis of secondary data, structured observation and content analysis (de Vaus 1986).

3.1.3.2.1 QUESTIONNAIRES

In the social sciences the survey questionnaire is often thought of as the main method of data collection which embodies the features of quantitative research. Most survey questionnaires are based on the underlying research design of correlational or cross - sectional investigation. Bryman (1993) describes this design as meaning that the data is collected on a cross - section of people at a single point in time in order to discover the ways and degrees to which variables relate to each other.

The survey questionnaire seeks an understanding of what causes phenomenon by looking at variation in that variable across cases, and looking for other characteristics which are systematically linked with it. As such it aims to draw causal inference by a careful comparison of the various characteristics of cases (de Vaus1986).

Moser and Kalton (1979: 308) point out, questionnaire design is of critical importance.

"....no matter how efficient the sample design or sophisticated the analysis, ambiguous questions will produce non - comparable answers, leading questions biased answers and vague questions vague answers."

Moser and Kalton (1979: 308) suggest that question design "is a matter of art rather than a matter of science".

Considerations involved in formulating the questionnaire are its content, structure, format and sequence (Nachmias and Nachmias 1981).

Questionnaire design can be approached as a process which involves several steps these steps may be presented as follows:-

- 1 Specify what information will be sought
- 2 Determine type of questionnaire and method of administration

Belson (1959: 68) has shown "that prompt short, precise list questions produce a higher yield than open questions. However responses may be unusable, non-comparable or irrelevant to the study".

- 3 Determine content of individual questions
- 4 Determine form of responses to each question
- 5 Determining wording /form of each question

A question may be unreliable due to bad wording. For example a person may understand the question differently on different occasions, or different people may interpret the question differently. The language should be familiar without being ambiguous, with a glossary if appropriate. The questions should never lead the respondent. Negative questioning should be avoided as they can cause confusion. Similarly questions which are memory dependent should not be used. Prestige bias in the question can be a

problem. Because certain behaviours, characteristics and attitudes have a higher prestige value, people may distort their answers to impress the interviewer or fool themselves. The implications of an answer should never seem negative i.e. the respondent should never be made to sound or feel selfish, unethical, immoral, complacent as a result of their response. Both of these features may lead to inaccurate results. Decisions must be made with regard to whether the questions should be direct or indirect, personal or impersonal, or in open or closed formats.

- 6 Determine sequence of questions
- 7 Determine physical characteristics of questionnaire
- 8 Re examine steps 1 7 and revise if necessary
- 9 Pre test questionnaire and revise if necessary

(Adapted) Source Churchill (1987), Moser and Kalton (1979), Belson, 1985

Questionnaire design is a cyclical process involving various levels of refinement at every stage.

3.1.3.2.2 RESPONSE FORMAT - FORCED CHOICE USING ATTITUDE SCALES

Thurstone (1928) recognised the possibilities in measuring attitudes. Since then scales are commonplace in attitude questionnaires. At its simplest, the scale may have only two possible answers Yes/No or Agree/Disagree. More commonly, a point scale is used. This may take several forms, verbal scales, spatial or diagrammatic scales.

"The object should be to find the way that discriminates most effectively between respondents" (Hoinville et al 1983: 9).

Osgood's semantic differential scale combines verbal and diagrammatic techniques. It consists of choosing adjectives to represent the two extremes of a continuum and asking respondents to put a mark between the two extremes. e.g.

An alternative to verbal and diagrammatic scales are numerical scales where respondents are asked to assign scores or marks to indicate the level of agreement, satisfaction or whatever is being measured.

Likert scales provide people with statements and ask them to indicate how strongly they agree or disagree. No uniform view has emerged on the best number of points to include in scales. For bipolar scales odd numbers have the advantage of offering a mid-point between two poles. Five point scales are probably the most frequently used however seven or nine can also be used on the grounds that they offer more scale positions and therefore offer greater discrimination (Hoinville 1983).

Likert scales lend themselves particularly well to correlation and to multivariate analysis such as factor analyses enabling the identification of patterns determining the broad dimensions underlying the particular attitudes expressed.

3.1.3.3 STRENGTHS AND WEAKNESSES OF QUANTITATIVE METHODS

The advantage of using quantitative questionnaires is that it is relatively cheap and less time consuming than other methods, particularly when the population is large. Fundamentally, the measurement of responses allows for a multitude of possible statistical analysis.

Questionnaires particularly in the form of a mail questionnaire are however notorious for low response rates when the sample is scattered. Non - response is of fundamental

importance as it may result in an unacceptable reduction in sample size and as a result introduce bias. These issues are most crucial when the aim is to achieve a sample that is representative of a wider population (de Vaus 1986).

When the aim is not to make generalisation about the population as a whole the issue of non - response is less crucial (de Vaus 1986), however every attempt should be made to reduce non-response. An initial sample can be drawn that is larger than is needed, attention should be made to the structure and wording of the questionnaire, and to the interviewer's approach. Common sense suggests that the longer and more complex a questionnaire the lower the rate of response and the greater the risk that respondents will be heavily biased towards those with spare time, mental stamina knowledge or interest in the subject being dealt with.

Bailey (1987) suggests that questionnaires that are clear (not ambiguous) and that have clear answer options are more likely to be successful in terms of minimising non response as well as generating valid and reliable responses.

3.1.4 QUANTITATIVE Vs QUALITATIVE PARADIGMS

The discussions about the nature and relative virtues of quantitative and qualitative research reveal a mixture of philosophical issues and considerations of the methods of data collection with which each of these research traditions is associated.

In early textbooks there was a clear awareness of the differences between quantitative and qualitative research. However, all of these discussions operated almost exclusively at the level of the technical adequacy of the techniques. Bryman (1993: 3) suggests that in the 1970s broader philosophical issues emerged into discussions about methods of research. "The pivotal point for much of the controversy was the appropriateness of a natural science model to the social sciences". "Much of the argument against the orthodoxy of quantitative research derived from the growing awareness and influence of phenomenological ideas, which gained a considerable following in the 1960's". It was

argued that the application of a 'scientific approach' in the form of surveys and experiments fail to take account for the differences between people.

In general, it is believed that qualitative methods allow for a more probing study of personal values, attitudes and experiences which is much more difficult to achieve when using quantitative methods. "Individuals are interviewed in sufficient detail for the results to be taken as true, correct, complete and believable reports of their views and experiences" (Hakim 1987: 27).

Walker (1985) suggests that sometimes the information required is just too subtle and too complex to be tailored to structured, standardised techniques or too detailed to be elicited by quantitative means

Hirschi and Selvin (1967) in contrast to the authors previously mentioned expressed their preference for quantitative research. They took the view that because quantitative data can be analysed statistically it is possible to examine complicated theoretical problems far more precisely than with the verbal analysis of qualitative data.

Labowitz and Hagedorn (1971) similarly support quantitative research for it's ability to identify causal links between data and recognise the strengths of quantitative methods in an aim to generate valid and reliable results.

One might suggest that quantitative and qualitative research are different ways of conducting research and that the choice between them should be made in terms of their appropriateness in answering particular research questions (epistemological). Equally one might suggest that the choice between quantitative and qualitative research is a technical decision, a view presented by Bryman (1993).

Guba and Lincoln (1982: 240) refer to quantitative and qualitative research as "resting on divergent paradigms, and hence assumptions, about the proper study of social life".

3.1.5 TRIANGULATION

Bryman (1993) proposed that by and large researchers have viewed the main message of the idea of triangulation as entailing a need to employ more than one method of investigation and hence more than one type of data. Within this context, quantitative and qualitative research may be perceived as different ways of examining the same research problem. By combining the two, the researcher's claims for validity of his or her conclusions are enhanced if they can be shown to provide mutual confirmation.

This eclectic approach has been adopted by many authors and researchers to include (Newby (1977), Patrick (1973), Oakley (1979), Roberts et al (1977) cited in McNeil (1990)).

If "surveys offer the bird's eye view, qualitative research offers the worm's eye view". This quote by Hakim (1987: 27) suggests inadvertently that an eclectic use of both methodologies can be advantageous in that it would give the full picture rather than just one perspective.

Veal (1997) believed the two approaches qualitative and quantitative complement each other and quantitative research should always be based on initial qualitative work.

Denzin (1970: 310) treats triangulation as an approach in which "multi-observers, theoretical perspectives, sources of data and methodologies are combined". He argues that "empirical reality is a reality of competing definitions, attitudes and personal values" he concludes that "multiple methods and theoretical approaches must be used. Four types of triangulation are proposed;

- data with respect to time, place, person and level;
- 2 between multiple observers of the same phenomenon;
- between multiple theoretical perspectives with respect to the same set of objects;
- 4 methodological triangulation."

The linking of qualitative and quantitative research methods is an increasingly common one, the results of the two linked studies sometimes being presented in a single report. As mentioned in the previous section a qualitative study is often carried out before the survey, as an exploratory first step that paves the way as well as offering a greater depth of information to complement the quantitative surveys results. Alternatively or additionally the qualitative study may be carried out after the main survey, which can then provide a rich sampling frame for selecting particular types of respondent for depth interviews. These types of linkage greatly extend the survey's results, and it may be possible to set the qualitative results in a statistical context by directly linking the two sets of data.

Qualitative research can be used in combination with virtually all other types of study, it may however be a particularly fruitful element of a research programme when used to inform the interpretation of more impersonal statistical data. This approach may reduce the risk that invalid conclusions will be drawn from the researcher's untested assumptions about the motivation and processes underlying correlations in the quantitative data, the attitudinal factors underlying observed behavioural differences between sub - groups, or the range of factors that might affect change in behaviour over an interval of time.

Hakim (1987) suggested that it is already wide spread practice in policy research and market research to use triangulation. However, the idea is still treated as sufficiently novel in some academic circles.

The key argument for triangulation is that it can add qualification to research. "Different methods can also complement each other, as when a survey provides a context for qualitative work which in turn permits commentary on the findings of the survey" (Bebbington 1983: 16).

3.2 PART TWO. RESEARCH DESIGN AND PROCESS

A wide variety of research methods are available for collecting empirical data. It is important that one chooses the most appropriate for the research in question. The selection process for the proposed research involved consideration of:- the research objectives, the theoretical background to the research and the practical constraints imposed by external factors.

As detailed in Chapter One the overall aim of the research was to investigate and clarify how the social values, as identified by Taylor Nelson and reported by MacNulty (1985), influence consumers' perceptions of food related risk and preferred risk relieving strategies.

The research aims and objectives required that an eclectic approach be used i.e. a combination of both quantitative and qualitative techniques.

More specifically the methods chosen needed to facilitate the:-

1. Measurement of :-

Social Values - Part Two of the main study questionnaire -

The Screening Instrument

Food Risk Perceptions - Part Three of the main study questionnaire

Risk Relieving Strategies - Part Four of the main study questionnaire

(Part One of the questionnaire was a filter through which the main study sample was selected i.e. women aged 25 - 45, with no dependent children, economically active, with a household income exceeding £12,000 p.a.)

The questionnaire survey was piloted before being used in the main study.

2. Categorisation of the survey respondents into those who were predominantly:

Sustenance driven

Outer directed

Inner directed

No dominant group indicated

(Done on the basis of the social value data generated by the survey and using a set of decision rules devised for the purpose)

- Quantitative analysis of the relationship between the social values, risk perceptions and preferred risk relieving strategies of all respondents in the main study using the data generated from the main survey (Pearson correlations and factor analysis). Data treated as interval.
- 4. Quantitative analysis of similarities / differences in food risk perception and preferred risk relieving strategies for the members of the three main Social Value Groups identified in 2. above (Analysis of Variance using survey data directly and using derived factor scores). Data treated as interval.
- 5. Qualitative interviews to establish that the rules for categorising respondents were working in practice (Interview One undertaken during the pilot stage).
- Qualitative interviews to establish the motivations that lay behind the similarities / differences in food risk perception and risk relieving strategies for the three main Social Value Groups under investigation.
 (Interview Two on food risk perceptions only, undertaken during the pilot study)
 (Interview Three on food risk perceptions and risk relieving strategies, undertaken after the main survey)

The research process is detailed in Fig 1 Chapter One.

The remainder of this Chapter which is divided into three sections focuses on each stage of the research design / process as it happened i.e. chronologically. The first section details the design of the Social Value Group screening instrument and Interview One. The second section details the design of the Food Risk Perception and Risk Relieving Strategies parts of the questionnaire and Interview Two. The third section of this chapter provides details of the main study to include the sample and Interview Three.

3.2.1 SOCIAL VALUE GROUP (SVG) SCREENING INSTRUMENT (Part Two of the Questionnaire)

3.2.1.1 DESIGN OF THE SCREENING INSTRUMENT

Part Two of the questionnaire (Appendix 1.1) was designed to measure the degree to which the respondent agreed with value statements that were inherently Inner directed, Outer directed and Sustenance driven, the ultimate aim was to categorise the respondents in relation to Taylor Nelson's three Social Value Groups.

The screening instrument was in the form of a structured, closed end questionnaire consisting of a number of value statements. The statements initially twelve in number i.e. four per Social Value Group then eighteen i.e. six per Social Value Group, highlighted the key traits of each of Taylor Nelson's Social Value Groups as identified by MacNulty (1985) (see Table 2.1 in the Literature Review)

Whilst MacNulty (1985) provided a clear description of the social values of the three Social Value Groups, it was felt there was a need to pilot the instrument to be confident of its reliability and validity, and to confirm its currency given that MacNulty's paper was published in 1985. As with Rokeach's (1960) research on dogmatism, a 'known group' of individuals was selected for the pilot. The group comprised nineteen colleagues whose social values and attitudes were broadly known by the researcher as a

consequence of workplace contact with them, to include individuals who were likely to fall into each of the three Social Value Groups.

During the pilot, the (SVG) screening instrument was refined on four occasions. This was achieved by using those respondents in the pilot who evidenced clear cut SVG orientations, as determined by the results of their initial responses to the screening instrument and the outcome of interview One, described later. This process essentially enabled the reliability of the instrument to be examined via a test and re-test method. A high degree of correspondence between respondent's answers on each occasion confirmed the reliability of the instrument.

3.2.1.2 USE OF THE SCREENING INSTRUMENT TO CATEGORISE RESPONDENTS INTO SOCIAL VALUE GROUPS

When completing the screening instrument respondents were asked to indicate their responses to the statements using a five point Likert scale. In the final version of the instrument the respondents were given a score for their responses to all of the eighteen statements, receiving a score for each response as follows; Agree a lot = +2, Agree = +1, Neither Agree or Disagree = -1, Disagree a lot = -2.

The scores were summated allowing each respondent to be allocated a score for the degree to which they expressed Inner directed, Outer directed and Sustenance driven values. In the final version of the instrument the highest and lowest possible score was +/-12 on each dimension. Respondents were then categorised into a Social Value Groups on the basis of these score profiles (See Chapter Four)

This system of awarding each respondent a score for each value dimension and categorising respondents accordingly was successfully used by Earl and Cvetkovich (1995) cited in Cvetkovich and Lofstedt (1999).

Whilst the primary aim at this stage of the research was to design a screening instrument to identify members of Taylor Nelson's Social Value Groups, other exploratory work was undertaken during the pilot. This included an investigation into how the members of Taylor Nelson's Social Value Groups performed when responding to Rotter's Locus of Control questionnaire (Rotter 1966, 1971). The results of this work illustrated interestingly that Inner directed members of the pilot sample had an internal locus of control, whilst Sustenance driven and (though to a lesser extent) Outer directed members of the pilot group were externals.

In addition to the administration of Rotter's Locus of Control questionnaire the degree to which the respondents in the pilot perceived that they were personally at risk from related food risks compared with other members of society was explored, together with the degree to which they believed that they were personally responsibility for protecting themselves against the risk they perceived they were exposed to.

Whilst these areas of further exploration were interesting and worthy of investigation they were not strictly within the scope of the research and as a result were not investigated any further in the main study.

3.2.2 INTERVIEW ONE

In order to refine the screening instrument (Part Two of the questionnaire), to test the reliability of responses to it and increase its validity, in-depth interviews were undertaken with the nineteen respondents involved in the pilot. The interview took place one week after they had completed the screening instrument.

At the beginning of the interviews, interviewees were asked to comment on structure and content of the screening instrument questionnaire. They were also asked to highlight any potential areas of confusion, and suggest additional issues important to them that appeared to have been missed. This short period of time gave the interviewee time to relax a little before the interview became more focused and in - depth.

The interviewees were asked to talk in depth about their motivations, values and attitudes regarding fundamental elements of their life, to include the family, work, preferred organisational structures (working environment), leisure and attitudes towards authority. These were the areas covered by MacNulty (1985). In addition they were asked what they thought might improve their lifestyle and what made them feel good about themselves.

A relaxed informal atmosphere was created by meeting the interviewees on their own territory, a room was always used where interruptions would not occur. A conscious attempt was made to relax the interviewee, they were told there were no wrong answers and that all views, behaviour, attitudes and opinions were important and of interest. The aim of this approach was to ensure that the interviewee would feel able to speak openly and freely.

In an attempt to appear relaxed and unselfconscious the interviewer dressed smart / casual, and tried to be as approachable, agreeable, interesting, interested, enthusiastic and neutral in terms of political stance, social class and status as possible. The interviewer also tried to give support without introducing bias, and was careful about her verbal and non verbal language.

All of the interviews were recorded and latter transcribed. This meant that it was easy to go back and verify points, find quotations to support theories, and more generally to avoid biased reporting arising from too much reliance on first impressions and memory.

3.2.3 RESULTS OF PILOTING THE (SVG) SCREENING

INSTRUMENT (Part Two of the Questionnaire) AND INTERVIEW ONE

During the pilot the screening instrument successfully identified dominant members of each of the three Social Value Groups. All predictions with respect to the 'known group' members were confirmed.

Five members of the pilot sample exhibited a mixture of the values and attitudes of more than one of the three Social Value Groups. Four respondents clearly and unquestioningly exhibited motivation, values and attitudes exclusive and typical of the Sustenance consumer, five of the Inner directed consumer group and three of the Outer directed consumer group.

A high degree of correspondence between the motivations, attitudes and values demonstrated in the interviews and those detailed in MacNulty (1985) indicated the validity of the screening instrument as a mechanism for identifying Social Value Group members. Of the nineteen respondents in the pilot study sample two demonstrated contradictions between their questionnaire and interview responses. As mentioned previously three members of each Social Value Group (identified as a result of the pilot) completed refined and revised versions of Part Two of the questionnaire over a five month period. Their responses were cross checked with their previous responses. A high degree of correspondence between their answers on each occasion indicated the reliability of the instrument.

As a result of the pilot several modifications were made to the screening instrument. The sequence of the questions was changed so that the least controversial and simplest were at the beginning. Very negative words were replaced e.g. 'I am resistant' was changed to 'I am not keen'. An initial twelve statements (four for each Social Value Group) were extended to eighteen (six for each Social Value Group), the aim of the additional statements was to repeat issues already presented but in a different context, to improve

the sensitivity of the instrument. Another important addition was the attachment of a glossary. During the interviews respondents' understanding of the words in the questionnaire were cross checked by asking them to discuss their responses. It became apparent that some words were being interpreted differently by different respondents. A glossary was therefore added to the end of the questionnaire with a note as a way of ensuring the consistent interpretation of potentially ambiguous words such as 'unconventional' and 'alternative' (see Appendix 1.1). The predicted problem of 'prestige bias' was confirmed with some of the Inner directed statements, the scoring criteria for the categorisation of respondents into Social Value Groups was designed to compensate for this (see Chapter Four).

The results of the pilot indicated that members of each Social Value Group could successfully be identified using the screening instrument alone.

An in - depth analysis of the results of the Screening Instrument and Interview One can be found in Chapters Four and Five respectively.

3.2.4 FOOD RISK PERCEPTION AND RISK RELIEVING STRATEGY QUESTIONNAIRES

(Part Three and Part Four of the Questionnaire)

Parts three and four of the questionnaire (Food Risk Perceptions and Risk Relieving Strategies) was piloted using the same nineteen respondents that were used when designing and piloting the screening instrument. Having developed a more in-depth understanding of these consumer's social values, the aim was now to investigate how their social values were influencing their responses to parts three and four of the questionnaire.

3.2.5 DESIGN OF FOOD RISK PERCEPTION QUESTIONNAIRE

(Part Three of the Questionnaire)

Consumers are everyday confronted with purchase decisions. Research suggests that we perceive a variety of risks associated with the consumption and purchase of food. It has been recognised that different consumers perceived the same risks in varying degrees and often perceive very different risks all together. Douglas and Wildavsky (1982a) suggested that we choose what to fear in a way that reinforces our social values. What influences these differences is key as it is fundamental to consumer decision making.

When one examines previous research we observe that risk is often discussed in the context of perceived loss. Many authors have categorised loss, (Mitchell and Greatorex (1990), Kaplan et al (1974), Soby et al (1994), Toh and Heeren (1982)). Collectively these categories of loss are financial/money, time, physical/health, psychological/ego, social, performance, ethical. It is suggested that risks ultimately threaten the loss of one or more of these and consumers will fear these losses to a greater or lesser extent and perceive risks accordingly.

With the objective of having a broad perspective and being conscious of the dangers of setting narrow boundaries, a comprehensive list of potentially perceived risks was constructed. This carefully designed list included risks that potentially threatened the values and attitudes of all three of Taylor Nelson's Social Value Groups (important for sampling validity). All risks potentially having lesser or greater significance depending on the respondents and their value system / 'world view'. It was predicted for example that Sustenance driven respondents would, due to their need to feel secure, be highly concerned about financial risks. Similarly it was predicted that Outer directed respondents, aspiring to high social and financial status, may be concerned about the image and reputation of the shops they use and the brands they buy. In contrast Inner direct respondents would be concerned about what were labelled 'spiritual risks' due to their awareness of world / social events and ecological awareness.

Part Three of the questionnaire included questions on the following potentially perceived food risk issues: nutrition, additives, preservatives, labelling, technology, contaminants, environment, price, reputation and image, food poisoning, unfamiliarity, time and skill, stimulants, eating out, place of origin.

Forty eight variables in total were included. The respondents were asked to indicate on a nine point Likert scale the degree to which they were worried about the risks concerned. The spectrum of the scale was designed to give respondents the opportunity to clearly differentiate between their responses.

3.2.6 DESIGN OF RISK RELIEVING STRATEGY QUESTIONNAIRE

(Part Four of the Questionnaire)

It is human nature to attempt to reduce or relieve potential harm. Research detailed in the literature review suggests how consumers attempt to manage perceived risk by adopting different risk relieving strategies.

If one has knowledge of what the most effective risk reducing / relieving strategies are for particular target markets, this can be integrated into the design and marketing of the products concerned and into relevant communication strategies. The strategy most effective for one consumer group might be totally ineffective for another. Only a deep understanding of one's target market can enable the development of effective and efficient strategies. Part four of the questionnaire aimed to investigate what consumers might perceive as effective ways of reducing / relieving risk and how these strategies differed between Social Value Groups.

In designing and developing the risk relieving strategy questionnaire, the values and attitudes of the three Social Value Groups provided an initial framework. Strategies which might support and reinforce consumer's social values were included. Following on from previous research the perceived risk was the starting point. Scenarios and

relationships were theorised, for example if a consumer perceived a financial loss as a key risk, it was hypothesised that shopping in a 'cheaper' supermarket might be an effective risk relieving strategy, whilst shopping in an 'expensive' supermarket clearly would not. In contrast if a consumer values social status, image and reputation and perceives the loss of these as a risk, shopping in an expensive supermarket may indeed be an effective risk reliever.

As the review of the literature highlighted, previous research into risk relieving strategies has identified the key methods adopted by consumers in attempts to reduce perceived risk, these are:- endorsement, branding, brand loyalty, shopping around, image (store and brand), reputation, private testing, free sample, money back guarantee, government testing, buying the expensive, word of mouth, seeking information (Roselius 1971)

Derbaix (1983) added sales person advice. Other important features of risk relief raised in the literature were level of knowledge and information, source of information, degree of trust in that source, and familiarity with the risk. These were all incorporated into the questionnaire. It was important that there be risk relieving strategies for all Social Value Groups and that there be a risk relieving strategy for all potentially perceived risks.

Part four comprised a list of seventeen risk relieving strategies, the respondents were asked to indicate on a nine point Likert scale the degree to which the strategy might relieve or reduce any worry they might have about a food risk / food purchasing risk. As with Part three the nine point Likert scale gave a wide spectrum so that respondents could discriminate between the strategies.

3.2.7 INTERVIEW TWO

The aim of Interview Two was to explore, clarify and verify the responses to Part Three of the questionnaire. Consumers food risk perceptions were thought, at the design phase, to be the primary focus of the research. As a result Interview Two focused on responses to part three of the questionnaire. As the research progressed it was recognised that risk reliving strategies should be focused on to the same degree as risk

perceptions. This was indeed the case in the Interview Three (which followed the main study), the findings of which can be found in Chapter Five.

From the original pilot sample, three of the four Sustenance driven respondents, three of the five Inner directed respondents and the three Outer directed respondents were selected to be interviewed with respect to their responses to Part Three of the questionnaire. All interviewees were told that the aim of the interview was to avoid inaccurate assumptions being made about their food risk perceptions. It was explained that they may be asked questions that may seem very obvious, but that there was a need to clarify and verify assumptions. The interviewees were allowed to 'ramble' to allow important issues to be raised that might otherwise have been missed.

The aim of the interview was to investigate why the respondents had indicated a high degree of worry about some of the risks whilst others did not worry them at all. The interviewees were asked, "why does this worry you", "why does this not worry you".

3.2.8 RESULTS OF PILOTING PART THREE AND PART FOUR OF THE QUESTIONNAIRE AND INTERVIEW TWO

Respondents from the three different Social Value Groups indicated differing degrees of worry about different types of risks and about the types of risk relieving strategies that might relieve those risks. The outcomes of Interview Two (some of which are reported in Chapter Five) supported the results of the questionnaire and clarified why respondents answered the questionnaire in the way that they did.

The refined and redrafted Food Risk Questionnaire (Part Three) reduced the number of perceived risks from forty eight to twenty eight. Variables were omitted when the responses were the same for all respondents i.e. all respondents perceived that the risk was something to be worried about or alternatively all respondents perceived that the

risk was not a worry. Food poisoning fell into the former category and as such was removed form Part Three of the questionnaire.

Only very minor changes were made, mainly grammatical, to Part Four of the questionnaire.

3.2.9 MAIN STUDY

3.2.9.1 THE SAMPLE

Part One of the main study questionnaire was designed as a filter to obtain a suitable sample.

Taylor Nelson identified members of each Social Value Group at every social and economic level and every socio economic and demographic variable within each of the Social Value Groups. It might therefore be assumed that a representative sample of the general population would serve the needs of the research. However it was judged that there were a number of biases that might obscure the possible relationships between social values and risk perceptions had a mixed sample been used in this way as follows:-

Firstly it was important that the responses made to the food risk and risk relieving elements of the questionnaire were informed and considered. It was therefore felt that the sample should consist of women rather than men. Women are on the whole more involved and more knowledgeable about food provisioning. In addition there was some concern that men may play down their responses to food risk to 'save face'.

Taylor Nelson's theory is strongly influenced by Maslow's hierarchy of needs, it therefore is appropriate to exclude what Maslow (1970) describes as 'one source of confusion' by following his lead and 'confining the sample to 'older' people' of twenty five years or more. Maslow (1970) believed that self actualisation, (a key trait of Inner directed consumers) does not occur in young people. With regard to Taylor Nelson's

research, younger respondents would not have the social values of Inner directed consumers and would therefore bias the sample.

It was believed that respondents whom worked full time and had no children living at home would have more independent values predominantly motivated by their own wants and needs rather than their consideration of those of others such as their children.

With a total household income of over £12,000 p.a. (11,855pa average household income (Source - 1992 family expenditure survey, Annual Abstract of Statistics 1995) it was believed respondents who did indicate that financial risk was an issue for them were doing so because of their social values rather than because they were in fact financially constrained.

For these reasons the sample was thus to consist of women aged 25 - 45, with no dependent children, economically active with an income above the national household average.

Given that the intended sample was not representative of the whole population and given that the research had no funding at all, it was decided to gather a convenience sample of this type of individual from the staff of Sheffield Hallam University. This was achieved by distributing the questionnaire across the entire full time staff (interested parties were excluded). Men were also surveyed, to provide data for a possible further analysis of this group at a later date.

The total number of individuals surveyed was 1957. Of those surveyed 62% responded. Of those, 215 respondents fell into the group of interest (the homogeneous sample). These individuals formed the sample. As the distribution of women across the whole University in the category of interest was not known, it is not possible to say what percentage of this group responded to the survey. However there is no reason to believe that the non-respondents differed from the respondents in ways that would have effected the results. The consequence of using a convenience sample are that the results of the research are suggestive rather than generalisable to a wider population.

The final question in this part of the questionnaire asked the respondent if they would be prepared to participate at a later date in a follow-up. This gave a pool of respondents from whom a sample could be later selected for follow up Interviews (Interview Three).

3.2.9.2 DISTRIBUTION OF THE QUESTIONNAIRE AND COVER LETTER

A cover letter accompanied the questionnaire, it was written with the aim of overcoming any resistance or prejudice the respondent may have against completing the questionnaire.

The cover letter identified the researcher and the department conducting the study, explaining the purpose of the study, the importance of the study and what it aimed to achieve. Confidentiality and anonymity were assured.

An addressed return envelope was enclosed with an aim to increasing response rates. Internal post was used avoiding the cost of stamps.

The questionnaire was sent out in October 1996. October was chosen to avoiding both summer and Christmas vacations. Posted on a Monday to arrive Tuesday with the hope that respondents might have time to complete it before the week end.

As the return date approached the questionnaire was followed up with a reminder via e-mail. After the closing date another e-mail was sent out thanking those who had returned their questionnaire. Stressing the importance of the questionnaire a third request was made asking respondents to still return completed questionnaires even though the return date had passed. A contact was given on each occasion encouraging people to contact the researcher with any problems or / and questions.

3.2.9.3 **QUESTIONNAIRE RESPONSE RATE**

As previously mentioned the response rates were as follows:-

Total Returned and Completed Questionnaires

1216

Men

515

Women

701

of which were Homogeneous

group members

215

$$R = 1 - (n - r) / n$$

Response Rate = $1 \times (1957 - 1216) / 2000$

Response Rate

87% (1056) of the completed questionnaires were returned by the first deadline date. The other 13% (160) followed after reminders had been sent to the sample.

18% (215) of the returned and completed questionnaires were from the homogeneous group of interest as previously described (the largest homogeneous group arising). The 215 responses were used for analysis.

3.2.9.4 RESULTS OF THE MAIN STUDY

Results of the quantitative analysis of parts, two three and four of the main study questionnaire can be found in Chapter Four together with a full discussion of the statistical techniques used in the analysis. Results of the qualitative analysis can be found in Chapter Five.

3.2.10 INTERVIEW THREE

Interview Three took place after the questionnaire responses had been analysed.

Interview Three provided a vehicle through which to explore motivations beneath perceived risks and preferred risk relieving strategies identified in the main study. So far assumptions about motivation had been inferred, so for example, when a respondent had indicated that endorsement via a family member was an effective way of feeling less worried about a food risk the researcher assumed that this was because that respondent valued the judgement of that family member. The interviews gave the opportunity to confirm these assumptions.

Six members of each Social Value Group were randomly selected for follow up interviews. Five Inner Directed consumers, six Sustenance Driven and four Outer Directed consumers agreed to take part in the follow up interviews.

At the beginning of the interview interviewees examined their responses to Parts One and Two of the questionnaire (which they had filled out the previous year). They were asked to indicate if any of their demographic or socio-economic circumstances had changed. They were also asked to comment on whether they still agreed with all their previous responses. They were given a pencil and instructed to make changes as they saw fit. Minor changes were made to Parts Two, Three and Four, thus one respondent indicated that she had recently had a baby, (however she was sure that she could recall her mind set from when she had completed the questionnaire). This was the only change to Part One. The interview then progressed concentrating firstly on responses to Part Three of the questionnaire, Interviewees were interviewed in the same fashion as described for Interview Two. Questions such as, "are these worries for you?", "why?", "is this not a worry to you?", "why?" were included. The interview then progressed to Part Four of the questionnaire. The Interviewees were again asked to discuss their previously given responses and comment on why some of the strategies were effective risk relievers for them whilst others were not.

The outcomes of the interviews provided verification of and added validity to the findings of the quantitative analysis of the main study data.

The analysis of the results of Interview Three can be found in Chapter Five.

CHAPTER FOUR

QUANTITATIVE ANALYSIS

The following chapter details the statistical analysis of the questionnaire data. Analysis of the qualitative interviews follows in Chapter Five.

Exploratory analysis provided the opportunity to develop an initial insight and feel for the data and importantly generated the information needed to establish whether it was suitable for parametric or non - parametric analysis. The data was interpreted as interval, whilst it is possible to question this interpretation, it has been recognised as common practice to use parametric tests for social science data of the type used in this research (Kaplan 1974, Davis 1999, Goldsmith et al 1997). Parametric tests also assume scores to be normally distributed therefore during the initial exploratory phase a series of frequency distributions were generated to verify that the data was sufficiently normal to allow for parametric tests. The final investigation undertaken to corroborate the use of parametric tests was to test for homogeneity of variance, the results of these tests confirmed that the variability of the scores in each condition were indeed approximately the same. The use of parametric tests is discussed further in Chapter 6 (Discussion).

Following the exploratory analysis, the data set from each part of the questionnaire were firstly analysed independently before examining any possible interrelationships between them. A series of univariate and multivariate techniques were used in the analysis, these included correlation, factor analysis and analysis of variance.

4.1 TECHNIQUES OF ANALYSIS

4.1.1 PEARSON CORRELATION

Pearson correlation is a technique typically used to identify the strength of relationships among parametric data. Measuring the relationship between variables and the strength of the association, it is useful in showing the degree to which variables are linearly related. No claim is made that this necessarily indicates cause and effect. Correlation is useful in that it may give further confirmation of a relationship that theory suggests exists.

A series of correlation matrices allowed for both correlation within and between responses to Part Two, Part Three and Part Four of the questionnaire. Positive and negative correlation coefficients were equally important, allowing for the examination and identification of the direction of correlation. Non significant correlations were examined in depth along with significant correlations recognising their value in supporting or refuting emerging themes and patterns.

A correlation matrix was generated for the whole sample of 215 respondents and for the smaller sample of the 54 most dominant respondents. The results for the two samples were examined and compared. On the whole there was little difference between the two other than that the correlation coefficients for the sample of 54 were generally higher. The direction and the themes were on the whole congruent between the two samples.

4.1.2 ANALYSIS OF VARIANCE

A One - way (unrelated) Analysis of Variance was used to measure the differences between the mean scores of the responses to each dependent variable in both Part Three and Part Four of the questionnaire for the three Social Value Groups (Independent variable). Comparing the results for three or more independent groups, Analysis of Variance tests the equality of the means of the populations concerned and calculates

whether these are significantly different. Post hoc tests allow for the specific identification of the location of these differences.

4.1.3 SIGNIFICANCE LEVELS

A key aim of statistical tests is to indicate the percentage probability that one's results are due to chance rather than due to the predicted effects of the independent variable. The significance level allows for the statistical measurement of the degree of chance and as a consequence supports or rejects the hypothesis at a pre-determined level.

The significance level is established by calculating from the scores in an experiment the proportions of total variance which are due to the independent variables, and the proportion due to all other variables (error variance).

For the purpose of this research it was decided that the level of significance should be set at < 0.05, a level of significance widely adopted in the social sciences (Greene and D'Oliveira 1995). Using this level there is a 0.05 probability that your results are due to chance, i.e. a 95% chance that the results are statistically significant.

4.1.4 FACTOR ANALYSIS

Factor analysis is a statistical technique used to identify a relatively small number of factors that can be used to represent the relationships amongst sets of inter-related variables, thus generating a smaller number of variables to explain most of the variance in the data and thereby providing a clearer and more concise description of that data. Principal Component Analysis was used to generate factors and Varimax rotation was applied to aid interpretation of them.

An exploratory phase of factor analysis was used to examine eigenvalues and possible levels of suppression for factor loadings. Tucker et al (1969) cited in Hair (1995) suggest that the most commonly used technique for identifying factors i.e. to select only

those with an eigenvalue of more than 1 is not always the best solution. Despite this, only factors with eigenvalues greater than one were used in line with the results of the relevant scree plots and bearing in mind the view of Aaker (1971) cited in Hair (1995) and Nunnully (1967) that this approach is generally the most appropriate one. Similarly the level at which factor loadings should be suppressed is commonly regarded as 0.5, however this text book solution is not always the most valuable. Many of the factor loadings in this research were less than 0.5 and therefore the question arose to whether 0.5 was the best level at which to suppress them. The results of the factor analysis undertaken are therefore presented displaying the results at three levels of suppression i.e. 0.5, 0.4 and 0.3.

Factor analysis was used to analyse the data from Part 2, Part 3 and Part 4 of the questionnaire for the whole sample of all 215 respondents. The aim of the factor analysis to identify factors or dimensions that represented the data on risk perceptions and risk relieving strategies in terms of a set of simplified constructs.

4.2 ANALYSIS OF THE SCREENING INSTRUMENT

(Part Two of the Questionnaire)

4.2.1 PEARSON CORRELATION RESULTS

Pearson correlations were carried out to examine the relationships between the eighteen social value variables within the screening instrument. The eighteen variables (six Sustenance driven ones, six Outer directed ones, and six Inner directed ones) were designed to represent the social values of the Social Value Groups. The Pearson correlations identified and confirmed the expected internal relationships between the social value statements and the robustness of the screening instrument when used in practice.

The following tables summarise the findings of the Pearson correlation matrix for Part Two of the questionnaire for the total homogeneous sample of 215 (Appendix 3.1).

Please refer to Appendix 1.1 for a copy of the questionnaire and Appendix 1.2 for variable codes / abbreviations.

Please refer to Appendix 3.1 for the correlation coefficients and significance levels for the total homogeneous sample of 215 and Appendix 2.1 for correlation coefficients and significance levels for the dominant sample of 54.

4.2.1.1 SUSTENANCE DRIVEN SOCIAL VALUE VARIABLES SIGNIFICANT CORRELATIONS (215 respondents)

The following set of tables (4.1 - 4.6) examine each Sustenance driven social value variable in turn ('Secure', 'No change', 'Setways', 'Tradition', 'Hold existing life', 'Making few contacts') and identify both positive and negative significant correlations (<0.5) between that variable and the other seventeen social value variables included in the screening instrument.

Table 4.1 Sustenance driven value variable "I am primarily motivated by a need to feel secure"

Secure		S		S		s		s
		v		$ \mathbf{v} $		$ \mathbf{v} $		v
		g		g		g		g
+ve correlation	nochange	S	setways	S	trad	S	holdlife	S
-ve correlation	world	I	unconve	I	** **		-	

⁴ out of a possible 5 +ve correlation

Table 4.2 Sustenance driven value variable "I am not keen to seek change"

Nochange	1	S		s		S		S		S		S
		v		v		v		v		v		v
		g		g		g		g		g		g
+ve correlation	secure	S	setways	S	trad	S	holdlife	S	appear	O	contacts	S
-ve correlation	unconve	1	winning	O	ahead	O	voiceops	I	status	О	alternat	I

⁵ out of a possible 5 +ve correlation

Table 4.3 Sustenance driven value variable "I tend to be set in my ways"

Setways		s		s		s	7	s		S
		v		v		v		v		v
٠.	:	g		g		g		g		g
+ve correlation	secure	S	nochange	S	trad	S	holdlife	S	appear	0
-ve correlation	alternat	I								

4 out of a possible 5 +ve correlation

Table 4.4 Sustenance driven value variable "I believe that the breakdown of tradition causes instability in society"

Trad		s		s		s		s	-	s
		v		v		v		v		v
		g		g		g	٠	g		g
+ve correlation	secure	S	best	0	nochange	S	setways	S	appear	0
-ve correlation	unconve	I	voiceops	I	alternat	I				

3 out of a possible 5 +ve correlation

Table 4.5 Sustenance driven value variable "I tend to hold onto my existing lifestyle"

Holdlife		s		s		s
		v		v		v
		g		g		g
+ve correlation	secure	S	nochange	S	setways	S
-ve correlation						

3 out of a possible 5 +ve correlation

Table 4.6 Sustenance driven value variable "I invest relatively little time in making new contacts / friends"

Contacts		S		S
		v		v
		g		g
+ve correlation	nochange	S		
-ve correlation	ahead	0	voiceops	I

1 out of a possible 5 +ve correlation

SUMMARY: SUSTENANCE DRIVEN VALUE VARIABLES

The tables identify that Sustenance driven variables significantly and positively correlate with other Sustenance driven variables. Importantly, whilst the tables above only detail those correlations which are significant, examination of Appendix 3.1 illustrates that all Sustenance driven value variables correlate positively with each other with no negative correlations amongst them.

There are significant positive correlations between the Outer directed variable, "I take pride in keeping up appearances" and the Sustenance driven variables "I tend to be set in my ways", "I believe that the breakdown of tradition causes instability in society" and "I am not keen to seek change" and between the Outer directed variable "I am motivated by the desire to have the best" and the Sustenance driven variable "I believe that the breakdown of tradition causes instability in society" these are the only non-Sustenance driven variables that correlate positively with Sustenance driven variables. If one examines the variables that correlate significantly in a negative direction these are all Inner directed and Outer directed value variables. The variable "I invest relatively little time in making new contacts / friends" correlates significantly in a positive direction with only one other Sustenance driven variable, whilst all correlations with other Sustenance variables are positive they are not significant, this may be an indication that this variable is the least helpful in identifying Sustenance driven consumers.

4.2.1.2 OUTER DIRECTED SOCIAL VALUE VARIABLES SIGNIFICANT CORRELATIONS

The following set of tables (4.7 - 4.12) examine each Outer directed social value variable in turn ('Best', 'Winning', 'Ahead', 'Status', 'Appearance', 'My success') and identify both positive and negative significant correlations (< 0.5) between that variable and the other seventeen social value variables included in the screening instrument.

Table 4.7 Outer directed value variable "I am motivated by the desire to have the best"

Best		s		S		s		s		S
		$ \mathbf{v} $		v		v		v		v
		g		g		g		g		g
+ve correlation	winning	0	trad	S	ahead	0	appear	0	status	0
-ve correlation					· · · · · · · · · · · · · · · · · · ·					

4 out of a possible 5 +ve correlation

Table 4.8 Outer directed value variable "Winning is a priority for me"

Winning		s		S		s		s		s		S
		$ \mathbf{v} $		v		v		v		v		v
		g		g		g		g		g		g
+ve correlation	best	0	ahead	O	voiceops	I	status	O	appear	0	mysucces	0
-ve correlation	nochange	S	····									

5 out of a possible 5 +ve correlation

Table 4.9 Outer directed value variable "I am strongly motivated to get ahead of others"

Ahead		S		s		s		S		s
		v		v		v		v		v
		g		g		g		g	٠.	g
+ve correlation	best	o	winning	O	status	0	appear	О	mysucces	0
-ve correlation	nochange	S	contacts	S						

5 out of a possible 5 +ve correlation

Table 4.10 Outer directed value variable "I'm motivated by a desire to improve my status"

Status		S		S		S	-	S		S
		v	!	v		\mathbf{v}		v		v
		g		g		g		g		g
+ve correlation	best	0	winning	Ō	ahead	0	appear	O	mysucces	0
-ve correlation	nochange	S	<u>-</u>				,			

5 out of a possible 5 +ve correlation

Table 4.11 Outer directed value variable "I take pride in keeping up appearances"

Appear		s		S		s		s		S		S
		v		v		v		v		v		v
		g		g		g	٠	g		g		g
+ve correlation	best	0	nochang	S	winning	0	setways	S	trad	S	ahead	O
-ve correlation	unconve	I	alternat	I								

Appear (cont.)		S		S
		$ \mathbf{v} $		v
		g		g
+ve correlation	status	O	mysucces	0
-ve correlation				

5 out of a possible 5 +ve correlation

Table 4.12 Outer directed value variable "I tend to make sure that other people know all about my successes"

Mysucces		s		S		s		s		s		S
		v		v		v		v		v		v
		g		g		g		g		g		g
+ve correlation	discover	I	winning	О	ahead	0	status	0	appear	O	mysucces	0
-ve correlation												

5 out of a possible 5 +ve correlation

SUMMARY: OUTER DIRECTED VALUE VARIABLES

The tables identify a strong set of correlations, with comparatively high correlation coefficients (see Appendix 2.1). Outer directed variables significantly and positively correlate with <u>all</u> other Outer directed variables with only one exception i.e. the correlation between "I tend to make sure that other people know all about my successes" and "I am motivated by the desire to have the best", is not significant.

There is a significant positive correlation between the Sustenance driven variable; "I believe that the breakdown of tradition causes instability in society" and the Outer directed variable "I am motivated by the desire to have the best", and between

Sustenance driven variables "I am not keen to seek change", "I tend to be set in my ways", "I believe that the breakdown of tradition causes instability in society" and the Outer directed variable "I take pride in keeping up appearances" perhaps indicting that "I take pride in keeping up appearances" is not unique to Outer directed respondents.

A positive correlation exists between the Inner directed variable "I voice my opinions even if it means disagreeing with others or being unpopular" and the Outer directed variable "Winning is a priority for me" and between the Inner directed variable "I am primarily motivated by a need for self discovery" and the Outer directed variable "I tend to make sure that other people know all about my successes".

If one examines the variables that correlate significantly with Outer directed variables in a negative direction these are all Inner directed and Sustenance driven value variables.

4.2.1.3 INNER DIRECTED SOCIAL VALUE VARIABLES SIGNIFICANT CORRELATIONS

The following set of tables (4.13 - 4.18) examines each Inner directed social value variable in turn ('Self discovery', 'World', 'Unconventional', 'Voice options', 'Fewrules', 'Alternative') and identifies both positive and negative significant correlations (<0.5) between that variable and the other seventeen social value variables included in the screening instrument.

 Table 4.13
 Inner directed value variable "I am primarily motivated by a need for self discovery"

Discover		s		S		s		s
		$ \mathbf{v} $		$ \mathbf{v} $		v		v
		g		g		g		g
+ve correlation	unconve	I	voiceops	I	alternat	I	mysucces	0
-ve correlation								

3 out of a possible 5 +ve correlation

Table 4.14 Inner directed value variable "I am largely unconcerned about the opinion of me held by the world at large"

World		S		S
		v		v
		g		g
+ve correlation	unconve	I	fewrules	I
-ve correlation	secure	S		

2 out of a possible 5 +ve correlation

Table 4.15 Inner directed value variable "I would describe myself as fairly unconventional"

Unconve		S		S		s		S		S
		v		$\int \mathbf{v}$	·	v		v		v
		g		g		g		g		g
+ve correlation	discover	1	world	I	voiceops	I	fewrules	I	alternat	I
-ve correlation	secure	S	nochange	S	trad	S	appear	О		

4 out of a possible 5 +ve correlation

Table 4.16 Inner directed value variable "I voice my opinions even if it means disagreeing with others or being unpopular"

Voiceops		s		S		S		S
		v		v		v		v
		g		g		g		g
+ve correlation	discover	I	unconve	I	winning	0	alternat	I
-ve correlation	nochang	S	trad	S	contacts	S		

3 out of a possible 5 +ve correlation

Table 4.17 Inner directed value variable "I work most effectively where there are few rules and little structure"

Fewrules		S		S		S
		$ \mathbf{v} $		v		v
		g	,	g		g
+ve correlation	world	I	unconven	I	alternat	I
-ve correlation						T

3 out of a possible 5 +ve correlation

Table 4.18 Inner directed value variable "I believe groups with alternative values and lifestyles are a positive force in society"

Alternat		s		s		s		S
		v		v		v		v
	:	g		g		g		g
+ve correlation	discover	I	unconven	I	voiceops	I	fewrules	I
-ve correlation	nochange	S	setways	S	trad	S	appear	0

⁴ out of a possible 5 +ve correlation

SUMMARY: INNER DIRECTED VALUE VARIABLES

The tables identify that Inner directed variables significantly and positively correlate with the majority of other Inner directed value variables. There is significant positive correlation between the Inner directed variable "I am primarily motivated by a need for self discovery" and Outer directed variable "I tend to make sure that other people know all about my successes" and between the Inner directed variable "I voice my opinions even if it means disagreeing with others or being unpopular" and the Outer directed variable "Winning is a priority for me". These are the only variables other than Inner directed ones that have a significant positive correlation with Inner directed variables. If one examines the variables that correlate significantly with Inner directed variables in a negative direction these are all Sustenance driven and Outer directed ones. Examination of Appendix 2.1 illustrates that whilst the majority of Inner directed variables correlate positively with each other there are two exceptions. The variables, "I am largely unconcerned about the opinion of me held by the world at large" and "I am primarily motivated by a need for self discovery" correlate negatively as do the variables "I am largely unconcerned about the opinion of me held by the world at large" and "I believe groups with alternative values and lifestyles are a positive force in society". Whilst this correlation is not significant it suggests that the variable "I am largely unconcerned about the opinion of me held by the world at large" is not the best variable with which to identify Inner directed consumers.

4.2.1.4 SUMMARY OF ANALYSIS OF THE SCREENING INSTRUMENT

There are no significant positive correlations between Sustenance driven and Inner directed value variables, indicating that there is no overlap between these two groups. There is however an indication of overlap between Inner directed and Outer directed value variables and Sustenance driven and Outer directed variables. Importantly these overlaps are very slight compared to the high degree of correlation between the variables within the Social Value Groups. This interestingly supports MacNulty's (1985) suggestion that the three social value groups can be positioned in the style of Maslow's hierarchy of needs with Sustenance driven at the bottom, Outer directed in the middle and Inner directed at the top.

The results of the analysis for the screening instrument do on the whole confirm its robustness as a tool to identify the values of the three social value groups Sustenance driven, Outer directed and Inner directed.

4.2.2 GENERATION OF SOCIAL VALUE SCORE PROFILES FOR HOMOGENEOUS SAMPLE

Confident that the screening instrument was operating as expected across a large number of respondents, the next stage was to score each respondent according to their responses to each of the Social Value group variables. This was done on the same basis as in the pilot (described in 3.2.4) and is reiterated below.

The responses of all 215 respondents in the homogeneous sample were examined, each response to each of the 18 social value variables (6 Inner directed value statements, 6 Outer directed value statements and 6 Sustenance driven value statements) being given a score of +2, +1, 0, -1 or -2, this scoring system was used rather than 1-5 so to generate negative scores, these would be a clear indication that the respondent disagreed with the social value statement. The scores for each set of six social values were then added

together. In this way, three scores were generated for each set of values for each respondent, the first indicating the degree to which the respondent was Sustenance driven, a second indicating the degree to which the respondent was Outer directed and the third indicating the degree to which the respondent was Inner directed. Appendix 5 illustrates these score profiles for all 215 members of the homogeneous group.

If we examine the scores profiles (Appendix 5) one can clearly identify a number of respondents who have no dominant values for example respondents 8, 74 and 89. There are also respondents that exhibit two sets of values whilst disagreeing with the third, for example, respondent 139 exhibits both Inner directed and Outer directed social values. Importantly there are also respondents who demonstrate a dominant set of values.

The identification of respondents who clearly exhibited one dominant set of values was undertaken to investigate differences in consumer perceptions of food risk and risk relieving strategies for the three social value groups, Sustenance Driven, Outer Directed Inner directed (objective 3 and 4 Chapter One) and to measure the statistical significance of those differences through an analysis of variance. Similarly these groups would provide the sample for Interview Three to further investigate food risk perceptions and risk relieving strategies for Inner directed, Outer directed and Sustenance driven respondents (analysed in Chapter Five).

Once all the respondents had been given a score profile (Appendix 5) a mechanism needed to be devised which could be used to select an equal number ¹ of dominantly Sustenance driven, Outer directed and Inner directed respondents. This was achieved by considering mean and standard deviation (SVG) scores for all 215 respondents (see Table 4.19). These formed the basis for deriving a set of rules (See section 4.2.2.1) for allocating respondents to the most appropriate Social Value Group. The members of the three dominant groups selected in this way are detailed in Table 4.20.

¹ An equal number of respondents were needed in each group to improve reliability of the analysis of variance subsequently undertaken as described by Greene and D'Oliveira (1995)

Table 4.19 Means and Standard Deviations for Social Value Group Profile
Scores for Homogeneous Group of 215

	Inner directed	Outer directed	Sustenance driven
Mean	+1.72	87	- 1.21
Standard Deviation	3.30	4.17	3.56
68 % of scores were in the range of	-1.58 - +5.01	-5.05 - +3.30	-4.76 - +2.35

As can be seen in Table 4.19 above the Inner directed mean SVG score was the highest of the three perhaps due to a positive bias of responses to the Inner directed value statements. This bias was compensated for in the conditions used to select the 18 most dominant Sustenance driven, Outer directed and Inner directed respondents (see section 4.2.2.1).

4.2.2.1 CONDITIONS FOR THE IDENTIFICATION OF DOMINANTLY SUSTENANCE DRIVEN, OUTER DIRECTED AND INNER DIRECTED RESPONDENTS

To be labelled as dominantly Sustenance driven the respondent had to fulfil all of the following criteria:-

- 1. have a Sustenance driven score that is no less than 3
- 2. have an Outer directed score which is a minimum of 3 points lower than that of the Sustenance score
- 3. have an Inner directed score which is a minimum of 3 points lower than that of the Sustenance driven score
- 4. have no more than one -1 (disagree) response or 2 x 0 (neither agree or disagree) responses to any Sustenance driven social value.

To be labelled as dominantly Outer directed the respondents had to fulfil all of the following criteria:-

- 1. have a Outer driven score that is no less than 3
- 2. have an Sustenance driven score which is a minimum of 3 points lower than that of the Outer directed score
- 3. have an Inner directed score which is a minimum of 3 points lower than that of the Outer directed score
- 4. have no more than one -1 (disagree) response or 2 x 0 (neither agree or disagree) responses to any Outer directed social value.

To be labelled as dominantly Inner directed the respondents had to fulfil all of the following criteria:

- 1 have an Inner directed score that is no less than 6
- 2. have an Sustenance driven score which is a minimum of 6 points lower than that of the Inner directed score
- 3. have an Outer directed score which is a minimum of 6 points lower than that of the Inner directed.
- 4. have no more than 1 x 0 (neither agree or disagree) response to any Inner directed social value.

To be labelled as a member of the non dominant group respondents had to :-

1. have a score profile that did not fit with any one of the other three categorises.

Thus the 18 most dominant Outer directed, Inner directed and Sustenance driven respondents were selected and coded as such, the rest of the 215 sample were coded as members of the non dominant group.

Table 4.20 Social Value Score (SVG) Profiles for Dominant Groups
(Sample of (54) 18 Sustenance driven, 18 Outer directed, 18 Inner directed)

Sı	ıstenan	ce Dri	ven	(Outer 1	Directe	d	I	nner I	Directed	<u> </u>
ID	Sust	Inner	Outer	ID	Sust	Inner	Outer	ID	Sust	Inner	Outer
	S/c	S/c	S/c		S/c	S/c	S/c		S/c	S/c	S/c
150	5	-2	-5	1	-3	2	5	. 81	-7	10	-7
19	7	-1	0	3	0	1	4	153	-2	7	-6
164	5	-1	-1	75	-2	2	6	83	-7	7	-3
99	5	-3	-4	10	0	3	6	85	-5	5	-2
175	6	-4	2	154	4	-1	11	158	-4	7	-4
41	5	2	-1	95	1	0	7	162	-6	6	-4
43	8	-7	-5	25	2	-1	5	163	-4	9	-4
187	4	0	-2	28	1	1	6	166	-1	6	-9
197	5	-6	-4	184	1	1	5	24	-4	6	0
54	7	3	-2	118	-2	3	7	170	-4	9	-5
201	5	1	-2	50	-6	2	8	35	-4	6	-6
202	6	2	1	62	-3	-3	6	46	-4	9	-4
203	5	-1	-1	134	-2	1	4	193	0	6	2
206	4	0	-2	135	-2	1	6	198	-3	8	-6
209	7	3	-2	205	-6	0	9	200	-2	8	-7
138	4	-5	1	207	-3	2	5	129	-7	7	-7
212	7	-1	0	211	-4	1	7	63	-2	9	-7
58	4	-1	-2	144	-2	3	6	68	-5	6	-4

(Appendix 5 details the Score Profile for total homogeneous sample of 215)

Whilst the decision making process and the conditions for identification of the 54 respondents were, from a positivistic perspective rather crude, one was confident that the respondents chosen where indeed dominantly Sustenance driven, Outer directed or Inner directed. This broad approach was later supported by Earle and Cvetkovich (1999) who adopted a similar strategy for categorising respondents in their research and is discussed further in Chapter Six.

4.3 ANALYSIS OF FOOD RISK PERCEPTION OUESTIONNAIRE (Part Three of the Questionnaire)

Analysis of the results of Part Three of the questionnaire aimed to investigate how social values influenced food risk perceptions (research objective 3 Chapter One).

4.3.1 EXAMININATION OF FOOD RISK PERCEPTION MEAN SCORES

Table 4.21 details the mean scores for the Food Risk Perception variables for the homogeneous sample of 215 respondents as a whole and for the dominant sample of 18 Sustenance driven, 18 Inner directed and 18 Outer directed respondents.

Importantly responses were expressed using a Likert scale of 1-9, 5 being the midpoint on the scale, 9 indicating the highest level of perceived risk. The means are presented in descending order those at the top of the table indicating the highest level of perceived risk, those at the bottom of the table the lowest perceived risk.

 Table 4.21
 Food Risk Perception Variable Mean Scores

Part 3	Mean Score	Part 3	Mean	Part 3	Mean	Part 3	Mean
	for sample		Score		Score		Score
	of 215		18		18		18
			Susten		Inner		Outer
pestfert	7.52	pestfert	8,06	pestfert	8,50	highfat	7.94
	<u> </u>		7.33	informis	8.06	informis	7.67
highfat	7.38	exppack	7.55	intornis		miormis	
informis	7.31	informis	7.28	controv	7.94	pestfert	7.06
controv	6.83	highfat	7.22	highfat	7.39	highsalt	6.89
artipres	6.44	BSE	7.11	p3geneng	7.33	liminfo	6.61
liminfo	6.37	p3geneng	6.94	потгерас	7.33	artipres	6.44
BSE	6.29	artipres	6.67	exppack	7.18	p3geneng	6.39
p3geneng	6.15	liminfo	6.61	prepared	7.17	controv	6.33
artflav	6.12	highsalt	6.44	BSE	7.06	BSE	6.28

notrepac	5.97	controv	6.44	artipres	6.83	lowfibre	6.28
lowfibre	5.93	expsup	6.33	liminfo	6.72	artflav	6.11
artcolor	5.76	artflav	5.94	highsalt	6.61	notrepac	5.83
exppack	5.59	prepared	5.89	artcolor	6.28	artcolor	5.44
prepared	5.17	artcolor	5.56	artflav	6.22	throwbbd	4.94
throwbbd	5.08	cheapsup	5.33	throwbbd	5.83	imagesup	4.94
cheapsup	4.67	throwbbd	5.22	lowfibre	5.56	exppack	4.78
artsweet	4.50	artsweet	5.00	artsweet	5.18	cheapsup	4.61
nonorg	4.32	expbrand	4.89	expsup	5.06	cheapbra	4.59
imagesup	4.13	lowfibre	4.83	nonorg	4.89	difbrand	4.39
cheapbra	4.01	notrepac	4.72	expbrand	4.30	artsweet	4.28
expsup	3.80	difbrand	4.39	imagesup	3.88	prepared	4.06
difbrand	3.66	unfaming	3.72	cheapsup	3.78	nonorg	3.94
expbrand	3.50	nonorg	3.56	eatout	3.39	expbrand	3.67
highsalt	3.38	cheapbra	3.33	cheapbra	3.29	difsup	3.22
unfaming	3.06	imagesup	3.22	unfaming	3.00	unfaming	3.22
eatout	2.52	eatout	2.72	difbrand	2.94	expsup	2.56
difsup	2.51	ownlabel	2.33	ownlabel	2.29	eatout	2.11
ownlabel	2.33	difsup	2.06	difsup	2.22	ownlabel	2.06

If one examines the top five food risk perceptions for the 18 Sustenance Driven, 18 Inner Directed and 18 Outer Directed respondents, 'pesticides, fertilisers, veterinary drug residues in foods', 'Information on packaging that may be misleading' and 'High fat diet' are all present. Interestingly these are the top three food risk perceptions for the total homogeneous sample of 215 indicating that these are broad common food risk perceptions.

A more in-depth examination of these mean scores begins to identify differences between the Sustenance driven, Outer directed and Inner directed Social Value Groups. The Outer directed mean scores are on the whole lower that those for the Inner directed or Sustenance driven groups. Inner directed mean scores are on the whole the highest. 'High fat diet' is the number one perceived risk for Outer directed respondents whilst, 'pesticides, fertilisers, veterinary drug residues in foods' is number one perceived risk for the Inner directed and the Sustenance driven groups.

Differences between the Food Risk Perceptions of the Sustenance driven, Outer directed and Inner directed groups needed to be confirmed using Analysis of Variance. Similarly respondents motivations for their food risk perceptions needed to be investigated, for example, whilst the mean scores indicate that Inner directed and Sustenance driven respondents share a concern for 'pesticides, fertilisers, veterinary drug residues in foods' the rationale for this should not be assumed to be the same, Interviews (Chapter Five) would be used to explore this further.

4.3.2 ONE WAY ANALYSIS OF VARIANCE (Independent Groups)

Analysis of Variance was used to further investigate the food risk perception mean scores for the 18 Sustenance driven, 18 Outer directed and 18 Inner directed respondents. It was believed that any differences in food risk perceptions could be attributed to differences in the social values identified by the screening instrument (Part Two).

The differences between mean scores (Table 4.21) were examined. An asterisk indicates significant differences at <0.05 between the food risk perception mean scores of the three Social Value Groups. The standard Post Hoc significant test (LSD) was used along side Tukey's HSD (test a more conservative and stringent test) to identify where the significant differences lie.

Table 4.22 Results of the Analysis of Variance for Food Risk Perception

Questionnaire

		Mean	-	Sig	dif u	sing	Sig dif using			
		Score					Tukey's HSD			
Food Risk	Susten	I/O	I/S	S/O	I/O	I/S	S/O			
Perception										
Variable					}					
p3geneng	6.94	7.33	6.39							
expbrand	4.89	4.29	3.67							
exppack	7.33	7.18	4.78	*		*	*		*	

BSE	7.11	7.06	6.28						
prepared	5.89	7.17	4.0556	*		*	*		
pestfert	.8.06	8.50	7.06	*		 	*		
cheapsup	5.33	3.78	4.61						
highsalt	6.44	6.6111	6.89						
artipres	6.67	6.83	6.44						
highfat	7.22	7.39	7.94						
informis	7.28	8.06	7.67						
notrepac	4.72	7.33	5.83		*			*	
difbrand	4.39	2.94	4.39	*	*				
throwbbd	5.22	5.83	4.94						
artsweet	5.00	5.18	4.28						
imagesup	3.22	3.88	4.94			*		-	
artcolor	5.56	6.28	5.44						
ownlabel	2.33	2.29	2.06						
liminfo	6.61	6.72	6.61						
artflav	5.94	6.22	6.11						
eatout	2.72	3.39	2.11						
cheapbra	3.33	3.29	4.59						
lowfibre	4.83	5.56	6.28						
difsup	2.06	2.22	3.22						
nonorg	3.56	4.89	3.94						
unfaming	3.72	3.00	3.22						
controv	6.44	7.94	6.33	*	*		*		
expsup	6.33	5.06	2.56	*		*	*		*
 									

Table 4.23 Summary Table

Results of the Analysis of Variance for Food Risk Perception

Questionnaire. Significant Differences Only (Data extracted from Table 4.22)

	Mean Scores			Sdu	Sd using LSD			Sd using		
				!			Tukey's HSD			
	Susten	Inner	Outer	I/O	I/S	S/O	I/O	I/S	S/O	
exppack	7.33	7.18	4.78	*		*	*		*	
prepared	5.89	7.17	4.06	*		*	*			
pestfert	8.06	8.50	7.06	*			*			
notrepac	4.72	7.33	5.83		*			*		
difbrand	4.39	2.94	4.39	*	*					
imagesup	3.22	3.88	4.94			*				
controv	6.44	7.94	6.33	*	*		*			
expsup	6.33	5.06	2.56	*		*	*		*	

Table 4.23 highlights significant differences between the food risk perceptions of the Sustenance driven, Outer directed and Inner directed groups. Significant differences were identified between the means of the Sustenance group and the Outer directed group and similarly between the Inner directed group and the Outer directed group for 'Pay more for expensively and attractively packaged food', 'paying more for pre prepared vegetables' and 'using an expensive food supermarket'. Whilst mean scores for the Sustenance driven group and the Inner directed group are higher than the Outer directed group it is impossible to be sure of the motivation for these perceptions. Interestingly significant differences were identified between the means of the Sustenance driven group and the Inner directed group for 'Food packaging which is not recyclable' and between the means of the Inner directed group and Outer directed group for 'Pesticides, fertilisers, veterinary drug residues in foods' in both cases the Inner score being the higher. In contrast the Inner directed mean score for 'having to buy a different or new brand because your usual brand is out of stock' was significantly lower than that for the Outer directed and Sustenance driven groups. Mean scores for the variable 'foods grown or/and produced by a company or country that operate controversially' were above the mid way point for all three Social Value Groups nevertheless significant differences

were identified between Inner directed and Outer directed and Inner directed and Sustenance driven groups.

Results indicate that there are indeed differences between the Sustenance driven, Outer directed and Inner directed food risk perceptions.

The internal relationships between the food risk perception variables are explored via Factor Analysis and Pearson Correlations (Appendix 2.2 (54) and 3.2 (215)).

4.3.3 FACTOR ANALYSIS

The correlation matrix for Part Three of the questionnaire was explored in great length (Full matrix in Appendix 3.2 for the sample of 215, Appendix 2.2 for the sample of 54). The matrix provided an invaluable insight into the relationships between the food risk perception variables. The factor analysis was used to simplify these with a view to identifying underlying themes.

Varimax rotation was applied to aid the interpretation of the principal components (factors). The analysis was conducted on the total homogeneous sample of 215, the number of respondents in the dominant sample (54) was far too small to carry out the analysis.

As suggested by Piggott and Sharman (1986) cited in Hair (1995) only those factors with an eigenvalue of greater than one were retained in the final solution. Factor loadings of less than 0.5 were initially suppressed and then at 0.4 and 0.3 with a view to exploring and optimising the interpretation of the results.

4.3.3.1 RESULTS OF THE FACTOR ANALYSIS

The Kaiser - Meyer - Olkin (KMO) measure of sampling adequacy was .74603 i.e. greater than the 0.5 regarded as satisfactory for factor analysis to proceed.

A Barlett test of Sphericity of 2061.7348 (significance of 0.00000), indicated that the correlation matrix was suitable for factor analysis.

The following table 4.24, shows the nine factors extracted and the loading of the original variables on those factors. The Table illustrates the results of the factor analysis when loadings were suppressed at the levels of 0.5, 0.4 and 0.3 respectively.

Table 4.24 Nine Factor Solution for Food Risk Perception Variables

1	2	3	4	5	6	7	8	9
Technology	High Cost	Nutrition /	Cheap	Ecology / Ethics	Unfamiliar	Unfamiliar	BSE	
Artflav	Exppack	Highfat	Cheapsup	Notrepac	Unfaming	Difbrand	BSE	Throwbbd
.80864	.80179	.83993	.79978	.67159	.69368	.80150	.88641	.78256
Artcolor	Expbran	Highsalt	Cheapbra	Pestfert	Nonorg			Lowfibre
.78926	.78320	.75059	.76353	.58247	.59547			.54949
Artipres	Expsup	Informis	Imagesup	Controv	Difsup			
.70905	.74782	.63119	.71824	.63011	.54038			
Artsweet	Prepared		Ownlabel					
.6 7 987	.70194		.54146					
Geneng	Eatout							
.62558	.51715		1					
Pestfert		Lowfibre		Prepared		Ownlabel		
.44353		.49522		.4 0 900		.42145		
Liminfo		Artipres	7					
.41142		.408 3 4						
Nonorg		Liminfo	Difsup	Eatout	Liminfo	Geneng		Liminfo
.34785	į	.33937	.36658	.38213	.33391	.38422		.36254
Notrepac						Difsup		
.31794					5	. 3 6919	•	
Controv						Infomis		
.30418						.30526		

Nine factors were identified which accounted for 67.8% of the variance in the overall data, these are identified and illustrated in table 4.24. The first factor was highly

associated with 'Technology', the second with 'High Cost', the third with 'Nutrition / Labelling', the fourth with 'Cheap', the fifth with 'Ecological / Ethical issues', the sixth and seventh with 'Unfamiliarity', the eighth with 'BSE'. The KMO measure of sampling adequacy for the variable BSE was, however low at 0.566, similarly the KMO measure of sampling adequacy for the variable 'throwbbd' was 0.556. Though the lowest of the KMO scores for all the variables included in the analysis these levels are acceptable (above 0.5) and as such were retained. Interestingly however, if these two variables are removed from the factor analysis 'low fibre' falls into factor 3 'nutrition'.

Labels were given to each of the factors after the composites had been examined.

4.3.3.2 RESULTS OF THE ANALYSIS OF VARIANCE FOR FACTOR SCORES FOR THE DOMINANT SAMPLE OF 54 (18 SUSTENANCE DRIVEN, 18 OUTER DIRECTED AND 18 INNER DIRECTED)

The differences between mean factor scores for the three were examined. An asterisk indicates the position of significant difference at <0.05 between the food risk perception mean factor scores of the three Social Value Groups. The standard Post Hoc significant test (LSD) was used along side Tukey's HSD test (a more conservative and stringent test) to identify where the significant differences lie.

Table 4.25

	Me	an Scores		Sd u	sing	LSD	So	l usin	g
							Tuk	ey's F	ISD
Factor	Susten	Inner	Outer	I/O	I/S	S/O	I/O	I/S	S/O
1 Artificial		· · · · · · · · · · · · · · · · · · ·							ļ —
2 High Cost	.8400	.6638	3551	*		*	*		*
3 Nutrition		· · · · · · · · · · · · · · · · · · ·							
4 Cheap				1					
5 Eco/Ethics	200	.6425	2876	*	*		*		
6 Unfamiliar									
7 Unfamiliar		,							

Factor 2 'High Cost' and Factor 5 'Ecological / Ethics' are the only two factors where significant difference lie. Interestingly the three groups demonstrate quite different profiles. These results support those presented in Table 4.23.

4.3.4 INVESTIGATION OF THE RELATIONSHIP BETWEEN SOCIAL VALUE VARIABLES AND FOOD RISK PERCEPTIONS VARIABLES

The following results are based on the Pearson Correlations for the sample of 54 which express the relationships between the individual social value variables (screening instrument Part Two of the questionnaire) and the individual food risk perception variables (Part Three of the questionnaire) (Appendix 2.4 details the full correlation matrix).

Plus and minus signs illustrate positive and negative significant correlations (< 0.05) between social value variables and food risk perception variables.

Tables 4.26 - 4.31 present the results of this analysis. Each table focusing on one of the 7 factors identified in Table 4.24, (factors 6 and 7 are combined and factor 9 is omitted).

A detailed examination of the relationships between the scores for the individual social value group variables (Part Two) and those for the individual food risk perception variables (Part Three) for the homogeneous sample of 215 respondents was also conducted. Appendix 6 presents the results of this analysis. Appendix 3.4 details the full correlation matrix for the homogeneous sample of 215. Importantly the results for both sets of analysis are comparable and supportive.

Table 4.26 Factor 1 Technology (man made additions to foodstuffs) Risks

,	Social Value	Genetic	Artificial	Artificial	Artificial	Artificial
	Group	Engineering	Preservative	Sweeteners	colours	Flavours
Status	Outer	_			_	_
Mysuccess	Outer		_			
Discovery	Inner	+		+	+	+
Fewrules	Inner				+	
Alternat	Inner	+	+	+	+	+

Results indicate that Inner directed values positively correlate with concern over technological risks. In sharp contrast Outer directed values are negatively correlated suggesting that an increase in Outer directedness seems to be associated with decrease in concern about such risks.

Table 4.27 Factor 2 High Cost as a Risk

	Social Value	Expensive	Expensive	Paying more	Eating	Expensive
	Group	Brand	Packaging	for Pre	out	Store
				prepared food		
Trad	Susten				~	
Secure	Susten	+				
Nochange	Susten		+			+
Setways	Susten					+
Holdlife	Susten	+	+			
Appear	Outer		_	-	-	
Status	Outer				-	-
Best	Outer		_			_
Winning	Outer	_	_			-
Mysuccess	Outer		-			
Ahead	Outer		_	-		_
Alternat	Inner		+	+		
World	Inner			+		
Unconven	Inner				+	

This set of results are consistent and strong. There is a positive correlation between Inner directed and Sustenance driven values and high cost as a perceived risk (interestingly, results of Interview Two indicated that whilst Sustenance driven

consumers associated high cost with financial risk, Inner directed consumers viewed high cost as unnecessary and wasteful). In sharp contrast to the Inner directed and Sustenance driven values, all six Outer directed value variables have a significantly negative correlation with one or more of the high cost variables.

Table 4.28 Factor 3 Nutritional Risks

	Social Value	High Salt	High Fat	*Misleading
	Group			Information
Contacts	Susten		-	_
Alternat	Inner	+		+
Voiceops	Inner			+
Mysuccess	Outer			
Best	Outer	+	+	
Status	Outer		+	

^{*} Results of the factor analysis indicate that there is an association between misleading information and nutrition, perhaps via the issue of nutritional labelling.

It appears that the values most associated with nutrition per se are the Outer directed ones. The results indicate that Inner directed concern is in the context of misleading information.

Table 4.29 Factor 4 Cheap (Image Associated) Risks

	Social Value	Cheap Store	Image of	Own label	Cheap Brand
	Group		Store		
Secure	Susten			_	
Setways	Susten		-		
Trad	Susten	+			
Contacts	Susten		_		
Nochange	Susten	+			
Best	Outer		+		
Ahead	Outer		+		+
Winning	Outer				+
World	Inner	_			

Outer directed value variables correlated positively with the food risk 'cheap' variables.

This relationship is not however as convincing as was expected. Similarly there is some

indication of a negative correlation between this dimension and the values of the Sustenance driven and Inner directed groups though again this is not as persuasive as might have been anticipated.

Table 4.30 Factor 5 Ecological (conservationist / environmental) Risks

	Social Value	Pesticides /	Non	Controversial
	Group	Fertilisers	recyclable	operations
			Packaging	
Contacts	Susten		-	
Setways	Susten			
Nochange	Susten			
Winning	Outer	_		_
Status	Outer	_		
Ahead	Outer	-	+	
Mysuccess	Outer			
Alternat	Inner	+	+ .	+
Voiceops	Inner		+	
Fewrules	Inner			+
Discovery	Inner	+	+	+

A strong set of results indicating that the more a consumer exhibits Inner directed values the more likely it is that they will perceive ecological and environmental risks as a concern (Inner directed consumers have environmental issues embedded deep in their personal philosophy). In contrast the results suggest Outer directed and Sustenance directed values correlate negatively with environmental / ecological concern.

Table 4.31 Factor 6/7 Unfamiliarity as Risk

	Social Value	Different	Unfamiliar
	Group	Brand	ingredients
Hold	Susten	+	
Trad	Susten	+	+
Secure	Susten		
Nochange	Susten		+
Appear	Outer	+	
World	Inner	-	
Fewrules	Inner	-	

Whilst this set of results presents an unclear picture for the Outer directed set of variables there is some indication of a negative correlation for the Inner directed ones. There is a relatively clear indication that 'unfamiliarity' is an increasing concern the more the respondent exhibits Sustenance driven values. (Unfamiliarity is a big issue for Sustenance driven consumers, change and unfamiliarity threatens their stability and makes them feel insecure).

Factor 8 BSE Risk

Inconclusive results may have been the result of this research occurred at the height of the BSE hysteria and in the midst of much confusion over the scare.

This set of results is summarised in section 4.3.4.1

4.3.4.1 INVESTIGATION OF THE RELATIONSHIP BETWEEN SOCIAL VALUE GROUP SCORES AND FOOD RISK PERCEPTIONS VARIABLES (215 respondents)

The following table identifies the results of the Pearson correlation (using the sample of 215) between respondents' Sustenance driven score, Inner directed score, Outer directed score and their scores for each risk relieving strategy variable. All positive (+ve) and negative (-ve) correlations identified in Table 4.32 are significant (<0.05).

Table 4.32 Results of Pearson Correlation for Social Value Group (SVG) Score and Food Risk Variable Score for sample of 215

Sustenance	F	Sustenance	F	Outer	F	Outer	F	Inner	F	Inner	F
Driven		Driven		Directed		Directed		Directed		Directed	
+ve		-ve		+ve		-ve		+ve		-ve	
correlation		correlation		correlation		correlation		correlation		correlation	
Expensive	2	Artificial	1	Cheap	4	Expensive	2	Artificial	1		
Supermarket		Colouring		Supermarket		Packaging		Sweeteners			
Expensive	2	Artificial	1	Cheap	4	Expensive	2	Artificial	1	 	
Brand		Flavouring		Brand		Supermarket		Colouring			
Expensive	2	Artificial	1	Image of	4	Pre prepared	2	Artificial	1		
Packaging		Preservative s		Supermarket				Flavouring			
Unfamiliar	6	Non	5	Different	6	Eating Out	2	Expensive	2		-
Ingredients	/	Recyclable		Brand	7	. .		Packaging		,	
	7	Packaging				•					
Different	6	Controversia	5			Non	5	Pre prepared	2		
Brand	1	1 Operations				Recyclable					
	7					Packaging					
		Low Fibre	9			Pesticides	5	Expensive	2		\Box
						and		Supermarket			
						Fertilisers					
						Controversia	5	Eating Out	2		├-
						1 Operations					
								Controversia	5		
								1 Operations			
								Pesticides	5		
					I			and			
								Fertilisers			
								Non	5		
								Recyclable			
			_					Packaging			
							\neg	Non Organic	6		

F = Factor illustrated in Table 4.24

Table 4.32 identifies that the higher the Sustenance driven score the higher the concern for risk associated with 'high cost', and 'the unfamiliar' and the less concern there is for risks associated with 'technology' and 'ecology / ethics'. The higher the Outer directed score the more concern there is for risks associated with 'cheap/image' and the less concern there is for risks associated with 'high cost' and 'ecology / ethics'. Whilst the Sustenance driven and the Outer directed scores both negatively correlate with 'ecology / ethics' related risks, they correlate in opposite directions with 'high cost' and 'cheap/image' related risk. The higher the Inner directed score the more concern there is for 'technological', 'high cost' and 'ecological / ethics' related risks. Interestingly and importantly no food risk perception variables significantly correlated in a negative direction with the Inner directed score, implying that the more Inner directed the score the less likely it is that risks will not be perceived.

4.4 ANALYSIS OF RISK RELIEVING STRATEGIES OUESTIONNAIRE (Part Four Of The Questionnaire)

Analysis of the results of Part Four of the questionnaire aimed to investigate how social values influenced risk relieving strategies (research objective 4 Chapter One).

4.4.1 EXAMININATION OF MEANS

Table 4.33 details the mean scores for the Risk Relieving Strategy variables for the dominant sample of 18 Sustenance Driven, 18 Inner Directed and 18 Outer Directed respondents. These scores are presented next to the mean score for the total homogeneous sample of 215 respondents.

Importantly responses were expressed using a Likert scale of 1-9, 5 being the midpoint on the scale, 9 indicating the highest level at which the risk relieving strategy was effective. The means are presented in descending order those at the top of the table indicating that the strategy is highly effective at relieving risk and those at the bottom are least effective at relieving risk.

Table 4.33 Risk Relieving Strategy Variable Mean Scores

Part 4	Mean	Part 4	Mean	Part 4	Mean	Part 4	Mean
	Score for		Score		Score		Score
	sample		18		18		18
	of 215		Susten		Inner		Outer
indeapro	6.34	fambrand	6.00	indeapro	6.61	indeapro	6.61
comlabel	5.85	indeapro	5.17	comlabel	6.50	brandrep	6.44
fambrand	5.37	brandrep	5.06	fambrand	4.82	fambrand	6.06
brandrep	5.20	comlabel	4.83	brandrep	4.22	comlabel	5.89
expert	4.65	freesamp	4.28	freesamp	3.89	expert	5.72
freesamp	4.14	expert	4.28	expert	3.78	supimage	5.11
supimage	3.80	supimage	4.28	supimage	3.33	govappro	4.94
moneybac	3.55	moneybac	4.22	moneybac	. 3.06	freesamp	4.83
similyou	3.35	similyou	4.06	similyou	3.00	expbra	4.83
govappro	3.27	family	3.89	cheapers	2.89	similyou	4.56
family	3.16	govappro	3.44	family	2.78	expsuper	4.56
expsuper	3.12	cheapers	3.11	expsuper	2.72	moneybac	4.39
expbra	3.07	cheaperb	2.94	cheaperb	2.72	family	4.00
p4advert	2.64	p4advert	2.72	expbra	2.61	p4advert	3.83
cheapers	2.37	celeb	2.56	govappro	2.06	cheapers	3.06
cheaperb	2.25	expsuper	2.33	p4advert	1.67	celeb	2.94
celeb	1.85	expbra	1.94	celeb	1.22	cheaperb	2.67

Interestingly the top four risk relieving strategies are shared by all three Social Value Groups and the total homogeneous sample of 215. These being 'The foodhas been approved by an independent / private testing company', 'Buy the brand that has a good reputation', 'Buy the brand that I have always bought and am familiar with' and 'The food has clear and comprehensive labelling' indicating that overall these are the most effective risk relieving strategies. Examining the Social Value Group risk relieving strategy profiles as a whole one can observe that mean scores are the highest for the Outer directed group and lowest for the Inner directed group.

Six Outer directed mean scores are above the midway point of 5, indicating these are perceived as effective strategies. In contrast, whilst the Inner directed scores for 'The foodhas been approved by an independent / private testing company' and 'The food has clear and comprehensive labelling' are convincingly above the midway point of a

mean score of 5 all other strategies are under the midway point indicating perhaps that these are the only two strategies which are effective risk relieving strategies for the Inner directed group. The Sustenance driven group identify 'Buy the brand that I have always bought and am familiar with' as a clearly perceived effective strategy. 'The foodhas been approved by an independent / private testing company' and 'Buy the brand that has a good reputation' both follow at the midway point.

4.4.2 ONE WAY ANALYSIS OF VARIANCE (Independent Groups)

Analysis of Variance was used to further investigate the risk relieving strategies mean scores, for the 18 Sustenance driven, 18 Outer directed and 18 Inner directed respondents. It was believed that any differences in perception of effective risk relieving strategies could be attributed to differences in the social values identified by the screening instrument (Part Two). The differences between these mean scores were examined. An asterisk indicates significant differences at <0.05 between the risk relieving strategy mean scores of the Sustenance driven, Outer directed and Inner directed Social Value Groups. The standard Post Hoc significant test (LSD) was used along side Tukey's HSD test (a more conservative and stringent test) to identify where the significant differences lie.

Table 4.34 Results of the Analysis of Variance for Food Risk Relieving
Strategies

		Mean Scores			Sif dif using LSD			Sig dif using			
								Tukey's HSD			
	Susten	Inner	Outer	I/O	I/S	S/O	I/O	I/S	S/O		
p4advert	2.72	1.67	3.83	*		*	*				
similyou	4.06	3.00	4.56	*							
freesamp	4.28	3.89	4.83								
fambrand	6.00	4.82	6.06				-				
govappro	3.44	2.06	4.94	*		*	*				
expert	4.28	3.78	5.72	*							
brandrep	5.06	4.22	6.44	*			*				

family	3.89	2.78	4.00						
cheapers	3.11	2.89	3.06						
expsuper	2.33	2.72	4.56	*		*	*		*
comlabel	4.83	6.50	5.89		*			*	
celeb	2.56	1.22	2.94	*	*		*		
moneybac	4.22	3.06	4.39						
supimage	4.28	3.33	5.11	*			*		
cheaperb	2.94	2.72	2.67						
expbra	1.94	2.61	4.83	*		*	*		*
indeapro	5.17	6.61	6.61	1	*	*		-	

Table 4.35 Summary of Results of the Analysis of Variance for Food Risk Relieving Strategies (Significant differences only)

		Mean		Sif	dif us	sing	Sig	dif u	sing		
		Scores			LSD			Turkey's HSD			
	Susten	Inner	Outer	1/0	I/S	S/O	I/O	I/S	S/O		
p4advert	2.72	1.67	3.83	*		*	*				
similyou	4.06	3.00	4.56	*							
govappro	3.44	2.06	4.94	*		*	*				
expert	4.28	3.78	5.72	*				-			
brandrep	5.06	4.22	6.44	*			*				
expsuper	2.33	2.72	4.56	*		*	*		*		
comlabel	4.83	6.50	5.89		*			*			
celeb	2.56	1.22	2.94	*	*		*				
supimage	4.28	3.33	5.11	*			*				
expbra	1.94	2.61	4.83	*		*	*		*		
indeapro	5.17	6.61	6.61		*	*					

A high proportion of the risk reliving variables (eleven of the seventeen) display significant differences between the three Social Value Group's mean scores. Many significant differences were identified between the Inner directed and the Outer directed groups. Interestingly far fewer significant differences were identified between Inner

directed and Sustenance driven and Sustenance driven and Outer directed. These differences are presented and discussed further in Table 4.37.

4.4.3 FACTOR ANALYSIS

The correlation matrix for the risk relieving strategies (Part Four of the questionnaire) was explored in great length (Full matrix in Appendix 3.3 for sample of 215, 2.3 for sample of 54). The matrix provided an invaluable insight into the relationships between the risk relieving strategies. Factor analysis was used to simplify these with a view to identifying underlying themes.

Varimax rotation was applied to aid the interpretation of the principal components (factors). The analysis was conducted on the total homogeneous sample of 215, the number of respondents in the dominant sample (54) was far too small to carry out the analysis.

As suggested by Piggott and Sharman (1986) cited in Hair (1995) only those factors with an eigenvalue greater than one were retained in the final solution. Factor loadings of less than 0.5 were initially suppressed and then at 0.4 and 0.3 with a view to exploring and optimising the interpretation of the results.

4.4.3.1 RESULTS OF THE FACTOR ANALYSIS

The Kaiser - Mayer - Olkin (KMO) measure of sampling adequacy was .78421 i.e. greater the than 0.5 regarded as satisfactory for factor analysis to proceed.

A Barlett test of Sphericity of 1658.8794 (significance of 0.00000) indicated that the correlation matrix was suitable for factor analysis.

Table 4.36, shows the five factors extracted and the loadings of the original variables on those factors. The Table illustrates the results of the factor analysis when loadings are suppressed at levels of 0.5, 0.4 and 0.3 respectively.

Table 4.36 Five Factor Solution for Food Risk Relieving Strategy Variables

1	2	3	4	. 5
Societal	Low Cost	Branding	High Cost/	Authoritarian
Acceptance /		(Product	Image	Reassurance
Endorsement		Orientation)		
Advert	Cheapers	Comlabel	Expsuper	Expert
.73495	.85109	.74397	.82656	.79993
Similyou	Cheaperb	Fambrand	Expbra	Indeapro
.71014	.84605	.609 2 9	.88348	.73165
Family	Moneybac	Brandrep	Supimage	Govappro
.56154	.61463	.59314	.61936	.58688
Celeb		Freesamp		
.53960		.58732		
Govappro	Celeb	Moneybac		
.49482	.47736	.42934	•	
	Family	Indeapro		·
	.42189	.40115		·
Freesamp		Supimage	Brandrep	
.39489	-	.39856	.30032	
Brandrep				
.38520				
Fambrand				
.33816				
Expert				
.33357				

Five factors were identified which accounted for 67.9% of the overall variance in the data. The first factor was highly associated with 'Societal Acceptance / Endorsement' the second with 'Low Cost', the third with 'Branding (product orientation)' the fourth with

'High Cost / Image' and the fifth with 'Authoritarian Reassurance'. Labels were given to each factor after the composites had been examined.

4.4.3.2 RESULTS OF THE ANALYSIS OF VARIANCE FOR THE FACTOR SCORES FOR THE DOMINANT SAMPLE OF 54 (18 Sustenance Driven, 18 Outer Directed and 18 Inner Directed)

The significant differences between the mean factor scores for the three groups were examined. An asterisk indicates the position of significant difference at <0.05 between the risk relieving strategy mean factor scores of the three Social Value Groups. The standard Post Hoc significant test (LSD) was used along side Tukey's HSD test (a more conservative and stringent test) to identify where the significant differences lie.

Table 4.37

	Mea	n Scores		Sd u	Sd using LSD			Sd using		
				,			Tukey's HSD			
Factor	Susten	Inner	Outer	I/O	I/S	S/O	I/O	I/S	S/O	
1 Societal	.5706	6731	.6130	*	*		*	*		
2 Low Cost										
3 Branding										
4 High Cost	4799	1846	.7284	*	'	*	*		*	
5 Author	4218	1136	.2800			*	·····		*	
Reassurance			_							

These results mirror those presented in Table 4.35. Factor 1 'societal acceptance / endorsement' illustrates a significant difference between the Inner directed and Outer directed and between the Inner directed and the Sustenance driven groups. If one examines the Inner directed mean factor score it is clearly significantly lower than the Sustenance driven and Outer directed mean factor scores which are comparable. Factor 4 'High Cost' presents results which support the results of Part three of the questionnaire. Those groups which identified 'high cost' as a perceived risk (Sustenance driven and Inner directed) clearly reject it as a risk relieving strategy and those rejecting it as a perceived risk identify it as an effective risk relieving strategy. These results

confirm the validity of the 'high cost' variables in Part Three and Part Four of the questionnaire. Factor 5 'Authoritarian reassurance' presents a significant difference between the Sustenance driven and the Outer directed. Table (4.35) also identified significant differences between Inner directed and Outer directed in this area. Table 4.37 highlights that the three Social Value Groups have quite different profiles. These results confirm those illustrated in Tables 4.34 and Table 4.35.

4.4.4 INVESTIGATION OF THE RELATIONSHIPS BETWEEN SOCIAL VALUE VARIABLES AND RISK RELIEVING STRATEGY VARIABLES

The following results based on Pearson Correlations express the relationships between the scores for the individual social value variables (screening instrument Part Two of the questionnaire) and those for the individual risk relieving strategy variables (Part Four of the questionnaire) for the sample of 54 respondents. (Appendix 2.5 details the full correlation matrix).

Plus and minus signs illustrate positive and negative significant correlations (< 0.05) between social value variables and food risk perception variables.

Tables 4.38 - 4.42 present the results of this analysis. Each table focusing on one of the 5 factors identified in Chapter 4.

An identical analysis was undertaken for all 215 respondents, the results which are very similar to those for the 54 most dominant are presented in Appendix 7 (correlation matrix Appendix 3.5).

Tables 4.38 - 4.42 examine the aforementioned relationships. Each table focusing on one of the 5 factor solutions illustrated in table 4.36.

Table 4.38 Factor 1 Societal Acceptance / Endorsement as a Risk Relieving Strategy

	Social Value	Advertised	Someone	Family	Celebrity
	Group		similar to you	approves	approves
	·		approves		
Contacts	Susten			+	
Trad	Susten	+			
Best	Outer	+			
Winning	Outer	+	·		
Status	Outer	. +	·		+ .
Ahead	Outer	+	+		+
Mysuccess	Outer	+	+		
Appear	Outer				+
World	Inner				
Unconvent	Inner				. –
Voiceops	Inner				_
Alternat	Inner	-			_

A significant positive correlation exists between Outer directed values and the effectiveness of 'Societal acceptance / endorsement' as a risk relieving strategy (to a lesser degree this relationship also holds for Sustenance driven values). Negative correlations are indicated for Inner directed values in respect of this strategy.

Table 4.39 Factor 2 Low Cost as a Risk Relieving Strategy

	Social Value	Cheaper	Money back	Cheaper
	Group	Store	Guarantee	Brand
Secure	Susten	+	+	
Best	Outer		·	+

The results are not convincing. There is some evidence of a positive correlation between low cost as an effective risk relieving strategy and Sustenance driven variables.

Table 4.40 Factor 3 Branding (Product Orientation) as a Risk Relieving Strategy

	Social Value	Familiar	Brand	*Comprehens
	Group	brand	reputation	ive labelling
Holdlife	Susten			
Contacts	Susten			_
Nochange	Susten			_
Best	Outer		+	
Winning	Outer		+	
Ahead	Outer		+	·
Status	Outer		+	
Alternat	Inner	-		
World	Inner		_	
Unconvent	Inner			+
Voiceops	Inner			+

^{*} Results of the factor analysis indicates that their is an association between comprehensive labelling and Branding.

Positive correlations exist between Outer directed variables and brand reputation as an effective risk relieving / reducing strategy. A positive correlation exists for Inner directed values and comprehensive labelling as an effective risk relieving strategy and in so far as this feature is associated with branded products / services, Inner directed consumers give their support to brands. In sharp contrast the more Sustenance driven values correlate negatively with comprehensive labelling a risk relieving strategy.

Table 4.41 Factor 4 High Cost (Image) as a Risk Relieving Strategy

	Social Value	Expensive	Store Image	Expensive
·	Group	Store		Brand
Nochange	Susten			_
Contacts	Susten	_		
Status	Outer	+	+	+
Ahead	Outer	+	+	+
Best	Outer	+	+	+
Appear	Outer		+	
Mysuccess	Outer		, +	
Winning	Outer	+		+
World	Inner		_	·
Alternat	Inner			

With <u>all</u> Outer directed value variables positively correlating with a minimum of one high cost variable these results indicate that high cost is an extremely effective way of relieving perceived risk for Outer directed consumers. This strategy appears to be unique to Outer directed consumers with both Sustenance driven and Inner directed values presenting negative correlations.

Table 4.42 Factor 5 Authoritarian Reassurance as a Risk Relieving Strategy

	Social Value	Government	Independent	Expert
	Group	Approval	Approval	Approval
Holdlife	Susten		_	
Contacts	Susten			
Nochange	Susten		_	
Status	Outer	+		+
Best	Outer	+		+
Winning	Outer	+		
Ahead	Outer	+		+
Mysuccess	Outer	+		+
Discovery	Inner	_	+	
World	Inner			-
Alternat	Inner			-

Authoritarian reassurance presents interesting results. Outer directed values are positively correlated with the approval of the Government and experts as effective risk relieving strategies. In contrast Inner directed values negatively correlate with these strategies and are more likely to positively correlate with the approval of Independent agencies, whilst for Sustenance driven values the reverse is the case.

4.4.4.1 INVESTIGATION OF THE RELATIONSHIP BETWEEN SOCIAL VALUE GROUP SCORES AND RISK RELIEVING STRATEGY VARIABLES (215 respondents)

The following table identifies the results of the Pearson correlations (for total sample of 215) between respondents Sustenance driven scores, Inner directed scores, Outer directed scores and each risk reliving strategy variable. All positive (+ve) and

negative (-ve) correlations identified in Table 4.43 are significant (<0.05).

Table 4.43 Results of Pearson Correlation for Social Value Group Score and
Risk Relieving Strategy Variable Scores for Sample of 215

Sustenance	F	Sustenance	F	Outer	F	Outer	F	Inner	F	Inner	F
Driven		Driven		Directed		Directed		Directed		Directed	
+ve		-ve		+ve		-ve		+ve-		-ve	
correlation		correlation		correlation		correlation		correlation		correlation	
Advertised	1	Comprehens	3	Advertised	1			Comprehens	3	Advertised	1
		ive						ive		(-ve)	
		Labelling						Labelling			
Similar to	1	Independent	5	Family	1					Familiar	3
yourself		Approval								brand	
										(-ve)	
Family	1			Celebrity	1					Brand	3
		I		Endorsemen						Reputation	
				t		•				(-ve)	
Celebrity	1			Money Back	2					Supermarket	4
Endorsemen										Image (-ve)	
t											
Familiar	3			Familiar	3					Government	5
Brand				Brand			ι.	,		Approved	
										(-ve)	
				Brand	3						
	Ì			Reputation							
				Free Sample	3						
				Supermarket	4						
			İ	Image							
				Expensive	4		7				
				Brand			İ				
	_		1	Expensive	4						П
				Supermarket							
	\dashv		\dashv	Government	5		_			···	
				Approved					ļ		

F = Factor illustrated in Table 4.36

Table 4.43 identifies that the higher the Sustenance driven score the more effective risk relieving strategies associated with 'societal acceptance / endorsement' and 'familiar

brand'. 'Comprehensive labelling' negatively correlates with Sustenance score as does 'Independent approval'. Negative correlations suggesting that the more Sustenance driven the respondent the more likely it is that 'comprehensive labelling' and 'Independent approval' are ineffective risk relieving strategies. The Outer directed results are interesting and a number of significant positive correlations are observed. 'Societal acceptance / endorsement', 'branding (product orientation)' and 'high cost' are key risk relieving strategies. Outer directed results suggest that these are no significant negative correlations with any of the risk relieving strategies investigated. In contrast to the Outer directed results the Inner directed results present a different picture with only one significant positive correlation for 'comprehensive labelling'. A large number of strategies are negatively correlated with the Inner directed (SVG) score across the range of factors. Analysis of the sample of 54 generated comparative results.

These results support results presented in Tables 4.38 - 4.42.

4.5 SUMMARY

Analysis of the results of the Social Value Group screening instrument support its robustness when used in practice on a large number of respondents.

Factor analysis greatly simplified a complexity of variables associated with risk perception and risk relieving strategies.

Relationships between Social Values and Food Risk Perceptions and Risk Relieving Strategies were identified, as were significant differences between these for the three Social Value Groups investigated.

With respect to the Food Risk Perceptions of the entire sample of 215 respondents the analysis identified positive significant correlations between Sustenance driven values and 'high cost risks' and 'unfamiliarity risks'. Significant negative correlations were identified between this set of social values and the set of risks labelled as 'technology' (in the context of artificial) and the set of risks labelled as 'environmental / ecological' (in the

context of non-recyclable packaging and controversial operations not 'pesticides and fertilisers'). Interestingly Outer directed values positively correlate with the set of risks labelled as 'cheap' and with the risk labelled 'different brand'. Significant negative correlations were identified between Outer directed values and the set of risks labelled as 'high cost' and those labelled as 'environmental / ecological' (to include non-recyclable packaging, controversial operations and pesticides and fertilisers). Inner directed values positively correlated with the areas 'technology', 'high cost', 'environmental / ecological' and the risk variables 'non-organic' and 'misleading information', importantly Inner directed values did not correlate negatively with any food risk perception.

The analysis of variance identified significant differences between Inner directed and Outer directed groups and between Sustenance driven and Outer directed groups in the area of 'high cost' as a perceived risk. Similarly significant differences were identified between Inner directed and Outer directed groups and between Inner directed and Sustenance driven groups for 'environmental / ecology' as a perceived risk.

With respect to Risk Relieving / Avoiding Strategies the analysis identified positive significant correlations between Sustenance driven values and the set of risk relieving strategies labelled as 'societal acceptance / endorsement' and the strategy 'familiar brand'. The risk relieving strategies 'comprehensive labelling' and 'independent approval' correlated negatively with Sustenance drivenness. Interestingly whilst Outer directed values do not negatively correlate with any of the risk relieving strategies this set of values positively correlate with an extensive range of strategies including 'societal acceptance / endorsement', 'high cost', 'authoritarian reassurance' and 'branding'. In contrast Inner directed values positively correlated with only one relieving strategy, i.e. 'comprehensive labelling'.

The analysis of variance identified significant differences between Inner directed and Outer directed and between Inner directed and Sustenance driven in the area of 'societal acceptance / endorsement' as a risk relieving strategy. Similarly significant differences were identified between Inner directed and Outer directed and between Sustenance driven and 'Outer directed' for 'high cost' and between Sustenance driven and Outer directed for 'authoritarian reassurance' as risk relieving strategies.

The results indicate that where few or no risk relieving strategies are perceived (i.e. in the case of the Inner directed respondents) a correspondingly high number of risks are perceived, the reverse being similarly true i.e. where many risk relieving strategies are perceived (as in the case of Outer directed respondents) few risks are perceived. This finding raises the issue to what comes first and which area is most critical for consumer research.

Qualitative interviews were used to confirm and clarify quantitative results whilst increasing the depth of understanding.

CHAPTER FIVE

QUALITATIVE ANALYSIS

The following chapter presents the qualitative results of the research.

The chapter is structured into sections focusing on the results of each of the three sets of Interviews undertaken as described in Chapter Three. Each section is subdivided to illustrate the responses of the members of each of the three Social Value Groups in turn. The structure also allows readers to refer back to the quantitative results described in Chapter Four.

The results of Interview One will firstly be presented and analysed. Interview One took place during the pilot study, its purpose to confirm the validity of the SVG screening instrument as a tool for identifying Social Value Group membership and to further explore the social values of those groups. All nineteen members of the pilot sample participated in Interview One. The analysis presented in this chapter focuses on the interviews of the three Inner directed, three Outer directed and the three Sustenance driven respondents draw from the pilot sample of nineteen.

The results of Interview two and three are then presented and analysed.

Interview three was designed to explore, clarify and verify responses to parts three and four of the main study Questionnaire (food risk perceptions) and (risk relieving strategies). The analysis is focused on interviews of fourteen respondents obtained from the main study sample (Interview 3) as well as a small number of food risk perception interviews undertaken during the pilot (Interview 2).

In all cases the individuals fell into the group of interest identified in Chapter Three i.e. women aged 25 - 45, with no dependent children, economically active with an income above the national household average.

5.1 ANALYSIS OF QUALITATIVE INTERVIEWS

The richness of the data combined with a relatively small number of transcripts meant that it was feasible to manually analyse the transcripts. Whilst this was a lengthy process it allowed for a thorough, in-depth and rigorous examination of the responses.

The selection of quotes was both systematic and objective. The aim was that the criteria for selecting quotes should ensure that all interviewees were represented for all the areas to be investigated and that the quotes selected represented the views of those respondents accurately. Fundamentally in an attempt to produce valid interpretations every effort was made to avoid bias when selecting quotes; similarly contradictions were actively sought out.

Interviewees were presented with a structured framework of questions for Interview One and as such a set of responses for all of the areas, 'Family', 'Work and Leisure', 'Attitudes Towards Authority' and 'Organisational Structures' were obtained and are included in the following analysis. Some interviewees clearly responded in more detail than others and the number of quotes reflect this. Repetitions by the same interviewee are not included. A response in every question area by every respondent in Interview One is included with the exception of interviewee S1 who for personal reasons discussed the 'Family' in an way inappropriate for the analysis. Similarly interviewee O2 and interviewee O3 cut their interviews short due to work commitments and as such responses to 'what might improve their lifestyle' and 'what makes them feel good about themselves' were not obtained.

Unlike Interview One, Interview Two and Interview Three questions were determined by the Interviewee's questionnaire responses. The areas of questioning focused only on extreme responses, as such an Interviewee's transcript only reflected areas of risk that the interviewee was deeply concerned about or had very little concern about. Similarly only risk relieving strategies which were indicated as extremely effective or not at all effective were discussed.

The organising framework for selecting quotes was fundamentally that of MacNulty's work on the differing beliefs, attitudes, dispositions and motivations of sustenance driven, outer directed and inner directed consumers (1985).

Thus, for example, in the analysis of Interview One (social values), the quotes selected for inclusion evidenced the differing values, attitudes and dispositions of the three groups (sustenance driven, outer directed and inner directed) in the areas of 'family', 'work and leisure', 'attitudes towards authority' and 'organisational structures' as outlined by MacNulty (1985). The aim of this approach was to validate the usefulness of the social value group screening instrument.

Quotes derived from interviews three (food risk perception) and four (risk relieving strategies) were organised into the areas generated by the factor analyses. The outcomes of the two factor analyses undertaken providing meaningful and useful initial frameworks for this purpose.

As with interview one, the quotes selected for inclusion were those that highlighted motivational differences between sustenance driven, outer directed and inner directed groups in each of the areas concerned and which could be made sense of by using MacNulty's insights as the organising principal. Thus, for example in the area of (technology related food risk perceptions" themes such as fear of the unknown (sustenance driven consumers), issues of personal control (inner directed consumers) and belief in the system (outer directed consumers) were identified as significant and quotes were included.

By consistently using MacNulty's framework as the overarching frame of reference for analysis of the qualitative material, it was possible to address the research question, i.e. whether or not and, if so, how the social values concerned influenced respondents food risk perceptions and risk reliving strategies.

The three sets of interviews provide a means of triangulation which aids in both the clarification of interviewee values, attitudes and perceptions and the validation of the

individual interviews. Fundamentally the Qualitative analysis supports the Quantitative findings.

Appendix 8 details the qualitative samples for each of the three Interview stages.

Appendix 9 maps ownership of the quotes included in the analysis.

5.2 INTERVIEW ONE SOCIAL VALUES AND SOCIAL LIFE / LIFESTYLE

Interview One, as previously discussed in Chapter Three aimed to explore respondents' motivations, values and attitudes with regard to fundamental elements of their social life / lifestyle, focusing on those areas explored by MacNulty (1985) to include the family, work and leisure, attitudes towards authority and organisational structure. Interviewees were also asked 'what might improve their lifestyle' and 'what made them feel good about themselves'.

Interview One was conducted one week after the respondents had completed the screening instrument questionnaire.

As previously mentioned the responses of three Outer directed consumers, three Inner directed consumers, three Sustenance driven consumers are included. These nine respondents were drawn from an initial pilot sample of nineteen.

5.2.1 FAMILY

As previously examined in the Literature Review (Theoretical framework) MacNulty (1985: 333) identified Sustenance driven consumers as,

"conservative, traditional, conventional, pragmatic and devoted to his family, which he places above other things and for which he makes substantial self sacrifice".

"He tends to be against change and his interest in the future is to ensure his family's security"

Sustenance driven respondents believe the family unit supports their need for security and stability. Resistant to change he describes them as "clannish".

In contrast the Outer directed consumer's priority is that the family be 'a credit to them',

"They are interested in their family's appearance and behaviour..."

MacNulty (1985: 334, 342) suggested that Inner directed consumers are "highly individualistic"

"They reject traditional authority, and are a powerful force for individual autonomy and the use of new relationships".

The following quotes were extracted from Interview One transcripts.

5.2.1.1 SUSTENANCE DRIVEN RESPONDENTS

```
"The family is very important" (S3)
```

"Members of the family come first...." (S2)

"If we are in trouble there is always someone there we can call on" (S2)

"Always in touch" (S2)

"Really important for security for everyone that's in the family" (S2)

"Influence me definitely" (S2)

"(Family provides)...support, security...." (S2)

"(Family provides)....love, security, being wanted...." (S3)

"Family there to support everyone if they need it" (S2)

"I would always stick up for my family no matter what" (S3)

"Its an important thing in my life to have a good friend but its just not the same as family, it's no where near as important as family, they come first" (S3)

SUMMARY

These quotes strongly support MacNulty (1985). Sustenance driven respondents stressed that the family was fundamental to their day to day life. They do value friendship however the interviewees suggest that their family always comes first. As a whole Sustenance driven respondents appear proud, loyal and defensive of their family unit. Additionally they suggest that they were greatly influenced by their family particularly the nuclear family.

5.2.1.2 OUTER DIRECTED RESPONDENTS

```
"A unit of stability, they give me support" (01)
```

"I would like people to think that I came from a middle - working class respectable family where I have had a very loving supportive childhood" (01)

"If someone says something about someone in my family, against them I will find out what it is that has created that impression and do everything that I can to put things in a favourable light again. So yes it is very important." (01)

"Family is a background concern of mine.... I don't really see my family at all....time commitments and other things don't allow me to" (O2)

"I have more contact with friends than I do family" (O2)

[&]quot;(I rely on them) if I am down" (01)

[&]quot;(Influence me) no my family never have" (O2)

[&]quot;I am very independent" (01)

[&]quot;I like people to think favourably about the family" (O3)

[&]quot;People's opinions of my family influences my opinions of them to an extent"
(O3)

[&]quot;The family unit to me means stability" (O3)

SUMMARY

Outer directed respondents stressed their independence. Rather than viewing the family as an integral part of their life, the Outer directed respondents appear to be more distant seeing the family as a cushion to fall back upon if the need arises. They saw the family as a source of stability rather than security. They appear quite adamant that their family do not influence them. Respondents believed that it was important that people outside the family unit thought favourably about their family. The interviewees suggest that friends are just as important as family.

5.2.1.3 INNER DIRECTED RESPONDENTS

"The way my family works is that every body is left to their own devices" (I1)

"Everybody gets on with their own thing and no body really interferes"(I1)

"It doesn't concern me what other people's opinions of us are, they probably think that we are strange anyway" (I1)

"Friends are important, I would say on the same level (as family)" (I1)

"I wouldn't necessarily let it (her family) influence me. I would decide depending on the situation" (I3)

"As an individual, I have no ambitions to be married or have a family, so in that respect it (the family) is not important to me at all" (I2)

"I do not have to see them a lot, I don't have to contact them a lot" (I3)

"It does not matter what other people think at all" (I3)

"Friends.... yes they are important, yes" (I3)

"Family is important to me..... but not necessarily as defined in the standard way
(I3)

SUMMARY

The Inner directed respondents gave a strong sense of independence. They appeared to respect the family unit and recognise it as valuable but not vital to their sense of security or self identity. Often suggesting that they have a non typical family structure. They

don't feel their family influenced them. Friends are very important either on the same level or perhaps more so than the family.

Inner directed respondents described a complex system where the feeling and wishes of others were considered but the end decision would always be one made by one's self.

5.2.2 WORK AND LEISURE

MacNulty (1985: 332) suggested that Sustenance driven consumers,

"are motivated by the need for security"

they were frequently economically disadvantaged but even when this was not the case the Sustenance driven consumer would still remain thrifty. There is a suggestion that they perceive work as a necessity and leisure very much a luxury. Outer directed consumers are described by MacNulty (1985: 333) as,

"materialistic, pushy and motivated by seeking to improve their position in financial and social terms".

This being the case both work and leisure time would be fundamental to their self identity. In sharp contrast MacNulty (1985: 333, 334) suggested that the Inner directed consumers had a very different work/leisure ethic, emphasising the attitudes of,

" caring, autonomous behaviour and self realisation"

Work perhaps being aiding their aim of self expression and self realisation rather than just economic / material necessity or power.

The following quotes were extracted from the interview transcripts.

5.2.2.1 SUSTENANCE DRIVEN RESPONDENTS

"I would like a lot more (leisure)..... you have got to go to work and that is priority number one" (S1)

"I think they (work and leisure) are conflicting at the moment" (S2)

"Work is too important in my life I should enjoy life more" (S2)

"It's a separate part of my life and when I get home I forget about work and concentrate on my family, they are different time, parts of my life" (S3)

"I'm not bothered about going out to meet anybody else and mixing" (S3)

SUMMARY

The Sustenance driven respondents consistently expressed that work and leisure are very much in conflict with each other, two very separate elements of their life. Work is seen as a necessity, a means to provide security for the family, this perhaps being the motivation. Their orientation towards the family and 'clannish' trait is emphasised further as the respondents suggest that their leisure time is spent either with family or very close to home.

5.2.2.2 OUTER DIRECTED RESPONDENTS

"I would rather be at home" (S3)

"I would say that they (work and leisure) conflict" (O1)

"They (work and leisure) do conflict with each other yes.... I don't have as much leisure time as I would like" (O3)

"I am trying to train myself to see them as two very separate things so that when I come home I can switch off. A lot of my leisure time used to be with people from work" (O1)

"My life is very hectic.... my working time and my leisure time interact and destroy each other in a way" (O2)

"I am a member of a club and I take that very seriously, you compete for the club and yourself at the end of the day yourself has to come first" (O3)

"I play badminton yes, I play competitively it would be boring otherwise, I like to be good at what I do or what's the point" (O1)

"I think my work cuts into my leisure time a lot" (O1)

"Your leisure time, you have got to try and make the most of it, it's got to be seen as valuable, important to your self and for your self' (O1)

"I am living to work" (O1)

"I enjoy the competing... and the winning and the experience of travelling around and meeting people" (O3)

SUMMARY

Outer directed respondents suggested more contact with colleagues out of work hours than the other social value groups, often finding it difficult to distinguish between the two.

Outer directed consumers appear to be organised and carefully plan the use of their leisure time. One might further suggest that Outer directed consumers appear to view their leisure time as a commodity.

The competitive nature of the Outer directed consumer is very evident both in the context of their work and their leisure.

5.2.2.3 INNER DIRECTED RESPONDENTS

"They (work and leisure) have always had a close relationship, because what I enjoy doing in my leisure time is what I like to do as my work" (I1)

"Hobbies and my work both go straight together" (I1)

"I don't see a boundary, leisure and work time, they are both living time, my aim is to make them mesh" (I2)

"I enjoy my home very much I like to look to my life as a rounded whole where all my senses are used and enjoyed I do a fair amount of socialising. I feel like a hotel at times" (I3)

"I don't see a boundary, there is a boundary obviously in terms of what you put your time into but in terms of the underlying values and principles that I hold they are equally applicable" (I2)

SUMMARY

Work and leisure time appears almost interchangeable for the Inner directed consumers, when this is not the case it does appear to be an aim. Inner directed respondents suggested that they actively seek out work that has parallels with their personal interests.

5.2.3 ATTITUDES TOWARDS AUTHORITY, EXPERTS AND CONVENTION EXPLORED THROUGH AN EXAMINANTION OF HEALTH

MacNulty (1985: 333) suggested that Sustenance driven consumers are

"conservative, traditional and conventional"

MacNulty suggested that we might expect this group, "to be relatively limited in the scope of their behaviour. Alternative courses of action and imaginative responses to new situations will be difficult". Outer directed consumers are described as respecting and trusting experts and authority but being motivated by wanting 'the best' (: 334). MacNulty (1985) indicates that Inner directed consumers exhibit autonomous behaviour

(: 334). They are further described as unorthodox, as having little trust for traditional

authority and importantly, very likely to be concerned with spiritual values.

MacNulty described the Inner directed consumers as,

"...likely to reject doctrinaire solutions in favour of the moral holistic view that change is an organic process"

"broadest horizons, the highest tolerances and the largest propensity to solve problems on a global scale. They also have a tendency to hold spiritual values"

5.2.3.1 SUSTENANCE DRIVEN RESPONDENTS

"I tend to leave things..... if after a couple of weeks I was still ill I would go to the doctor's" (S1)

"I feel that when you visit the doctor you are just a statistic" (S1)

"I feel like I have to convince the doctor that something is the matter" (S3)

- "I would try and do things...before I resorted to going to the doctor's" (S2)
- "If I'm unwell I first try and treat it myself, if it persists I do go to the doctors" (S3)
- "I don't really agree that people with more money should get better health care" (S1)
- "I would only consider private health care, if the doctor wasn't helping me or I thought that there was something really serious" (S2)
- "I might be tempted to try acupuncture" (S2)
- "I've had a go at acupuncture" (S3)
- "I trust what they (the doctor) say I would never go against what they said to me" (S3)
- "I do tend to go along with what they (the doctor) say....." (S1)

SUMMARY

Sustenance driven consumers appear to view their doctor as an authority figure. There is an indication that they trust their doctor (on the whole), that they believe in the National Health Service, and would only venture into the world of private health care in the last resort. They are sometimes a little reluctant to go to the doctors indicating that they feel they might be 'troubling' the doctor. There is some evidence of a willingness to try alternative approaches to conventional medicine.

5.2.3.2 OUTER DIRECTED RESPONDENTS

- "If I thought it was something more serious than a cold or flu I would go (to the doctor) without hesitation" (O1)
- "I generally think I'll be alright tomorrow so I'll leave it" (O2)
- "as soon as anything is wrong yes I do seek medical advice" (O3)
- "If they prescribe me something I wouldn't doubt or question it, I would trust them implicitly, on the whole I do trust them" (O1)
- "I think its a great service (the doctors) when you need it" (O2)
- "I have been in BUPA until last year but it was just costing me too much. In an ideal world I would go back to that. If you can afford it, in a funny kind of way

you are helping the NHS. It's a bit selfish but if you can afford it I think its much better service" (O3

"I think it is worth it (private health care), rather than waiting around for the NHS. It's worth spending money on your health" (O2)

"If I thought I could benefit from (private health care) yes (I would use it)" (O1)

"I'm not sure how much I believe in all these aromatheropy and things, I would just stick to conventional medicine" (O1)

"If traditional medical practices don't achieve what you want I think it's justifiable to use alternatives" (O2)

"If there was a possibility of alternative medicine I would probably try it. I use herbs and vitamins and things instead of antibiotics" (O3)

SUMMARY

Outer directed respondents appear to have confidence and trust in their doctor. They show little reluctance to visit their doctor compared to Sustenance driven respondents. Striving towards gaining the best possible service, Outer directed respondents indicate that if they are not happy with the 'service' that the NHS offers they will actively seek a 'better service' in the private sector. Additionally they are not averse to using alternative medicine.

5.2.3.3 INNER DIRECTED RESPONDENTS

"Doctors have got a habit of prescribing things that are not always in your best interest" (I1)

"I really don't like to feel I have to go (to the doctor's)" (I1)

"The doctor where I go is very impersonal" (I1)

"I would do something like Thai chi...... Or I would try a Chinese herbalist" (I1)

"I'll use the NHS every time ... private doctors are not any better than NHS doctors. It's being paid for we should be using it" (I1)

"Some but not a lot (of confidence in doctors), I like to try and find out about things I don't like to just go along with what they say" (I1)

"I don't have any sort of fears about going (to the doctors)" (I2)

"I think the closer to nature it is the better and if it doesn't work then I am not frightened to try something more conventional" (I2)

"Politically I am not happy about private health care at all" (I3)

"I am increasingly having an alternative approach to health care, I haven't used the NHS for ten - fifteen years, I go to a homeopath and various others" (I3)

"I like the holistic nature of alternative medicine" (I3)

"I do not trust the NHS for things which are not purely physical" (I3)

"In general I have very low confidence (in doctors)" (I2)

"I have thought that if the NHS would just move its attitude just a bit and start realising that the patient is part of the cure and actually give you some say in what happens" (I3)

SUMMARY

The Inner directed responses indicate some scepticism of conventional doctors. They suggest that they are against private health care from a political and ethical stand point.

Of the three Social Value Groups, the Inner directed respondents indicated that they perceived that their health was something that they had a high degree of control over. They choose to adopt a more holistic approach, actively trying preventative and alternative medicines.

5.2.4 ORGANISATIONAL STRUCTURES (WORKING ENVIRONMENT)

MacNulty (1985: 333) suggests that Sustenance driven consumers tend to be worried, "pedantic", "rule-following" and generally against change. Outer directed consumers are described as being "traditional", "pro authority" and supporting order. Motivated by "acquisition", "status", "competition" and "getting ahead" MacNulty (1985: 333) suggests that they are concerned about "appearance" and position. He suggests that Inner directed consumers MacNulty (1985: 334) reject established procedures and are a powerful force for "individual autonomy" and the use of new organisational structures

and relationships. Importantly however they are further described as altruistic, unselfishly concerned about the welfare of other.

5.2.4.1 SUSTENANCE DRIVEN RESPONDENTS

- "I think it's important that you work in a happy atmosphere" (S2)
- "It's important that you all get along" (S3)
- "I think you need to feel secure" (S2)
- "..... that the company cares about their employees, you need to feel that what you do is important and valued" (S2)
- "If there wasn't any rules.... that's when problems occur" (S1)
- "It (structure) is definitely important" (S2)
- "Yes that (structure) is important" (S3)
- "There needs to be procedures in place" (S2)
- "I am motivated by responsibility, getting recognition and money" (S2)
- "Also working in a happy atmosphere where you get along with your colleges" (S3)
- "Motivated by "a nice atmosphere, things running smoothly, know what you are doing" (S3)
- "It's definitely important. I mean to have problems at work and not know that there are people in place to go and see, it must be really frustrating and demoralising. I don't think you get the best out of people when they don't know where to go or what to do about things" (S2)
- "I like to know what I'm doing and where I stand" (S3)

SUMMARY

The responses suggest that Sustenance driven consumers respond positively to structure, rules and procedures. The responses also suggest that Sustenance driven consumers enjoy and feel most happy when they are working in a harmonious environment where there is little conflict or change.

5.2.4.2 OUTER DIRECTED RESPONDENTS

- "I like to be the best in what I do. (O1)
- "Some sort of structure. I like to be given some direction and know where I am going" (O1)
- "I like an environment where there is lots of confidence and trust and you can go and get on with your job happily" (O1)
- "There has got to be some structure, but it can slow things down. Rules are there for a reason but they are also there to be broken" (O2)
- "I like an organisation to be structured so you know where you are" (O3)
- "I like to work with different people, so I suppose an organisation where you get lots of contact with lots of people" (O3)
- "I much rather do things myself" (O1)
- "I don't like dictatorial management. I like to be given the responsibility to do things my way" (O1)
- "I like the freedom to do things my way" (O3)
- "I like to voice my opinions" (O3)
- "I think in a strict ridged hierarchy it makes you very competitive, you might not do the job to your best because you spend so much time competing with the others to show the best results" (O1)
- "good resources are vital" (O3)
- "respect, challenge, commitment of those around me, status. I want to have a higher position a more senior position and yes a cracking salary. I prefer success and the recognition and achievement than the money, but at the end of the day I want to own a BMW" (O1)
- "Ideally I look for rewards, pension schemes, bonuses, and things like that. I think they are very important" (O1)
- "If I had the option I would like to work for myself or in a partnership with a few people, just so I get a better reward for the work I am putting in, more than if I am working for someone else" (O2)
- "financial rewards are priority" (O2)
- "People cannot be given too much information and I think everybody should be informed about decisions that have been made that affect them" (O1)

"I like respect" (O1)

"I like a challenge" (O1)

SUMMARY

If we examine these quotes it would be fair to suggest that the Outer directed consumer might believe the 'right' organisational structure is the one that facilitates their personal progression and development. Motivated by self gain with material rewards as an essential expression of success. Structure and rules are viewed as important but only to the point that they support them and support their progress. Resources to facilitate their personal progress are crucial.

5.2.4.3 INNER DIRECTED RESPONDENTS

"The attitude of the people I am working with is very important" (I1)

"If you have a free thinking organisation you can get more done than if you are stuck within strict guidelines" (I1)

"I work best in a small team" (I2)

"Know how your work links in (to the organisation as a whole)" (I2)

"Take pride in what you are doing" (I2)

"Encouraged to stretch yourself" (I2)

"Flexible" (I1)

"You have to have guidelines but there has to be flexibility" (I2)

"Nice atmosphere - open rooms" (I1)

"You always need some sort of procedure but you can't have things engraved in stone because there's always a point were things are going to change and you have to be able to adapt" (I1)

"A learning organisation" (I3)

"Work together, where you and your contribution is valued" (I3)

"People orientated - its only when you value your people that change can happen and you can move on and up" (I3)

"Communication is the key element, once that breaks down you can forget it"
(I2)

- "Where mistakes are allowed to happen because people are always learning" (I3)
- "A learning organisation, one that will grow one that will value the input of everybody" (I3)
- "They (managers) are there to steer and guide us to what their strategies and aims are" (I2)
- "Making a difference and being valued for doing so" (I3)
- "Having some level of power and being able to use that to empower others" (I3)
- "Money doesn't motivate me, it is important but it's a short term motivator other things are long term" (I3)
- "Being stretched intellectually" (I3)
- "If I think I can help someone that gives me a buzz" (I2)

SUMMARY

Inner directed consumers describe a complex environment. Fundamentally this environment is described as one that facilities effective communication, learning and is open and flexible to change.

Fundamentally Sustenance driven consumers strive to feel secure. Outer directed consumers strive to be seen as successful and Inner directed consumers strive to develop and self actualise.

5.2.5 WHAT MIGHT IMPROVE THEIR LIFESTYLE

This question was added to the end of the interview essentially for two reasons firstly purely as a matter of interest and exploration and secondly to the end of the interview on a light note.

5.2.5.1 SUSTENANCE DRIVEN RESPONDENT

"More financial security" (S3)

"Better holidays, perhaps even a better car, more things for the house" (S1)

"My husband not spending as long at work as he does" (S3)

"If the opportunity came up to move and we would still be financially stable I might like to, but, I would be worried that we might not have the happiness that we have got now" (S3)

"I've got happiness here so why risk changing things and things going wrong."
(S3)

"As long as the bills are paid and I have that piece of mind I am happy" (S1)

5.2.5.2 OUTER DIRECTED RESPONDENTS

(Time constraints for respondents O2 and O3 meant that only O1 completed this part of the interview)

"More money. I am very restricted in what I do in my leisure, there are things I want to do but just don't have the money to do them" (O1)

"I would like to own my own house" (O1)

5.2.5.3 INNER DIRECTED RESPONDENTS

"I quite like my lifestyle" (I2)

"To spend more time doing some things that I really want to do" (I2)

"Doing less" (I3)

"Less stress, less volume in relation to food" (I3)

"Stopping smoking" (I3)

"Being valued as well, self esteem is an issue" (I3)

"Having more time that would make me feel much better" (I3)

5.2.6 WHAT MAKES THEM FEEL GOOD ABOUT THEMSELVES

As with the previous question (5.2.5) this question was added to the end of the interview for two reasons firstly purely as a matter of interest and exploration and secondly to end the interview on a light note

5.2.6.1 SUSTENANCE DRIVEN RESPONDENTS

"To enjoy my work and get along with the people around me" (S1)

"This course that I have just done. It's definitely been morale boosting" (S1)

"To lose weight" (S3)

5.2.6.2 OUTER DIRECTED RESPONDENTS

(as with the previous section, time constraints for respondents O2 and O3 meant that only O1 completed this part of the interview)

"I like to make people laugh" (O1)

"Achieving things and getting over things" (O1)

5.2.6.3 INNER DIRECTED RESPONDENTS

"I like to see myself grow" (I3)

"I like to watch people learn that is what makes me feel good really good" (I3)

"Helping someone else grow" (I2)

"Helping others" (I2)

"When you have things in common and you have good communication with people" (I1)

SUMMARY

Responses to 5.2.5 and 5.2.6 supported the values expressed in response to other elements of Interview One.

SUMMARY

Interview One was successful in validating Part Two of the questionnaire, i.e. the Interviews provided the researcher with the confidence that Part Two of the questionnaire was a valid tool by which consumers could be categorised accurately with

regard to Taylor Nelson's Social Value Groups. The sample of consumers exhibited many of the values and attitudes of Taylor Nelson's Social Value Groups as detailed by MacNulty (1985).

The Interviews did however provide much more than confirmation of the validity of Part Two of the Questionnaire. The Interviews provided a means to gain an in-depth understanding of the motivations of these consumers and an indication to how their attitudes and values might influence food risk perceptions and risk relieving strategies.

5.3 INTERVIEW RESPONSES TO FOOD RISK PERCEPTION QUESTIONNAIRE

The following section aims to analyse the risk perceptions responses from Interviews two and three.

The following section headings correspond to the Factors derived during the quantitative analysis. This allows the reader to easily refer back to the quantitative analysis.

As with the previous section, each of the Social Value Groups responses to each of the risk categories will be presented and summarised.

The analysis focuses on the interviews of fourteen respondents obtained from the main survey (Interview three) as well as a small number of food risk perception interviews undertaken during the pilot (Interview two). In all cases the individuals fell into the homogeneous group previously discussed in chapter four.

5.3.1 TECHNOLOGY AS A PERCEIVED RISK

5.3.1.1 SUSTENANCE DRIVEN RESPONDENTS

"I don't really know anything about it (genetically engineered food) but it doesn't sound too good" (S2)

"It is not certain what sort of long term effects they (artificial preservatives colours) might have on people" (S13)

"It just seems you are not told enough about them (technological things) or you are told conflicting things, I don't trust the research" (S10)

"It (genetically engineered food) probably would worry me more if I thought about it a lot" (S11)

"Where is it (genetic engineering) going to develop, presumably there have got to be strict controls and are they going to be in place and enforced or will the controls be ignored anyway?" (S15)

"Looking at things from a consumer's point of view I think if it (technology) makes food cheaper......at the end of the day if they are making vegetables grow bigger or what ever getting a better product" (S14)

"I look for 'E' numbers" (S11)

"I think colouring unnecessary and to an extent flavourings are as well" (S14)

SUMMARY

Sustenance driven respondents demonstrated a lack of knowledge. Interestingly they confessed that their worry was largely fuelled by a lack of knowledge and rumours that there was a risk to be perceived. Their position could be interpreted as fear of the unknown. They were also concerned that rules / controls were being broken.

This group suggest that they are willing to bear the risk of technology if it facilitates a financial saving.

5.3.1.2 OUTER DIRECTED RESPONDENTS

- "No I don't see it (genetically engineered food) as particularly high risk there are other things that are much higher risk" (O1)
- "I guess its largely because I know that things are monitored and done under controlled conditions." (O1)
- "I also think that genes, well they're natural to start off with" (O1)
- "Pesticides and fertilisers I assume that they have been tested much more than the genetically engineered, I would think they would be safer" (O11)
- "Well the more I have heard about genetic engineering, sometimes it worries me that we might be getting into something that we can't get out of" (O11)
- "With genetically engineered food I know so very little about the background and the science of it so I find that a bit disconcerting" (O13)
- "I suppose it is a lack of knowledge that (worries me) not knowing where there artificial flavours and colours come from not knowing what they can do to you" (O13)

SUMMARY

For the Outer directed respondents there is a stronger sense of trust and belief in the controlling bodies. They accept technology (as part of life) to a larger extent than the Sustenance driven respondents. However there is still some concern over lack of knowledge.

5.3.1.3 INNER DIRECTED RESPONDENTS

"My concern in this is because I feel that it is totally outside my control, most of our control, that it sometimes difficult to trust the information that is put out by the media" (I12)

"If you like, it is that I have no control over the production or the preparation of the food before I consume it" (I11)

"I don't think there is enough information about it (genetic engineering) really there's not enough labelling so that would concern me" (I10)

"Genetic engineering, junk food, BSE, pesticides and fertilisers I feel the same about all these things" (I2)

"It seems that you can't trust producers to inform you as a consumer and I think that explains my worry in terms of genetic engineering and indeed BSE and pesticides" (I11)

"You don't know really whether to believe the research, that they come out with" (I11)

"It is to do with trust, but I am very cynical" (I11)

"It (genetically engineered food) doesn't worry me but I don't like it. It doesn't worry me in terms of the food risk it's maybe from an ethical point of view" (I3)

"As long as it is being controlled properly it (genetic engineering) doesn't worry me" (I13)

"All these healthy eating products but for all we know a lot of their products may be full of chemicals, there is a total hypocrisy" (I11)

"Chemically would worry me more than genetically" (I3)

"I wouldn't buy something that had a lot of e numbers" (I10)

"We don't know prefer natural to artificial" (I3)

"It's artificial to me" (I2)

SUMMARY

These quotes demonstrate the high degree of concern Inner directed consumers exhibited and the complexity of their thinking.

Importantly a perceived lack of control fuelled concerns for Inner directed respondents.

Inner directed respondents are also concerned about the availability and reliability of information, they often allude to concerns that information cannot be trusted. They seem to prefer the 'natural' to the 'artificial' and link 'ethics' to matters of technology.

Fundamentally low levels of trust appear to be leading to scepticism and cynicism.

5.3.2 HIGH COST AS A RISK

5.3.2.1 SUSTENANCE DRIVEN RESPONDENTS

"Good quality doesn't mean it is the most expensive thing" (S10)

"I wouldn't shop there (M&S) for things I could get cheaper elsewhere" (S2)

"I don't shop at the likes of M & S" (S11)

"At the end of the day you are not getting a better quality product you are just paying to improve marketing people's salaries, you throw it in the bin at the end of the day anyway and that just seems stupid to me" (S14)

"I certainly just don't buy pre packed vegetables.....they're more expensive I guess" (S15)

"I tend not to buy fresh fruit or vegetables from the supermarket, it tends to be far more expensive than you would buy in a local greengrocer or the market" (S12)

"I buy them from my local greengrocer, I buy them (potatoes) by the bag I think they come straight from the farm or wherever and they are unwashed. I am very conscious of cost, I budget working out an allowance it's one of my main priorities I have so much money each week to spend and I have to keep within those limits so I tend to go for good value" (S12)

"I wouldn't go to Marks and Spencer for food, I am not sold into that idea at all. You do have your routine of where you go. I am really not bothered about packaging and that is basically what it is for me. There is a certain snobbery about supermarket shopping, J Sainsbury and all that, at the end of the day I just want to get it all done as quick as possible as cheaply as possible" (S14)

"Well I don't, (shop at "expensive supermarkets) a classic example would be somewhere like M & S I never would dream of buying any fresh produce from there because I just don't think it is good value" (S12)

"I know that they put the same product into different packaging there are slight differences but for me that slight variation I won't pay for" (S14)

"I would think it (pre-prepared vegetables) too expensive" (S13)

SUMMARY

Sustenance driven consumers point strongly to their belief, that high cost did not act as assurance of a 'better' product. In the interest of looking for 'good value', high cost was indicated as a very real worry. They appear to perceive that they would be paying for the intangible image of the product rather than it's tangible characteristics. There was a common belief that their money would go further by purchasing food in 'cheaper' stores and/or purchasing 'cheaper' brands (which were perceived by the Sustenance respondents as 'of equal quality'.)

5.3.2.2 OUTER DIRECTED RESPONDENTS

"You are assured quality and that's value for money" (shopping at Marks and Spencer's) (O10)

"You know what you are getting and it will be as good as you can buy" (O10) "I do shop there for some things especially if I fancy something nice, there are some things in there that do taste much nicer, it comes down to the quality factor

"I don't think price puts me off because I think if you go to a more expensive supermarket they are more convenient and you get a better quality and choice" (O13)

but I wouldn't do my weekly shopping there" (O11)

"As long as I can get what I want, that it's convenient and that I am happy with the quality that's fine" (O12)

"Something to do with the level of quality and the content and the layout of the supermarket....... the aisles are bigger, wider it seems easier to do your shopping perhaps its because you are pampered more as a customer and everything is laid out better..." (O13)

SUMMARY

A very strong assumption that price and quality are positively correlated and that high quality (and consequently high cost) minimises risk.

Outer directed respondents were casual when discussing this section. It was common sense to them that 'you get what you pay for'. They appear happy to pay more and appeared almost proud of the fact that they pay more and as a result perceive that get 'the best'. They perceived that this was an effective way of re assuring themselves that they would be purchasing the 'right' product for them.

5.3.2.3 INNER DIRECTED RESPONDENTS

"The price to me is not an indicator of the content at all, not at all" (I2)
"I would rather pay more for something that is more natural" (I12)

SUMMARY

Unlike the Sustenance driven and Outer directed respondent who appear clear about price effecting their risk perceptions, the Inner directed consumers indicated quite assertively that on the whole they did not believe the cost of the product necessarily to be a good indicator of quality. The cost of the product did not appear to be a means by which they measure, judge and ultimately make a purchase decision. Responses did indicate however that they were not willing to pay for intangibles that they would only pay more for benefits that they valued i.e. natural food.

5.3.3 NUTRITION AS A RISK

5.3.3.1 SUSTENANCE DRIVEN RESPONDENTS

"I probably eat too many sweet things but I always try to eat plenty of fresh food fruit and veg" (S2)

"I tend to watch what I eat in the high fat (range) I use low calorie spreads skimmed milk" (S12)

"I have always used salt in cooking but I try to discourage it at the table" (S12)

"..salt I find it is a worry from the health problems that it can cause in later life" (S12)

"I have to be careful....I have high blood pressure cholesterol I am very aware" (S11)

"The latest thing at the moment is for the supermarkets to jump on this low fat band wagon and saying 'low fat' and then bumping up the price" (S14)

5.3.3.2 OUTER DIRECTED RESPONDENTS

"I do eat what I want but I have to exercise to keep it off, but I don't like to eat a lot of fat" (O11)

"I would look at the fat content if the more expensive one had more fat in it I would buy the cheaper one" (O11)

"I like to look after myself so nutrition (is important)" (O11)

"I have become aware of how important it is to eat the right things in the right quantities" (O12)

5.3.3.3 INNER DIRECTED RESPONDENTS

"I believe you are what you eat to a great extent" (I2)

"I am concerned about my diet as a whole" (I2)

"I eat to much" (I3)

"I do feel healthier when I eat less fat" (I13)

SUMMARY

Nutrition was a worry shared by all interviewees. The interviews did not highlight any clear differentiation between the Social Value Groups risk perceptions regarding nutrition. A high fat diet was raised most frequently as a perceived risk. This risk perception was shared by all three Social Value Groups. Salt was also mentioned on several occasions by Sustenance driven respondents. There was a suggestion that Sustenance respondents felt that healthy foods were more expensive.

If one group were to be pulled out as expressing the most concerned albeit marginally it would be the Outer directed consumers. They also raised the issue of exercise more than any other Social Value Group.

5.3.4 MISLEADING INFORMATION

Misleading information was pulled in with nutrition in the factor analysis probably because of a link in consumers' minds to nutritional labelling.

On reflection using the words 'misleading information' in the questionnaire leads the respondent. It is therefore not surprising the respondents perceived this as a risk and expressed their concern

5.3.4.1 SUSTENANCE DRIVEN RESPONDENTS

"I think it is new products that are sometimes misleading" (S12)

"Things like these energy drinks that have been in the news recently not telling the truth about how much sugar is actually in them" (S12)

"I suppose I don't always read the nutritional information on the back, but I do look at it. It does worry me though to think that there is something I should know about that isn't given or that things are misleading" (S2)

"Yes the healthy option isn't always the healthy option" (S14)

5.3.4.2 OUTER DIRECTED RESPONDENTS

"Sometimes you see packaging and they are very clever they give you this description of a food ... and say high fibre or something and its not until you get home that you realise that sugar is the biggest ingredient" (O13)

5.3.4.3 INNER DIRECTED RESPONDENTS

"I think it is a good idea that information be policed so that you know that the information you get is accurate" (I12)

"they try and make out that they (oven chips) are healthy" (I2)

"you can read the label but there is often not enough information there to make an informed choice" (I2)

"Some (supermarkets) use manipulative techniques when marketing food....."
(I11)

"....you need information to make / be able to take responsibility for things" (I3)
"No I think things are misleading things like cereals being low in fat but they
don't mention that it is high in sugar" (I10)

SUMMARY

All interviewees stressed how important reliable and accurate information was. They suggested that the more information the better. The responses indicated that the Inner directed consumer perceived information to be fundamental to decision making.

5.3.5 CHEAP / POOR IMAGE AS A RISK

5.3.5.1 SUSTENANCE DRIVEN RESPONDENTS

"I look for bargains but at the same time I wouldn't buy really cheap food because a lot of the time you get what you pay for. I have bought cheap food and regretted it" (S2)

"I don't like to spend over the odds for things, I like a bargain but I wouldn't buy rubbish because it was cheap" (S10)

"I live in Rotherham and I think that they (Asda/Aldi) are the best value for money supermarkets in my area. Asda do a lot of buy one get one free, and things like that" (S11)

"Yes it really doesn't bother me (shopping at Aldi, Netto, Kwiksave)" (S14)

- "If something is cheap it doesn't mean it is not good quality" (S10)
- "I tend to look for the best value for money, I get the bulk of my shopping at Asda and Aldi but if I wanted to buy something a little bit special I would perhaps go to somewhere more expensive and for that its good and I am not worried about that" (S11)
- "I like good value but you get what you pay for at the end of the day" (S13)
- "You can't beat Aldi for tinned stuff to be honest it's cheapest" (S11)
- "The only thing I wouldn't buy is cheap meat" (S11)
- "Yes I do use them (own label products) I am happy that if you go to a good supermarket their products will be good but you aren't paying for the name" (S12)
- "I like Aldi for some things you can't buy everything from there but I think they are quite good" (S15)
- "The financial saving and that it (Aldi) is as good as anywhere else for the products that I buy there" (S15)

SUMMARY

Very financially orientated. Consciously looking for 'value for money', 'the best buy"

5.3.5.2 OUTER DIRECTED RESPONDENTS

- "... the image of the product might make me buy or not buy it" (O1)
- "Reputation is more important the image." (O1)
- "Well, I wouldn't shop at Lo cost or somewhere like that because I know they don't have a very good reputation, I can't imagine that they are very strict on hygiene or quality" (O1)
- "I wouldn't go there (to Netto, Kwiksave) so I guess I don't like their image" (O13)
- "If you go into Kwiksave or cheaper supermarkets there are less things on offer and it is less attractively packaged, so it doesn't spark your imagination and it's just not enjoyable if you go to Tesco and Safeway they give you menu suggestions things look nicer and you get inspiration" (O13)

"(shopping at Aldi or Netto)....the quality isn't as good although I wouldn't rule it out if I was in that economic range of buying. If I didn't earn very much money then I would shop there I think, but I can afford to shop at a more reasonable level so I don't need to go there" (O11)

"They are good if you haven't got much money aren't they" (O11)

"I don't feel bothered or worried about using a cheaper supermarket brand I still feel that they are of quality" (O11)

"We use Morrisons, Asda, Safeway that sort of supermarket rather than your lower range supermarket" (O11)

SUMMARY

Image and reputation are both worries for the Outer directed consumer. All respondents were overtly worried that both the food product (inclusive of brand) and the place of purchase have both the right 'image' and a good reputation. Outer directed respondents are concerned about cheap whether brand of product or supermarket (they appear very brand conscious).

5.3.5.3 INNER DIRECTED RESPONDENTS

"Heavily packaged food and generally messed about with food is generally cheaper......the nutritional quality is probably lower....."(I11)

"Cheap means those things to me (additives / highly processed food) (I12)

"I don't worry about what it looks like, status doesn't come into it at all" (I13)

"If it was full of additives yes a cheap product would worry me" (I10)

"I usually do yes (buy own label products)" (I10)

"no I tend to stick to Aldi and Sainsbury's every week" (I10)

SUMMARY

Whilst Inner directed respondents perceive 'high cost' as no more or less of a risk to 'low cost', the word 'cheap' appears to generate a belief that the product in question will be

nutritionally poor. This is interesting as in contrast to the Sustenance driven consumer whom appear to associate the word cheap with a financial saving.

5.3.6 ECOLOGICAL / ETHICAL RISKS

5.3.6.1 SUSTENANCE DRIVEN RESPONDENTS

"I don't mind the idea of recycling but it is the thought that someone is making a lot of money out of it and they are not doing it for the environment they are doing it to line their own pockets" (S14)

"I think when you see the stuff littered in the street that doesn't disintegrate it's not only an eye sore but it goes into the land fills" (S12)

"Yes we buy these things that you top up with the bag again and again"

"I am generally concerned I like to try and use bottle banks. I am not an activist or anything like that but in general terms I do care" (S10)

"It concerns me that it (recycled packaging) is often more expensive and I can be put off by that" (S12)

"I do try and recycle because my sister lives in an area where they actually collect so I take them to her but I don't go to a supermarket and think I will buy that because of the packaging" (S15)

"No I don't look for recyclable packaging" (S11)

"I'm not very good, I do try, I have good intentions, I am sure I could do much more (recycling)" (S13)

"They are things (controversial operations) that I don't really take much notice of I don't know about them I guess" (S11)

"The way the labour market is controlled they don't get very good pay compared to the price it sells at and that worries me because they just aren't benefiting by it" (S12)

"They do concern me a bit (pesticides and fertilisers)" (S12)

"Yes I think they are worrying (pesticides and fertilisers) I don't think they are safe because I think a lot of people abuse the rules" (S14)

"I don't eat beef I am concerned about BSE but what about the pesticides they use on vegetables that you have got to eat?" (S15)

SUMMARY

Sustenance driven consumers responses do demonstrate a moral concern in this area. Sustenance driven consumers did, however, indicate that they found it very difficult to discuss this set of risks in any depth. They relate to them by referring back to previously discussed areas of concern notably to financial concerns, their fear of the unknown and their fear about rules being broken. Whilst they do indicate verbally that they have concerns, ecology and ethical issues don't appear a high priority.

5.3.6.2 OUTER DIRECTED RESPONDENTS

"I think it is trees and wood and paper the actual disposal of it which is bad for the environment, apparently, so I try to do my part" (O11)

"We have a recycling bank near to where I live and I tend to go on a fortnightly basis and take bottles" (O12)

"I used to live in Cardiff and they have a really good recycling system I don't know of anything like that in Sheffield" (O13)

"I suppose I'm unethical, it's (controversial operations) just something I don't really think about" (O1)

"most of the pesticides and fertilisers I assume that they have been tested much more... I would think they would be safer" (O11)

"I don't know lots (about pesticides and fertilisers) but I know enough to be wary" (O13)

SUMMARY

Outer directed respondents display significantly less concern compared to Sustenance driven and Inner directed consumers. These issues just don't really appear to be on the Outer directed consumers' agenda.

5.3.6.3 INNER DIRECTED RESPONDENTS

"There is far to much of it (food packaging)"(I11)

"All food packaging concerns me, People need to eat. I don't see the need to wrap it up and make it more attractive" (II1)

"(Packaging) is something I am always aware of and again I worry about. So I buy things that have packing which can be recycled." (I3)

"I do try to be eco friendly however the provision is awful I don't have a car, driving would pollute anyway so I don't know if I would really be helping or just contributing but I collect and collect" (I13)

"Yes it (pesticides and fertilisers) is an environmental concern I don't like the idea of it getting into the food chain which is a big problem I don't like waste either" (I3)

"Because of the lack of control (in respect to pesticides and hormones) it's often hard to know what exactly you can do to make a difference" (I3)

"(Recyclable packaging) I don't think it is as important as things like not using your car, I think its crazy, like using your car to take bottles to a bottle bank, if you think about the amount of energy that is actually saved I don't think as an environmental issue on the scale of things that you make a difference" (I10)

"All three of these things (included pesticides) have very far reaching

consequences for the environment and the quality of our future for pollution and the long term health of the species" (I12)

"Yes it does (worry me). I try to avoid them (companies that operate controversially) if I am aware and it is a big issue" (I10)

"Its not the country so much or even the politics, may be it should be but for me it's the way they treat, conduct themselves. I haven't got a problem with eating meat I'm not a vegetarian I just feel strongly that we should not abuse animals for an ultimate pleasure" (I13)

"I do buy Cafe direct I do have these things (controversial) at the back of my mind" (I12)

"It worries me a great deal how supermarkets operate and food manufacturers. Some use manipulative techniques when marketing food its crazy that people get away with it" (I2)

SUMMARY

Inner directed respondents expressed concern about whether packaging is recyclable or not. There also appear to be concerned regarding over-packaging of goods. One respondent was concerned that they often felt they simply just couldn't avoid over packaging. Strong concern was expressed about pollution and the environmental dangers of by-products in the food chain. The long term effects of fertilisers and pesticides were a concern.

All Inners indicated that they were worried about origins of food products and/or the ethics of organisations. Likewise that they would be against organisations operating controversially. Inners expressed knowledge giving examples of organisations and / or countries that they believed to be unethical. (derogatory references to organisations / company have not been used in this thesis)

Inner directed respondents appear much more 'environmentally' aware. They appear to view issues more holistically e.g. it is not of any environmental benefit to take your empty bottles to a bottle bank if you have to drive to get them there.

5.3.7 UNFAMILIAR AS A RISK

5.3.7.1 SUSTENANCE DRIVEN RESPONDENTS

"Yes I buy similar things all the time" (S12)

"I tend to stick to what I know, I just buy things that I know we like" (S11)

"I buy very similar things I am very stuck in my ways though I did try a broccoli quiche quite recently that was quite daring but that is a rarity" (S15)

"Yes but I tend to shop at the same places" (S12)

"No we tend to use the same places" (S13)

"If somewhere new opened I would try it but I would put the same things in my trolley" (S15)

"No that isn't a problem for me, I'll try somewhere new" (S14)

SUMMARY

The Sustenance driven respondents were on the whole creatures of habit and reassured by familiarity.

5.3.7.2 OUTER DIRECTED RESPONDENTS

"Well yes (happy to try a new store), but that's not to say that I would" (O1)

"I do tend to buy the same things but if something new came out or I had heard
that a different variety of product was good I would buy it. It's not a case of
looking for something but if I see something I would try it" (O11)

"I am not totally set in my ways" (O12)

SUMMARY

Though the Outer directed respondents appear to often stick with the same product / brand/ supermarket they suggested that they were very interested in anything 'new' perhaps that they are influenced by fashion and trends.

5.3.7.3 INNER DIRECTED RESPONDENTS

".... it's fun as much as anything" (trying new things) (I3)

"I do all the time" (try new and different things) (I13)

"No I will try anywhere (new)" (I10)

"No I try them (new supermarket / ingredients) if I don't like them I won't have them again" (I11)

"I would guite happily buy a different product" (I12)

SUMMARY

Very little fear was evident with regard to the unfamiliar. Trying different shops, foods and brands was typical behaviour for the Inner directed respondents.

5.3.8 NON ORGANIC

Through the factor analysis, non-organic, was grouped with risks in the "Unfamiliar" category. The reason for this is not entirely clear cut and therefore responses to this particular risk will be discussed separately.

5.3.8.1 SUSTENANCE DRIVEN RESPONDENTS

"I think all fruit and veg should be organic but the price needs to come down" (S2)

"I don't buy it because of the cost" (S2)

"No it's not something I think about" (S11)

"Its not really a priority I suppose" (S13)

"If it got cheaper" (S13)

"I have a friend who buys from an organic green grocer and has it delivered I think it's good but it's too expensive" (S15)

"If I thought I could afford organic I would but to me cost is the priority" (S12)

"It (organic food) was (pause) tended to be expensive and some of it looks wilted because nobody else bought it" (S13)

SUMMARY

For the Sustenance driven respondents the organic issue comes back to the matter of cost and financial worry. Sustenance driven respondents perceive organic produce to be too expensive. This appears to override any ethical desire to purchase 'organic'.

With an increasing number of retailers expanding their organic range at lower prices perhaps Sustenance driven consumers may be more willing to purchase organic produce

in the future. Naturally larger ranges and wider availability will make it more familiar also.

5.3.8.2 OUTER DIRECTED RESPONDENTS

"I would do, but I don't (buy organic produce). If I sat back and had the time to think about it, but I think just because it is the way of life you go to the supermarket and you buy what's there" (O12)

"No it's not something that bothers me (organic), if there was a difference in the taste and it was better I might do more than I do now, but at the moment what I eat doesn't seem to have damaged me so why change it" (O11)

"It is easier to buy it (non-organic food) straight from the supermarket" (O13)

SUMMARY

Outer directed consumers do not appear to include organic produce in their priorities. They seem to suggest that organic foods are not readily available and as a result don't satisfy their need for ease and convenience.

Outer directed consumers may however, as a result of increased positive publicity about buying 'organic' 'warm' to the idea of being seen to be purchasing products associated with the "latest trend".

5.3.8.3 INNER DIRECTED RESPONDENTS

"People like myself who tend to go for the organic route tend to have a more ethical approach to things anyway" (I3)

"I go into Tesco's usually at reduced price time admittedly to whip up all the organic stuff but it worries me because we have a huge inequality in the availability of good quality food" (I11)

"(Organic) I try to find them reduced or on offer. I guess the logic is that if they were produced on a bigger scale the prices would come down and made widely available. I would pay a bit extra, I accept that it is more labour intensive you

need more land and you need to rest the land and other things and because of that I would be happy to pay say up to 20% more for that I think seems reasonable" (II1)

"we have started to grow a lot of our own (organic food). I heard on Radio 4 about how supermarkets are thinking about going into this in a big way that would be great" (I12)

"I guess it (organic production) should be the way but our food system and the demand for cheap food (mean) it's not possible" (I13)

SUMMARY

Whilst Inner directed respondents appear to be actively seeking to purchase organic produce they do stress concerns about over pricing. Several Inner directed respondents claimed that they purchased organic produce when it was on offer/reduced or grew their own.

Inner directed consumers are likely to welcome an increase in the availability and reduction in the prices of organic produce, however characteristically they may well deplore (ironically) the high amount of advertising, marketing and packaging that supermarkets may undertake to market their organic food ranges.

5.4 INTERVIEW RESPONSES TO FOOD RISK RELIEVING STRATEGIES QUESTIONNAIRE

The following section aims to analyse the risk relieving strategy responses from Interview Three. Interview Three was designed to explore, clarify and verify the responses to Part Three and Four of the Survey Questionnaire. This part of the qualitative analysis chapter focuses on Risk Relieving Strategies (part four only). The interview responses are from the fourteen respondents obtained from the main survey sample.

As with the previous section the following section is structured so that the reader can refer back to the quantitative analysis with ease. The section headings follow the factor analysis labels previously detailed in Chapter Four.

5.4.1 SOCIETAL ACCEPTANCE / ENDORSEMENT

5.4.1.1 SUSTENANCE DRIVEN RESPONDENTS

"Its like anywhere if you go to a restaurant you go because someone has recommended it" (S12)

"If you have a recommendation about something you would try it anyway" (S12)

"I tend to trust my family's judgement on things, yes I guess I would take their views on board" (S12)

"She (my mother) was really fussy about food so anything she said I would think it must be alright then" (S13)

"People that I am friendly with, and my family are, I suppose similar to me and I value them and trust them, particularly my sister who is a Home Economics teacher" (S15)

"I would trust my family's opinions much more than I would an experts because I think, what exactly is an expert?" (S15)

"(Family) might encourage me to try something" (S13)

"Yes a family member or a friend if they said they had tried it and it was good" (S14)

"I had seen them advertised I thought I will try them" (S12)

SUMMARY

Sustenance driven respondents appear willing to quite readily trust others. Responses suggest that they do feel less worried when a product is endorsed by others close to them. Evidence supports the view that the family is a great influence and comfort to the Sustenance driven respondents.

5.4.1.2 OUTER DIRECTED RESPONDENTS

- "Yes, I would have thought that if they had tried and tested it and said it's alright, it's word of mouth again, if somebody tries it and passes it on, their opinion whether it is good or bad....." (O11)
- "A person similar to myself, it's just if they are similar to me so I think that they have got the same taste as me like the same sorts of things" ()11)
- "I think I trust other people's opinions, I do trust friends' opinions" (O12)
- "It depends on whether I thought they had any knowledge (friends)" ()13)
- "I guess I am (influenced by advertising)" (O12)
- "We never go somewhere "dodgy" we look in the local papers and magazines" (O10)
- "Usually through reputation, what people have told us" (O13)
- "(Advertising) well it does effect us whether we like it or not" (O13)
- "Well you think I have seen it, the product name or the jingle or whatever is in your head and you think I will try it" (O13)
- "I tend to go for more expensive brands because you tend to get brain washed in a way by advertising" (O12)

SUMMARY

As with the Sustenance directed respondents societal endorsement appears to be an effective risk relieving mechanisms for the Outer directed respondents however, the focus appears to be more external to the individual and away from the family. They appear ready to turn to people similar to themselves and admit to being influenced and interested in information communicated via advertising.

5.4.1.3 INNER DIRECTED RESPONDENTS

"This is certainly not the case, some people that I know eat the most horrendous things and really badly. I make up my own mind" (I11)

"If it is someone (a peer) who I trusted and thought that they were quite knowledgeable" (I10)

SUMMARY

Evidence suggests that societal acceptance / endorsement is largely an ineffective strategy for relieving risk for the Inner directed respondents. Inner directed responses were generally very negative with regard to this area. The only positive response was in respect of a trusted and knowledgeable peer.

5.4.2 LOW COST

5.4.2.1 SUSTENANCE DRIVEN RESPONDENTS

"Yes if it was cheaper, it would" (S14)

"I would be happy if they were cheaper....I'd rather save the money" (S10)

"If it was discounted I might look at it" (S13)

"I tend to look for the best value for money" (S11)

SUMMARY

The previous section highlighted how low cost was not perceived as a risk for the Sustenance driven respondents. In line with this Sustenance driven respondents appear to perceive a reduction in cost as an effective strategy for relieving risk.

5.4.2.2 OUTER DIRECTED RESPONDENTS

Outer directed respondents responded negatively to the proposal that low cost would relieve risk perception. This is not surprising since this group previously indicated that they perceive low cost as a risk.

5.4.2.3 INNER DIRECTED RESPONDENTS

"Low economy brands tend to be full of additives and god knows what else" (I11)

SUMMARY

As highlighted in the previous section Inner directed respondents indicate that cost, whether it be low or high is not as a reason to perceive risk per se. Similarly Inner directed respondents do not see 'low cost' as a effective risk relieving strategy. The only response here was that low cost was associated with foods which are nutritionally poor.

5.4.3 BRANDING (PRODUCT ORIENTATION)

5.4.3.1 SUSTENANCE DRIVEN RESPONDENTS

"If I have used something before and been happy with it I tend to buy it again and again and go on buying it. It gives me consistency......I think of Heinz as having consistency" (S10)

"You know what you are getting then (familiar brand)" (S11)

"I buy similar, the same things all the time you know the things you have tried and you like them so you buy them time and time again" (S12)

"I think it (free sample) helps in supermarkets that offer a taste of a new cheese it's silly to buy a pound of cheese when it is so expensive these day" (S12)

"You get to trust it, you know that you like it, if you are happy with it you stick to it" (S13)

"reputation I suppose it is the quality of the company trusting the company and the quality of the food....." (S13)

"I do try sometimes to not be so boring but it's me" (S15)

"I would look for a brand name that I knew, and where it was sold that's a big one" (S13)

"I do like to know what is in things" (S10)

- "I look at the portion the content size, additives as well" (S10)
- "Something like genetically engineered food it doesn't matter who's selling it even if it was M & S I still wouldn't feel happy to buy it" (S13)
- "Maybe a brand that I was familiar with but I am not really bought into brands that much" (S14)
- "I would want to know what is in it before I would try it" (S11)
- "It's a quality thing and familiarity as well" (S13)

SUMMARY

Branding (product confidence), particularly familiar branding appears to be a fundamentally effective relieving strategy for the Sustenance driven respondents.

Naturally they are inextricably linked, brands often being reassuring because they are familiar.

Interestingly however despite branding being a potential risk relieving strategy it does not appear to be powerful enough to override technological risk perception. Free samples were seen as a good way of reducing financial risks for this group.

5.4.3.2 OUTER DIRECTED RESPONDENTS

- "You just know what (brands with a good reputation) you are getting" (O10)
 "I think it is about the good reputation again you assume that they are doing a lot of research into the product before they put it on to the market and because of the name you are expecting good quality" (O11)
- "You need to be able to read the label and know what you are getting" (O13)
- "It wasn't the usual brand we didn't trust it so we threw it away without even opening it" (O13)
- "As long as it (the label) has the essential information like what it is who the manufacturer is and what is in it I am happy" (O13)

SUMMARY

Brand reputation / image is fundamental strategy for Outer directed consumers and repeatedly mentioned as a key component in their food choices. They also read labels to reassure themselves about product quality.

5.4.3.3 INNER DIRECTED RESPONDENTS

"You can make your own mind up if you have all the information that's important" (I10)

"We assume that labelling is giving information and you can then make an informed choice" (I11)

"I don't like to buy something unless I know what is in it, then you are equipped to know if you should be worried or not, it lets me make a decision" (I13)

SUMMARY

Labelling appears to be fundamentally important for Inner directed consumers. They feel empowered by information and believe it to be their right to be informed. Information facilitates decision making. They appear to be autonomous when it comes to decision making.

5.4.4 HIGH COST/EXPENSIVE AND IMAGE ASSOCIATION

5.4.4.1 SUSTENANCE DRIVEN RESPONDENTS

High cost was perceived as a risk for the Sustenance driven consumers and therefore was not perceived in anyway as a risk relieving strategy.

5.4.4.2 OUTER DIRECTED RESPONDENTS

"I do tend to go for more expensive brands because you tend to get brain washed by advertising" (O12)

- "The price relates to the quality" (O11)
- "I do think you do get a better product" (O12)
- "I don't think I would buy meat from the cheaper supermarkets" (O12)
- "They often taste better (more expensive products)" (O12)
- "I would go for the best quality one I think" (O11)
- "Its just that there is more choice and the way it is packaged in a more expensive supermarket, I just prefer it" (O13)
- "You get what you pay for" (O10)

Summary

Outer directed respondents clearly link price with quality. They appear to feel reassured by a high price as such it is perceived as an effective risk relieving strategy.

5.4.4.3 INNER DIRECTED RESPONDENTS

There is no evidence to suggest that Inner directed respondents perceived 'high cost' as an effective risk relieving strategy. They only presented disbelief that it might even be suggested to be an effective risk relieving strategy.

5.4.5 AUTHORITARIAN REASSURANCE

5.4.5.1 SUSTENANCE DRIVEN RESPONDENTS

"All these scares over the years with eggs and beef and things you are told you should and shouldn't eat I don't think the government should be involved in that. I think there should be an independent unit, not politically motivated and then it should be allowed to go into places where food is produced and have a lot more say in what is happening and it should be regarded as the authority of that. A government should do that for the good of the country but it seems to me that they have no idea what is actually happening compared to the people who are working day in day out, so I think there should be an independent unit..." (S12)

"I don't know really, I think maybe the government has it's own agenda" (S10)
"Experts as well they might be experts it certain fields but if they came together that would make me worry less if I thought there was a team of experts" (S12)
"something.... on the side of the consumer that are not afraid to speak out" (S13)

"What sort of expert is a government body. I'm not impressed at all. The last government tried to tell use that beef was safe. Probably now beef is safer than it has been for years. It's the beef burger that I had in 1979 that I worry about, that has probably done it for me" (S15)

"I don't entirely trust the government recommendations, they seem to fudge the issue most of the time they seem to be on the side of the industry" (S13)

"Yes but it would depend who the independent tester was" (S14)

SUMMARY

Responses suggest that the Sustenance driven respondents were fearful, additionally there is a suggestion that they don't totally trust the government, (that the government is out of touch). They suggest that they believe that the government was not really on the consumer's side.

5.4.5.2 OUTER DIRECTED RESPONDENTS

"It makes sense that independent approval should be trusted over government.

After the BSE the government can't be trusted"(O10)

"I think so, yes" (feel relieved by Government) (O12)

"I guess I would trust the Government or an expert more they are going to have more information" (O13)

"Sometimes experts say things just to take the worry off people when really they know what the truth is so you don't know what their motive was" (O11)

"I don't think I would (trust experts) as much now actually, with everything that has been coming out in the last years" (O12)

"These sorts of independent survey are more to do with quality taste and price than food safety. I would probably think that the Government might have more know how, (if it's a safety issue)" (O13)

SUMMARY

Outer directed respondents responses suggested that they were inclined to be reassured by the Government / experts but were a little hesitant to admit this. They suggested that their reluctance was due to issues such as the BSE crisis which had led to a decrease in their levels of trust in recent years.

5.4.5.3 INNER DIRECTED RESPONDENTS

- "These people (experts) know. They are better equipped than anyone to make decisions" (I13)
- "Yes, I think they (independent body) are more inclined to listen to the experts and tell people the way it is" (I13)
- "It would have to be an independent research body it is difficult" (I10)
- "I don't think so (trust the government) particularly in light of the BSE, no" (I10)
- "I would feel much more inclined to believe them if they were independent" (I10)
- "It would have to be an independent research body" (I10)
- "They are not independent (government). They have been in the pocket of the farmers for ages" (I11)
- "I trust them (independent body) slightly more than the Government, I suppose I assume that everybody is in the pocket of somebody" (I11)
- "No not at all (relieved by Government body). They tend to ignore and select information. They say what ever gets them the vote" (I13)

SUMMARY

Inner directed respondents are particularly aware of the 'potential ends' to which expert knowledge can be turned. They therefore distrust government bodies and are more likely to be reassured by an independent body.

5.5 QUALITATIVE ANALYSIS SUMMARY

The results of the qualitative analysis indicate clearly that within the themes discussed, whether they be those relating to food risk perception or those relating to risk relieving strategies, there are distinct differences between the views concerns and approaches of Sustenance driven, Outer directed and Inner directed respondents. These are supportive of the findings of the quantitative analysis in chapter four. The findings of both chapter four and five will be discussed in chapter six.

CHAPTER SIX

DISCUSSION

The discussion chapter will consider the overall findings of the research in the context of the literature discussed in Chapter Two. The Chapter is in four parts, the first presents a discussion of the research methodology highlighting its strengths and acknowledging its limitations. The second part discusses the key findings of the research focusing on both quantitative and qualitative methodologies. The third part discusses theoretical and philosophical issues which arise as a result of an examination of the findings in the context of the original literature reviewed. The final part of this chapter makes recommendations to the food industry in light of the research findings.

6.1 DISCUSSION OF THE RESEARCH METHODOLOGY

6.1.1 USING QUANTITATIVE AND QUALITATIVE METHODS

For both the pilot and the main study strong parallels are evident between the results of the questionnaire survey and the results of the qualitative interviews, illustrating the value of using both quantitative and qualitative approaches.

The results of the qualitative research illustrate clearly how it would have been possible for inaccurate assumptions to have been made had quantitative methods been used alone. Fundamentally the qualitative interviews complemented the quantitative responses by highlighting consumer motivations. This was critical. Whilst a risk may be of equal concern to the members of two different Social Value Groups, the motivation for those perceptions may be fundamentally different. For example, the research clearly shows that both Inner directed and Sustenance driven consumers perceive risk in relation to the use of pesticides and fertilisers. However, whilst the degree of worry is comparable for

the two groups, the motivation for these is very different, a reflection of their dissimilar worldviews.

The qualitative responses illustrated that whilst the Sustenance driven consumers seem to fear the 'unknown', the Inner directed consumers feared the long term global effects that pesticides and fertilisers may have on the environment. Importantly the Inner directed consumers demonstrated that they were not concerned about the 'unknown', indeed they make it their business to know and are also highly innovative consumers. This example illustrates how the qualitative interviews are crucial to the valid interpretation of quantitative findings.

The qualitative research was also very important in validating the use of the Social Value Group Screening Instrument as an appropriate tool for categorising respondents into Sustenance driven, Outer directed and Inner directed (SVG) groups

6.1.2 METHODOLOGICAL LIMITATIONS

6.1.2.1 A CONVENIENCE SAMPLE

It was decided for reasons discussed in Chapter Three, that the sample for the main study would be women aged 25 - 45, with no dependent children, economically active and in an household with an income greater than £12,000 p.a. Furthermore the sample of women used in the main study was drawn from staff at Sheffield Hallam University. This being the case the results of the study cannot be generalised to a wider population. The results do, however, indicate that for the group investigated there is a clear relationship between social values and perceptions of food risk and preferred risk relieving strategies. It could be argued that, in hindsight, the approach to the selection of a sample for the main study was perhaps over cautious. Indeed the outcomes of the pilot study which was conducted on a mixed group of respondents perhaps add some weight to this view.

6.1.2.2 LINGUISTIC AMBIGUITY

There is some indication that the questionnaire presented a small number of linguistic limitations. The words 'cheap', 'image' and 'non – organic' appear to be ambiguous having a range of interpretations, these dependent on the Social Value Group to which the respondent was allocated. Qualitative Interviews gave the opportunity to clarify and increase understanding in these areas.

6.1.2.3 SELECTION OF PARAMETRIC TESTS FOR QUANTITATIVE ANALYSIS

Parametric tests were selected when analysing the quantitative responses to the survey questionnaire. Greene and D 'Oliveira (1995) highlight that this is often justifiable in research where psychological variables are being investigated using Likert scales. The assumption being made that the intervals in such scales are equal. Davis (1999), Goldsmith et al (1997) and others in the field provide examples of social value research which successfully adopt Likert scales and subsequently use parametric tests for analysis of the data generated.

Whilst a preliminary screening of the data obtained in this research also led to confidence in using parametric tests, it is possible to debate whether the criteria for using them were met fully. In particular it was felt that for the purposes of the one way analysis of variance undertaken it was appropriate to use equal numbers of respondents in each Social Value Group to guard against any problems associated with the requirement for homogeneity of variance between conditions not being fully met. In addition where possible, the results of equivalent non – parametric tests were examined in an attempt to highlight any question over the use of parametric ones. All results were comparable and consequently the results of parametric analysis are presented as they enabled the use of values rather than ranks, the latter being less enlightening.

6.1.2.4 SOCIAL VALUE GROUP MEMBERSHIP

The scoring system used to allocate respondents to Social Value Groups might be regarded as crude by the 'purist'. Aware of the weaknesses in terms of scientific credibility, members of each group were selected with extreme caution. Only those exhibiting dominant values of one of the Social Value Group were selected to be a member of that group.

Earle and Cvetkovich (1994) cited in Cvetkovich and Lofstedt (1999) use a comparable screening device for the allocation of respondents into cultural-value groups in their research. Respondents were similarly given a score for each value set, Earle and Cvetkovich (1994) then (unlike this piece of research) categorised all respondents into one of the cultural-value groups. Taking a rather more cautious approach this research took the view that Social Value Group membership is neither exclusive nor certain, with many respondents demonstrating a combination of values drawn from more than one Social Value Group. With an aim of obtaining clear cut results it was decided to select respondents demonstrating only one set of dominate values for those elements of this study that focused on differences in perceptions between the members of the three Social Value Groups. It would however be interesting to investigate whether this strategy was again perhaps over cautious.

6.1.3 MARKET / CONSUMER SEGMENTATION

As discussed in the literature review much of the food risk research undertaken in the UK tends to present gross generalisations of the risk perception of 'The British Public'. Research by the Food Standards Agency (2000) and by the Department of Health (1998) are unfortunately examples of such research, of which there are many.

Whilst it is valuable to explore general issues within a society, it is misguided to assume that actions based on the results of such research can be effective and efficient in the context of a whole population. For policy makers in particular this simplification of what

are complex and fundamental differences within the British population is an attractive and preferable approach due to its strategic manageability.

At the other extreme, in marketing research in particular, there is sometimes a tendency for research to generate segmentation categories that have little value beyond the narrow focus of the application being considered.

The Literature Review discussed which variables are 'best' for consumer segmentation. Traditional variables for segmentation such as demographic, socio-economic and geographic were examined and found to be wanting in many respects. Value segmentation was presented as a favourable alternative. The findings of this and other research strongly supports the usefulness of using social values to explore consumer perceptions. The results not only suggest that social values influence consumer perceptions but also that analysis of these social values over time can help to identify trends which in turn predict the future demands of consumers. For those whose aim it is to manage risk perceptions such data can be a tool for improving the effectiveness of risk perception management.

6.1.3.1 SOCIAL VALUE GROUP SEGMENTATION

The literature review draws particular attention to Taylor Nelson's Social Value Groups as reported by MacNulty (1985).

Nelson (1986: 198) describes the three Social Value Groups broadly and simplistically in the following way;

"....inner directed, more concerned about the individual and creative aspects, the status-conscious outer – directed ones, and those whose primary concern is to keep their head above water the sustenance directed".

The results of this research clearly confirms the relevance of these groups for segmentation in Britain today and indicate that members of each group have a dissimilar portfolio of risk relieving strategies and risk perceptions in relation to food.

6.2 SUMMARY OF SOCIAL VALUE GROUPS GENERAL WORLD VIEWS

The following observations were made with regard to members of each of the Social Value Group's personal characteristics. This was as a result of both qualitative and quantitative responses.

6.2.1 SUSTENANCE DRIVEN CONSUMERS

Respondents categorised as Sustenance driven demonstrated the least confidence of the three groups, often seeming unsure and fearful. They often found questions difficult to answer, looking for prompts from the interviewer. Their responses indicate that they have limited knowledge particularly in the area of technology and ethics / ecology. Despite these issues they were the most willing to participate in the research, accommodating and generous with their time.

6.2.2 OUTER DIRECTED CONSUMERS

Respondents categorised as Outer directed demonstrated confidence. Rather chameleon like, and sometimes self-contradictory there is some evidence to indicate that this group of consumers wanted to be perceived favourably in the interview situation and that their responses may be somewhat biased on this account. In an attempt to avoid this, serious attempts were made to present an environment and atmosphere which would allow the interviewee to feel relaxed and, as such, free to give honest responses.

6.2.3 INNER DIRECTED CONSUMERS

Respondents categorised as Inner directed, demonstrated that whilst they see themselves as playing a relatively small part in the context of 'the whole', they believe that they have a contribution to make to 'the system'. Fundamentally they believe in both accountability and responsibility. They have a deeply embedded respect of 'the system'. They demonstrated a breadth and depth of knowledge in many areas. They appeared sincere

and were assertive, articulate and enthusiastic. Exhibiting complex value systems, there were times when they appeared frustrated particularly during Interview Three when risk relieving / avoiding strategies were discussed. Both holistic and analytical they often questioned themselves and others. The Inner directed presented the largest group of the main study sample, probably a higher proportion than would be found in the general population.

6.3 SOCIAL VALUES, AND THEIR CORRELATION WITH RISK RELIEVING STRATEGIES AND RISK PERCEPTIONS

Fundamentally the results (detailed in Chapters Four and Five) illustrate the relationships between social values, risk perception and risk relieving strategies. The results support the view that a risk is perceived when values / lifestyles are threatened and that consumers select both risk and risk relieving strategies largely to support their 'worldview' as discussed below for each of the Social Value Groups in turn.

6.3.1 SUSTENANCE DRIVEN CONSUMERS

Sustenance driven consumers clearly demonstrate a strong family orientation. Seeing work as a financial necessity as it provides security for the family. Disliking the unfamiliar, they appear on the whole to have a 'traditional' value system. The past both shapes and guides their future. This research suggests that the Sustenance driven consumer is most comfortable and most happy in a 'safe', harmonious environment that is secure and where 'things' rarely change. They appear to believe that traditional institutions and authority have a duty to care and protect.

This value system manifested itself clearly in both qualitative and quantitative responses. Sustenance driven consumers perceive risk when they are confronted with something unfamiliar and when the status quo is threatened. Appearing to have limited knowledge in the area of technology and ecology, the findings demonstrated that they appear to

adopt one of two strategies, they are either dismissive or fearful. They are ready to assume that they will be protected by an institution external to themselves. They feel relieved of their worry, on a somewhat superficial level, if a member of their family isn't worried, this demonstrating the high level of trust they have within the family unit. The results also suggest that Sustenance driven consumers fear financial risk. Financial risk was avoided by 'careful' selection of store and products. Equally, as one might expect a low price or a reduction in price was an effective risk relieving strategy for this group. Financial risk was repeatedly referred back to even when the discussion was in a very different area thus demonstrating the Sustenance driven consumers orientation to these concerns. Financial risk perception appears to be so strong that it overrides all other risks and risk relieving strategies. Security would appear to be the core value that these respondents seek to protect.

6.3.2 OUTER DIRECTED CONSUMERS

Outer directed consumers value themselves according to how they believe others value and see them. They want to be the best and therefore aspire to have the best. Motivated by financial and social status, the results suggest that they are most concerned about image and reputation. Whilst acknowledging risks they appear to distance themselves from these and as a result express relatively few perceived risks. Interestingly they appear ready to accept many of the risk relieving strategies investigated particularly branding, high price, image and reputation, and perhaps as a result of this, of the three Social Value Groups they seem to have the lowest level of perceived risk across the range investigated the exceptions being risks associated with 'cheap / image and 'nutrition'.

This group has the highest confidence in the government of the three investigated

6.3.3 INNER DIRECTED CONSUMERS

Inner directed consumers appear strong and independent. Perhaps the most complex group to analyse they are predicted by MacNulty (1985), as a consumer group to grow

in size. The results of the research points to a group of consumers who are deeply philosophical, analytical, reflexive and global/holistic in their approach. They appear to believe that their actions regardless of how small or seemingly insignificant will effect 'the whole'; as a result they adopt a moral and holistic philosophy in all aspects of their life. They appear to look to the future rather than the past when making decisions. The results identified that they share a number of risk perceptions with the Sustenance driven group, however the interviews indicate that the motivation for them is fundamentally different from that of Sustenance driven consumers. Whilst Sustenance driven consumers expect society's institutions to 'dutifully' protect them from risk, the Inner directed consumers reject this as an effective method of risk relief.

Inner directed responses indicate that they demand information to enable them to make informed decisions. Results suggest that Inner directed consumers do not automatically trust society's traditional institutions. Their cynicism and scepticism is repeatedly demonstrated and frequently focuses around the issue of the ways in which such institutions misrepresent information. They seek out knowledge preferably from independent sources to empower themselves, thus reducing the need to trust external / secondary sources with vested interests.

Perceiving many risks, particularly technology and ethical / ecology related risks, Inner directed consumers reject almost all of the risk relieving strategies investigated in the research. The only effective risk relieving strategy for this group appears to be information either factual or obtained from independent sources.

6.3.4 SUMMARY

Inners demonstrate the greatest degree and widest range of food related perceived risks with the Outer Directed consumers demonstrating the lowest levels and the smallest range of perceived risk. Interestingly for these groups, levels and range of risks perceived negatively correlate with the number of and degree to which risk relieving strategies are perceived as effective. We might interpret this another way. Inner directed consumers have few effective risk avoiding strategies in place (and those which

are effective require knowledge to be gained). As a consequence risk is perceived frequently and readily. In sharp contrast Outer directed respondents demonstrated a wide spectrum of risk relieving / avoiding strategies and as a consequence perceived much fewer risks. Sustenance driven consumers fall some way between these two extremes.

As the research developed the issue of consumers' management of their own risk perceptions was a major issue. Sustenance and Outer directed consumers seemed to be able to avoid perceiving risks by seeking reassurance from sources external to themselves. Inner directed consumers appear on the other hand to look to themselves i.e. they make up their own mind. Inner directed consumers it would appear are therefore motivated to acquire more knowledge than the other two groups with the possible outcome that they are much more aware of social and world-wide issues and are more confident in themselves.

6.4 CONSUMER STRATEGIES FOR LIMITING EXPOSURE TO PERCEIVED RISK (RISK AVOIDING STRATEGIES)

The view that risk relieving strategies instrumentally reduce levels of perceived risk is implicit in much of the literature. A rare example of an alternative approach is presented by Yeung (2001) who suggests that consumers attempt to limit their exposure to risk implying that the approach taken is preventative rather than reactive. Yeung's orientation is welcomed as it supports the findings of this piece of research.

This orientation is crucial both in terms of one's philosophical approach to understanding consumers risk perceptions and in terms of ones strategic approach to the management and intervention of them. The findings of this research raise a fundamental doubt in what appears to be, by and large, the accepted understanding that consumers firstly perceive risks and only then attempt to relieve or reduce them afterwards. This approach may be a result of much research focusing on either perceived risk or risk relief with a general

lack of recognition of the inseparability of the two areas. Likewise researchers aiming to manage perceived risks often look for quick remedies for management applications rather than carefully considering the processes involved.

The findings of this research strongly suggest that consumers have strategies which aid their decision making processes. These mechanisms may be subconscious but fundamentally they support and reinforce the consumer's social values, world view and lifestyle. Whilst capable of development and change these social values are embedded. The results of the research indicate that these mechanisms are used as a frame of reference and fundamentally determine whether a risk is ever perceived at all. Findings suggest that if a risk is perceived it is because the consumer has failed to adopt / find a strategy which will enable them to avoid the risk concerned.

The variable 'throwing food away' is an example of how an adopted strategy avoids risk perception in the first place. Quantitative results indicated that this was not a risk perceived to any great degree by any members of any of the social values groups. This was interesting as it had been hypothesised that both Inner directed and Sustenance directed consumers would feel concerned about throwing food away albeit for very different reasons. The quantitative results were explored during the qualitative interviews. Interestingly members of both groups indicated that whilst throwing food away was theoretically a risk for them, they had adopted a strategy to avoid ever having to do so and therefore no risk was perceived. Similarly respondents who indicated that they did not perceive BSE as a risk also indicated that they had taken action to avoid the risk (i.e. changed their purchasing behaviour).

This orientation questions the value of risk perception research which purely examines risk perception. This type of research may never offer more than observations. Risk perceptions are the consequence. As researchers we need to find the cause. To truly begin to understand the consumer's risk perception we have need to develop a knowledge and understanding of the strategies consumers adopt to avoid risk perception only then can we be proactive and manage the avoidance of risk perception.

6.5 RISK PERCEPTION AND RISK AVOIDING STRATEGY PORTFOLIOS

The results illustrate that consumers have a portfolio of perceived risks. The degrees of concern vary according to the risks perceived. These differing degrees of concern may reflect the extent to which the risk threatens the consumer's social values and the extent to which the value(s) threatened are fundamental to their lifestyle. It is thus possible to place the risks in a hierarchy, those at the top of the hierarchy are those which are the biggest threat to the consumer's most dominant values. These risks appear to overpower those lower down and are of the greatest concern. The chance to worry about risks lower down in the hierarchy is perhaps to neglect those higher. This is an explanation as to why Sustenance driven respondents always appeared to refer back to financial risk and the unfamiliar (their threatened core values) even when talking about technology or ecological / ethical issues. Inner directed consumers in contrast worry about ecological and ethical issues. That is not to say that Inner directed consumers do not worry about financial risk since they do not threaten their core values.

Perceptions of risks and risk avoiding strategies are deeply rooted in consumer social values, they are guided by the consumer's world view and are fundamental to how they manage the plurality of choices that they make on a daily basis.

6.6 TRUST AND RESPONSIBILITY

The research findings highlight the fundamental importance and intrinsic nature of both trust and responsibility for consumers' perceptions of risk and therefore risk avoiding strategies.

Trust is paramount to modern living. The dynamic and complex nature of today's society means that making choices is a necessity. Making a choice involves the assessment of what is a 'risk' and what / who can be trusted. Consumers are increasingly aware and

skilled at managing potential risk, those management strategies frequently require them to place trust in others.

Giddens (1991) interestingly describes trust by placing it at the opposite end of a spectrum to risk. Qualitative responses illustrated that trust presents itself as integral to consumers' risk assessment and decision making, level of trust appears to influence which risks are perceived and what constitutes an effective strategy for reducing / relieving or avoiding risk. Importantly the findings suggest that consumers with different social values trust different sources.

Cvetkovich and Lofstedt (1999: 5) suggest that because trusting involves a subordination of position through the relinquishing of control, trust also involves risk "One may trust someone because of a strong expectation that the trusted one will act in a certain way. But one can never be entirely certain that one's trust will not be violated. Trust then trades a primary physical risk for a social risk"

Who or what a consumer relinquishes control to, appears to be a reflection of their values. They need to feel a degree of empathy or commonality of values with 'the party' in order to justify trusting them and feel confident in their choice. If there is a conflict in values, trust will be at best strained. An alternative strategy is to retain control.

The concept of trust is complex. How consumers deal with the notion of trust is particularly interesting. The findings of this research suggest that Inner directed consumers are reluctant to make trade offs of the sort indicated by Cvetkovich and Lofstedt (1999) whereas Outer and Sustenance driven consumers in contrast appear much happier to do so.

Slovic (1993) recognised that "Trust is fragile. It is typically created rather slowly, but it can be destroyed in an instance". Building trust is extremely challenging yet fundamentally the aim of many strategies adopted by organisations within the food industry. Advertising, branding and labelling are just three areas where building trust is an objective, critical to long term success in fiercely competitive markets.

The British food industry has undoubtedly suffered a break down of trust despite the sophisticated strategies adopted by food managers. Consumer confidence and levels of trust in food systems are well documented and recognised as seriously low. Consumers trust in food and food systems has broken down most transparently in the last ten years. Whilst respondents indicated differing levels of trust for different sources, the research findings support a common lack of trust in Government agencies across all of the Social Value Groups (though Outer directed respondents demonstrated more confidence than the other Social Value Group members). The Food Standards Agency (2000) was launched in part to address consumers' low levels of trust by taking a strategic view of food safety and standards across the whole of the food chain importantly presenting itself as an independent agency. The research findings demonstrated that information from independent agencies was believed as an effective risk relieving strategy for consumers in all three Social Value Groups.

The Food Standards Agency proclaiming a strong consensus against over - regulation, stress that consumers must take their fair share of responsibility for food risks. Food Link News (March 2000) suggest that

"one of the FSA's tasks will be to empower the consumer towards informed decision-making. In order to reinvigorate consumer trust, the FSA will be striving to present as open a forum as possible by developing web sites and consumer help lines. There will also be an emphasis on two-way communication: consumer panels and public meetings will be encouraged and open lines of contact established with the media"

The research findings raise questions as to whether it is possible to regain trust that has been previously destroyed. The time required to rebuild trust, (if possible to rebuild

trust at all) would be substantial. Also consumers collectively have changed and continue to change so significantly that it appears highly likely that they would adopt alternative strategies rather than revert back to trusting those that have previously proven to be untrustworthy.

The Food Standards Agency do to some degree appear to recognise a more demanding and sophisticated consumer. The FSA's strategy being most appropriate for consumers who are adopting a more democratic approach to the assessment of risk. As consumers' values change and levels of empowerment increase so too does the demand for new means by which to gain the information which allows consumers to make informed choices. Communications technologies are a major facilitator of this process, which can however also be something of a two edged sword.

Inner directed consumers have values which are in conflict with the traditional institutions that consumers have in the past been persuaded to trust. This lack of compatibility has led the Inner directed consumer to adopt alternative strategies. When they cannot find a party to whom they feel comfortable relinquishing control, Inner directed consumers respond by keeping control, seeking information themselves and making the judgements and choices based on that information. Comprehensive labelling aids these processes as does the increasing ease of accessing information. Exit strategies may also be adopted such as purchasing organic food in response to perceived risk i.e demanding alternatives which do not threaten their social values. Food retailers are clearly responding to the demand for such exit strategies. It is perhaps more likely that increasing numbers of consumers will adopt these strategies than it is that trust in traditional sources will be regained.

Importantly however the results of this research indicate that whilst the Inner directed consumers are likely to feel encouraged by these new approaches, it is questionable whether Sustenance driven and Outer directed consumers will be encouraged by the prospect of having to acquire knowledge, take responsibility and empower themselves towards informed decision-making. Research findings identify that the Sustenance driven consumer may particularly lack the confidence required to adopt this strategy.

Whilst the Food Standards Agency appears to address the fact that food consumers levels of trust are at an all time low, there is little understanding of the fact that different strategies are needed for different consumer groups.

The findings of the research illustrate that Sustenance driven and Outer directed consumers have a willingness to trust (albeit different sources) and that they also want others to take responsibility. By trusting others to take responsibility they transform the risks concerned to social risks. They appear to be comfortable with this notion of 'guardianship', perhaps because it is familiar to them. In contrast Inner directed consumers appear to have much less willingness to trust institutions / bodies. Indeed they are sceptical of these. Importantly these observations tie risk perception into notions of responsibility, with Inner directed consumers taking responsibility and Sustenance driven and Outer directed being happy for others to be ultimately responsible.

Whilst this research did not investigate the issues of responsibility, qualitative responses highlighted that this is a key factor. The results of this research indicated that Inner directed consumers have a deeply rooted sense of responsibility, importantly this stretches far beyond themselves to a global level. They believe they are affected by global decisions not just those close to home, similarly they believe that their actions have global repercussions. This holistic view and wide sense of responsibility of the world is not shared by the Sustenance driven and Outer directed consumers.

Knowledge and understanding of consumer's perspectives on responsibility may prove vital, not only to marketers, but to all who aim to understand consumers food risk perceptions. Essentially one needs to design and develop strategies which will accommodate the scope of a target markets values and attitudes to include the values and attitudes of all whom that target market feel responsible.

6.6.1 SOCIAL VALUE GROUPS - WHO DO THEY 'TRUST'?

Qualitative responses demonstrated that risk perceptions and risk avoiding strategies are a direct indication of whom or what the consumer felt they could trust.

6.6.1.1 SUSTENANCE DRIVEN AND OUTER DIRECTED CONSUMERS

Sustenance driven consumers and Outer directed consumers both demonstrate a comfortable acceptance of traditionally approved systems / institutions / bodies. They demonstrate an acceptance of authority figures, society's institutions and those in positions of power. Importantly in addition Sustenance driven consumers demonstrate a deeply embedded reliance on and trust in their family. Outer directed consumers had a similar orientation and additionally appear to feel reassured by high prices and reputation. The results of the Locus of Control questionnaire conducted as part of the pilot demonstrating Sustenance driven and Outer directed consumer groups as externally directed.

6.6.1.2 INNER DIRECTED CONSUMERS

Inner directed consumers demonstrate a deeply rooted scepticism. It is important to recognise their beliefs as an illustration of healthy scepticism as opposed to distrust. Healthy because their scepticism drives them to acquire knowledge, take responsibility and be proactive.

The results demonstrate that Inner directed consumers strive to be informed and as so develop a knowledge and understanding of many issues. This knowledge and understanding facilitates informed decision making. Problems appear to arise when information is unattainable or there is some doubt as to the validity and reliability of the information. Inners believe that this is often the case. Interestingly Giddens (1991) suggested that in 'high modernity' the prime condition of requirements for trust is *not lack of power but lack of full information*, Inner directed responses support this.

6.7 RECOMMENDATIONS TO THE FOOD INDUSTRY IN LIGHT OF THE RESEARCH FINDINGS

The food industry, long recognised as dynamic and turbulent, is clearly presented with many challenges by the contemporary consumer. Whilst organisations embrace opportunities they also have to be fiercely competitive. Consumers in the food industry are bombarded with a plurality of choice and are consequently forced to make selections. A knowledge and understanding of consumer perceptions and behaviour is fundamental if one is to be successful.

Marketers in the Food Industry have long recognised the need for consumer segmentation, yet their reliance on demographic and socio economic segmentation is still common. The findings of this research clearly support recommendations for the use of segmentation by consumer social values. An understanding of consumer social values presents a base from which managers can develop strategies which take into account the complex and sophisticated nature of target markets in the food industry. This level of understanding, which is neither abstract nor static, allows for marketing environmental analyses which are effectively proactive.

It is crucial that marketers recognise the impact of social values on consumer decision making. The successful use of the screening instrument (part two of the questionnaire) demonstrated how, with relative ease, organisations can identify the social values of a group of consumers. Once an organisation is knowledgeable of the values held by a group of consumers, they can make assumptions with regard to food risk perceptions and risk avoiding strategies (this would need clarification and confirmation). Such knowledge and understanding can be used to design and develop products and marketing strategies which support and reinforce the values of their target market/s. Such a respect for consumer values is likely to be rewarded with loyalty, the key to long term success.

It is imperative that the food industry fully appreciates the significant challenges presented by all consumer groups, however one cannot avoid suggesting that an ignorance of the Inner directed group may prove to be most dangerous. Many

organisations will continue to target Sustenance driven and Outer directed groups of consumers. These present challenges in their own right, least of all in that they are predicted to remain static or decrease in numbers, thus presenting organisations with increasingly fierce, competitive and often saturated markets.

In contrast to the Sustenance driven and Outer directed groups, the numbers of Inner directed consumers are likely to increase. This trend is consistent with the social changes described by Giddens and Beck in conditions of high modernity. Organisations which target consumers with Inner directed social values will be presented with a complex group. The Inner directed consumers offer great opportunities for long term success, particularly in the areas of product innovation and the development of holistic, ethical practices throughout the whole of an organisation's management systems and operations.

The findings of this research and supporting work does, however, raise questions and highlight differences in the perceptions of this group. Many well grounded theories need to be addressed if we are to confidently understand and predict the consumer decision making processes of consumers with Inner directed values.

Traditional assumptions about consumer perceptions are in serious question, one example is the common belief that consumers make a strong positive link between price and quality. Fundamentally Inner directed consumers appear to largely disassociate the two variables. Outer directed consumers maintain a strong belief in this relationship, price is a very important indicator of quality to these consumers. Sustenance driven consumers indicate some scepticism, the food industry fuels their increasing belief that they are able to purchase products that are value for money (a strong motivator) by all too often bombarding them with cheaper alternatives too tempting to not try. Inner directed consumers, however, appear to reject any positive relationship between price and quality, this is significant to marketers particularly. Inner directed consumers adopt a very different means by which to measure quality, assessing functionality, processing, organoleptic qualities (perhaps the reason for an increasing need for consumers to taste before purchase), in addition they appear to evaluate a range of other components including country of origin and composition of packaging.

Other areas which the research findings indicate are in need for review include the food industry's environmental policies. Inner directed consumers indicate that these are insufficient. Research findings indicate that Inner directed consumers are globally aware and seriously concerned and proactive about ecological issues. Interestingly attempts by some organisations to develop 'green' products whether by the use of recyclable packaging or the offering a range of organic produce can in some instances be interpreted as superficial and purely a commercial strategy. Inner directed respondents presented a cynical attitude, believing that superficial attempts to 'save the environment' such as recycling bottles at a bottle bank were pointless when many people drive to the bottle bank. Inner directed responses clearly indicating that in their view there was a need for serious action.

Research findings suggest that effective risk avoiding strategies are a clear indication of who and what consumers trust. Knowledge of a target market's social values can be used strategically in the design, development and communication of products and services with an aim to increase consumer levels of trust.

The issue of responsibility is particularly interesting, evidence suggests that the scope of a consumer's risk perceptions and risk avoiding strategies may have much to do with the issue of responsibility.

Whilst it is valuable for the food industry to be knowledgeable of the food risks perceived by consumers there is a need to focus on designing and developing products and marketing strategies that avoid risks being perceived or certainly limiting consumers' perceived exposure to risk. Whilst there is a need for further research in this area, it is essential that marketing managers, food product developers and communicators integrate risk avoiding strategies into product and marketing planning. This strategy for prevention rather than cure may prove not only more efficient in the long term but proactive in designing products on a holistic level.

CHAPTER SEVEN

CONCLUSIONS

This study set out to investigate and clarify how social values influence consumer perceptions of food related risks and risk relieving strategies. In doing so the research aimed to contribute significantly to what is currently a limited body of empirical knowledge. The final chapter of this thesis will now consider the findings of the research in the context of the objectives outlined in Chapter One. It will then go on to outline recommendations for further research.

7.1 FINDINGS OF THE RESEARCH

While the previous chapter sought to explain the findings of the research in the context of the literature, the following will examine whether the objectives of the research established in Chapter One were achieved.

The first objective of the research was to critically evaluate existing theoretical and empirical research, in the area of perception of risk in general and more specifically food risk. An extensive examination of the literature revealed a great breadth of research both theoretical and empirical in the area of risk in general, some of this proved vital to the discussion and interpretation of the findings (Chapter Six). Whilst the literature review highlights several pieces of food risk research which observe and report on general food risk perceptions for a whole population, there is limited evidence of the use of value segmentation in such research despite the increasing recognition of the usefulness of this approach in marketing. No serious attempt has been made in the context of food risk perception research to apply the findings of Taylor Nelson's longitudinal survey of the Social Values of the British population, despite the successful application of these findings in a range of other areas.

The second objective of the research was to identify members of Taylor Nelson's three Social Value Groups 'Inner Directed', 'Sustenance Driven' and Outer Directed. A

screening instrument and consequent scoring system were designed which allowed for the successful categorisation of respondents. The instrument was confirmed as robust when used in practice on a large sample. Whilst successful in achieving the aim set, the screening instrument and scoring system could be further developed and refined to produce a less crude tool.

The third objective was to investigate consumer perceptions of food risk, for each of the three Social Value Groups. The research findings identified both the areas of perceived risk and the respondents' degree of concern. The research highlighted that a risk is perceived when individual values are threatened. The research findings reinforce the importance of understanding consumer values when aiming to understand risk perception.

Inner directed respondents perceived many food related risks. Sustenance driven respondents also highlighted a fairly wide range of perceived risks but fewer than the Inner Directed Group. Of these three groups Outer directed respondents identified the fewest.

Responses suggest that key areas of food risk perception for Inner directed consumers are 'technological', 'high cost' and 'environmental / ecological'. Sustenance driven consumers, perceive food risk in relation to 'high cost' and 'unfamiliarity'. The Outer directed consumer demonstrate a perception that 'cheap' is a risk for them.

Fundamentally the risks perceived were a reflection of the respondent's social values.

The fourth objective was to investigate how food risk perceptions may be reduced or relieved for each of the three Social Value Groups. This was initially approached, as other authors have done, by assuming that consumers perceive risks and then seek to relieve them. This approach is questioned. The findings of this research suggest that when a consumer is making a decision the choice that they opt for protects and reinforces their values. Choices which threaten their values (and constitute perceived risks) are avoided. This research therefore suggests that consumers adopt and develop strategies to avoid risks ever being perceived in the first place. When strategies are not adopted risks are likely to be perceived.

Inner directed respondents revealed that they have few risk avoiding strategies, 'comprehensive labelling' and 'independent agency approval' were the only strategies that Inner directed respondents identified as effective. Sustenance driven respondents identified, 'societal acceptance / endorsement' as a strategy for avoiding risk (this included advertising, someone similar to you, family and celebrity). They also indicated that a 'familiar brand' was an effective strategy. Outer directed respondents interestingly identified an extensive range of effective strategies to include, 'social acceptance / endorsement', 'branding', 'high cost' and 'authoritarian reassurance'.

The research findings suggest that what is perceived to be an effective strategy is closely related to whom the consumer trusts, and that in turn whom a consumer trusts influences the risk avoiding strategies they select as effective. Similarly the scope of risk perceptions and the risk avoiding strategies identified as effective appear to be influenced by a consumer's feelings of responsibility for others. These integral relationships were discussed previously in Chapter Six.

The final objective of the research was to make recommendations to the food industry, in light of the research findings. These recommendations can be found in the Discussion Chapter Six.

7.2 IMPLICATIONS AND RECOMMENDATIONS FOR FURTHER RESEARCH

Research of this type inevitably raises many issues which go beyond the objectives of those set. Whilst areas for further research are vast the aim of this section is to make key recommendations.

One of the most fundamental findings of this research was that the literature focuses primarily on what food risks are perceived. This appears to be at the expense of any real depth of research into risk relieving / avoiding strategies. Consequently there is a need for in-depth study of how a knowledge of risk avoiding strategies can be integrated into product development and marketing management strategies focusing on risk relief as a means of proactively avoiding risk rather than as a reaction to risk.

The second area recommended for further research arises as a consequence of the complex world view of the Inner directed group of consumers, and in response to the suggestion that this groups values will dominate in the future. Firstly the research findings highlighted that this group perceived many food risks and few risk avoiding strategies. Further research into these matters would be valuable.

Research findings suggested that a consumer's feelings of responsibility may fundamentally influence the scope of their risk perceptions and risk avoiding strategies. Outer directed respondents indicate that they felt responsible primarily for themselves, Sustenance driven respondents indicate that their feelings of responsibility stretched to the family, the Inner directed group suggested that their feelings of responsibility stretched much further to a more global level and believed that their actions ultimately have holistic consequences for 'the system'. These differences appear to influence the range of risks likely to be perceived by the groups concerned, and the risk avoiding strategies adopted. Whilst this research did give indications of what appears to be a fundamental relationship this issue was not within the scope of the aims of this research and needs further exploration.

The final recommendation for further research is to conduct a larger survey with the aim of identify may significantly higher number of Sustenance driven, Outer directed and Inner directed respondents to allow for a separate factor analysis for each in order to see more clearly how the identified patterns of risk perceptions and risk avoiding strategies for each group manifest themselves. In addition it would be interesting to confirm, via further work, that the relationships identified in the study extend to other groups in the population. It would, for example be interesting to examine a more mature group of consumers in light of Britain's ageing population and the marketing opportunities this group will increasingly offer. Similarly it would be interesting to investigate men, as they are becoming increasingly responsible for food shopping, and the young with their increasing purchasing power and media investment.

BIBLIOGRAPHY

Adam, B. Beck, U. and Van Loon, J. (2000). The Risk Society and Beyond Critical Issues for Social Theory. Sage Publications Ltd.

Arndt, J. (1967). Perceived Risk, Sociometric Integration, and Word of Mouth in the Adoption of a New Food Product. Risk Taking and Information Handling in Consumer Behaviour. Harvard University Press. pp. 289-393.

Anand, P. and Forshner, C. (1995). Of Mad Cows and Marmosets: From Rational Choice to Organisational Behaviour in Crisis Management. British Journal of Management. Vol. 6, pp. 221-233.

Annual Abstract of Statistics (1995) A publication of the Government Statistical Services p. 253

Arthur, C. and Wilkie, T. (1995). BSE: How did we get here, where do we go from here? **Independent on Sunday.** (10 December) p.10.

Atlas, J. (1984). Beyond Demographics. The Atlantic Monthly. (October) pp. 49-58.

Bailey, D.K. (1987). Methods of Social Research. The Free Press. London.

Bainbridge, H. (1999). Beyond Demographics. Wireless Review. (1 June) pp.1-3.

Barber, B. (1983). The Logical and Limits of Trust. Rutgers University Press Nerbrunswich: New Jersey.

Bauer, R. A. (1960). Consumer Behaviour as Risk Taking. In **Dynamic Marketing for** a **Changing World.** Proceedings of the 43rd National Conference of the American Marketing Association R.S.Hancock (eds.) (June 15, 16, 17) pp. 389-398.

Bauer, R. A. (1967). Consumer Behaviour as Risk Taking. Risk Taking and Information Handling in Consumer Behaviour. Graduate School of Business Administration Harvard University: Boston MA. pp. 23-33.

Baxter, M. (1990). Health and Lifestyles. Routledge.

Bebbington, A. (1983). Domiciliary Care for the Elderly, Research Design PSSRU Discussion Paper University of Kent cited in Walker R. 1985 Applied Qualitative Research. Gower, p.16.

Beck, U. (1992). Risk Society Towards a New Modernity. Sage Publications, London.

Beck, U. (1999). World Risk Society. Polity Press in association with Blackwell Publishers Ltd.

Belson, W.A. (1959). Matching and Prediction on the Principle of Biological Classification. Applied Statistics, viii pp. 65-75.

Belson, W. A (1985). Techniques of Market Research. A Practical Introduction to Survey Research. Research Centre: London.

Belson, W.A. (1981). The Design and Understanding of Survey Questions. Gower Pub Co Ltd., England.

Brown, R. (1993). Market Focus. Butterworth - Heineman.

Bowles, T. (1987). Does Classifying People by Lifestyle Really Help the Advertiser? **ADMAP.** pp. 36-40.

Bryman, A. (1993). Quantity and Quality in Social Research, Contemporary Social Research. Routledge: London.

Bulmer, M. (1983). Introduction and Further Reading in Sociological Research Methods. Eds. MacMillian, M: London.

Caulkin, S. (1987). Brand Wars. Marketing. (May 28) pp. 23-35.

Caplan, P. (2000). Risk Revisited. Pluto Press.

Churchill, G.A. (1987). Marketing Research: Methodology Foundations. The Dryden Press: London.

Collins, L. and Montgomery, C. (1969). The Origins of Motivational Research British.

Journal of Marketing. pp.103-113.

Conner, P. (1993) Understanding Determinants of Food Choice **British Food Journal** Vol. 95 (9) pp 27 - 31

Cunningham, S.M. (1964). Perceived Risk as a Factor in Product - Orientated Word of Mouth Behaviour: A First Step. Proceedings Educators Conference, American Marketing Society. pp. 229 - 238.

Cunningham, W.H. and Crissy, W.J.E. (1972). Market Segmentation by Motivation and Attitude. Journal of Marketing Research. Vol. 9, pp. 100 - 102.

Curley, S. (1992). Individual Differences in Risk Taking. in F. Yates eds. Risk Taking Behaviour. Edited Wiley and Sons Ltd. pp. 87-121.

Cvetkovich, G. and Lofstedt, R.E. (1999). Social Trust and the Management of Risk. Earthscan Publications Ltd.

Dade, P. (1988). Interesting Times - The New Consumer: Curse or Opportunity? British Food Journal. Vol. 90, No. 3, pp. 105 - 110.

Dake, K. (1991). Orienting Dispositions in the Perception of Risk. An Analysis of Contemporary Worldviews and Cultural Biases. **Journal of Cross - Cultural Psychology.** Vol. 22, No. 1, pp. 61 - 82.

Dalen, E. (1989). Trends in Customer Demand. Research into Values and Consumer Trends in Norway. **Tourism Management.** Vol. 10, No 3. pp. 183 - 186.

Daly, L. (1989). Irradiated Foods: Whose Choice? Whose Opportunity? British Food Journal. Vol. 91, No. 1, pp. 16 - 20.

Davis, G. (1999). Time, Food Shopping and Food Preparation: Some Attitudinal Linkages. British Food Journal. Vol. 99, No. 3, pp. 80 - 88.

Department of Health (1998). Communicating about Risk to Public Health Pointers to Good Practice.

de Vaus, D.A. (1986). Surveys in Social Research. George Allen and Unwin Publishers Ltd.

Dean, J.P, Eichorn, R.L. and Dean, L.R. (1967). Observation and Interviewing in An Introduction to Social Research. (Ed J.T. Dorby) Appleton Century-Crofts, New York.

Derbaix, C. (1983). Perceived Risk and Risk Relievers: An Empirical Investigation.

Journal of Economic Psychology. Vol. 3, pp. 19 - 38.

Denzin, N.K. (1970). Sociological methods: a sourcebook. Butterworth, London.

Dixon, B.R, Bouma, G.D. and Atkinson, G.B.J. (1987). A handbook of Social Science Research a comprehensive and practical guide. Oxford University Press.

Douglas, M (1992). Risk and Blame: essays in cultural theory. Routledge, London and New York.

Douglas, M. and Calvez, M. (1990). The Self As Risk Taker: A Cultural Theory Of Contagion In Relation To AIDS. Sociological Review. Vol. 38 pp. 445 - 464.

Douglas, M. and Wildavsky, A. (1982a). How Can We Know the Risks We Face? Why Risk Selection is a Social Process. Risk Analysis. Vol. 2, No. 2, pp. 49 - 52.

Douglas, M. and Wildavsky, A. (1982b). Risk and Culture. University of California Press.

Drucker, P.F. (1985). Innovation and Entrepreneurship. Heinemann, London p. 233.

Dolliver, M. (2000). Luxuriating in the Simple Pleasures of Life. Adweek. New York Vol. 41, Issue. 1, pp. 22 - 29.

Fife - Shaw (1995). Assessing Public Responses to Food Related Risks. British Association Annual (Sep.) pp. 9 - 15.

Filstaed, W.S. (1970). Qualitative Methodology. Markham, Chicago.

Feather, N. (1975). Values in Education and Society. NY Free Press.

Food Standards Agency (2000). Qualitative Research to Explore Public Attitudes to Food Safety. 30th May.

Food Link News. (2000). The Food Standards Agency Launch Approaches. No. 30.

Ford, N.J. and Rennie, D.M. (1987). Consumer Understanding of Food Irradiation Journal of Consumer Studies and Home Economics. Vol. 11, pp. 305 - 320.

Fowler, F.J. Jr. (1993). Survey Research Methods. 2nd Edition, Sage Publications.

Foxall, G.R. (1980). Consumer Behaviour A Practical Guide. Croom Helm, pp. 15, 16, 20, 21, 24.

Foxall, G.R. (1989). Marketing, Innovation and Customers. The Quarterly Review of Marketing. Vol. 1, No. 1, pp. 14 - 25.

Foxall, G.R. and Goldsmith, R.E. (1994). Consumer Psychology for Marketing. Routledge pp. 7, 50, 51, 57.

Frewer, L. (2000). Risk Perception and Risk Communication about Food Safety Issues. British Nutrition Foundation Nutrition Bulletin. Vol. 25, pp. 31 - 33.

Frewer, L.J., Shepherd, R. and Sparks, P. (1994). Biotechnology and Food Production: Knowledge and Perceived Risk. British Food Journal. Vol. 96, No. 9, pp. 26 - 32.

Frewer, L.J., Howard, C. and Shepherd, R. (1995). Consumer Perception of Food Risks Food Science and Technology. Today. Vol. 9, No. 4, pp. 212-216.

Frewer, L.J., Howard, C., Hedderley, D. and Shepherd, R. (1996). What Determines Trust in Information about Food Related Risks? Underlying Psychological Constructs. Risk Analysis. Vol. 16, No. 4, pp. 473-486.

Gerhardy, H., Hutchins, R.K. and Marshall, D.W. (1995). Socio Economic Criteria and Food Choice Across Meals. British Food Journal. Vol. 97, No. 10, pp. 24-28.

Giddens, A. (1990). The Consequences of Modernity. Polity Press in association with Blackwell Publishers Ltd.

Giddens, A. (1991). Modernity and Self – Identity: Self and Society in the Late Modern Age. Polity Press in association with Blackwell Publishers Ltd.

Giddens, A. (1976). New rules of sociological methods. Basic Books NY.

Goldsmith, R. (1983). Psychographics and New Product Adoption. An Exploratory Study Perceptual and Motor Skills Vol. 57. pp. 1071 - 6.

Goldsmith, R.E, Frieden, J. and Henderson, K.V. (1997). The Impact of Social Values on Food Related Attitudes. British Food Journal. Vol. 99, No. 9, pp. 352 - 357.

Gratton, C. and Taylor, P. (1991). Government and the Economics of Sport. Longman Harlow, pp. 43 - 49.

Greene, J. and D'Oliveira, M. (1995). Learning to use Statistical Tests in Psychology. Open University Press, Bristol.

Guba, E.G. and Lincoln, Y.S. (1982). Epistemological and Methodological Bases of Naturalistic Inquiry. Educational Communication and Technology Journal. Vol. 30, No. 4, pp. 233-252.

Gunter, B. and Furnham, A. (1992). Consumer Profiles An Introduction to Psychographics. Routledge, London and New York.

Hair, J.F (Jr), Anderson, R.E, Tatham, R.L and Black, W.C, (1995). Multivariate Data Analysis. Prentice Hall, New Jersey.

Hakim, C. (1987). Research Design Strategies and Choices in the Design of Social Research. Routledge, London, p. 27.

Harrell, G. (1986). Consumer Behaviour. Harcourt Brace Joranovich Place p. 66.

Hatton, A. (2000). The Challenge and Changing Technique for Effective Segmentation. The Chartered Institute of Marketing. Vol. 10, No. 2, p. 2.

Henry, W.A. (1976). Cultural Values Do Correlate with Consumer Categorisation Process. Journal of Marketing. Vol. 46, pp. 60-72.

Hirschi, T. and Selvin, H.C. (1967). Delinquency Research: An Appraisal of Analytic Methods. Free Press, New York, Collier – MacMillian London.

Hisrich, R.D., Dornoff, R.J. and Kernan, J.B. (1972). Perceived Risk in Store Selection.

Journal of Market Research. Vol. 9, pp. 435 - 439.

Hohenemser, D. (1983). The Nature of Technological Hazard. Science. Vol. 220, pp. 378 - 384.

Hoinville, G., Jowell, R. and Associates (1983). Survey Research Practice. Heinemann Books Ltd, p. 9.

Hoorens, V. and Bunk, B.P. (1993). Social Comparison of Health Risks: Locus of Control, the Person-Positively, and Unrealistic Optimism. Journal of Applied Social Psychology. Vol. 23, No. 4, pp. 291 - 302.

Howell, D.C. (1997). Statistical Methods for Psychology. Belmont, CA Duxbury.

Hughes, J.A. (1976). Sociological analysis: methods of discovery. Nelson London.

Kahle, L.R. (1983). Social Values and Social Change: Adaptation to life in America. Praegar, New York.

Kahle, L.R, Rose, G. and Shoham, A. (1999). Findings of LOV Throughout the World, and Other Evidence of Cross-National Consumer Psychographics: Introduction.

Journal of Euro - Marketing. New York, Vol. 8, No. 1/2, pp. 1-13.

Kahle, L.R, Beatty, S.E. and Homer, E. (1986). Alternative Measurement Approaches to Consumer Values The List of Values (LOV) and Values of Lifestyle (VALS). Journal of Consumer Research. Vol. 13, No. 4, pp. 5 - 9.

Kaplan, L.B, Szybillo, G.J. and Jacoby, J. (1974). Components of Perceived Risk in Product Purchase: A Cross-Validation. Journal of Applied Psychology. Vol. 59, No. 3, pp. 287 - 291.

Kates, R.W. and Kasperson, J.X. (1983). Comparative Risk Analysis of Technological Hazards (a review). **Proceedings of the National Academy of Sciences.** USA Vol. 80, pp. 7027 - 7038.

Labowitz, S. and Hagedorn, R. (1971). Introduction to Social Research. McGraw-Hill, New York.

Langer, E.J. (1975). The Illusion of Control. Journal of Personality and Social Psychology. Vol. 32, No. 2, pp. 311 - 328.

Laurence, M. (1989). Chris MacNulty's Crystal Ball. Industrial Marketing Digest. pp. 123-128.

Leach, C. (1987). How Conventional Demographics Distort Marketing Realities. **ADMAP.** (May) pp. 41 - 45.

Lowrance, W.W. (1976). Of Acceptable Risk Science and the Determination of Safety. William Kaufmann, Inc. Los Altos California.

Lupton, D. (1999). Risk. Routledge.

Lunman, N. (1980). Trust and Power. NY John Wiley.

MAFF Food Safety Information Bulletin. (June 1997) pp.15-17.

MacNulty, K.W. (1985). UK Social Change Through A Wide - Angled Lens. Futures. pp. 331 - 347. (Taylor Nelson Ltd).

Maslow, A.H. (1970). A Theory of Human Motivation. Harper and Row (2nd Edition).

Martindale, D. (1974). Sociological theory and the problem of values. Charles E Merrill Columbus Cited in Walker.R. (1985). Applied Qualitative Research. Gower, England.

Mayntz, R., Holm, K., Hueber, R. (1976). Introduction to empirical sociology. Penguin, Harmondsworth.

McNeil, P. (1990). Research Methods. Routledge, London.

McKenna, F.P. (1993). It won't happen to me: Unrealistic Optimism or Illusion of Control? British Journal of Psychology. Vol. 84, pp. 39 - 50.

Mennell, S, Murcott, A., van Otterloo, A.H. (1992) The Sociology of Food Eating, Diet and Culture Sage Publications Ltd

Miles, L. (1990). Gateway to Europe, Fortune Tellers. Marketing. p. 35.

Mitchell, A. (1983). The Nine American Lifestyles. Macmillan Publishing Inc.

Mitchell, V-W. (1999). Consumer Perceived Risk: Conceptualisation and Models. European Journal of Marketing. Vol. 33, No. 1/2, pp. 163-195.

Mitchell, V-W. and Boustani, P (1992). Consumer Risk Perceptions in the Breakfast Cereal Market. British Food Journal. Vol. 94, No. 4, pp. 17 – 26.

Mitchell, V-W. (1993). The Effects of Demographic Variables on Measuring Perceived Risk. Academy of Marketing Science Conference Developments in Marketing Sciences XVI. pp. 663-9.

Mitchell, V-W. (1998). A role for consumer risk perception in grocery retailing. **British** Food Journal. Vol. 100, No. 4, pp. 1-13.

Mitchell, V-W. and McGoldrick, P.J. (1996). Consumers risk – reduction strategies: a review and synthesis. The International Review of Retail, Distribution and Consumer Research. Vol. 6, No. 1, pp. 1 - 33.

Mitchell, V-W. and Gretorex, M. (1990). Consumer Perceived Risk in the UK Food Market. British Food Journal. Vol. 92, No. 2, pp. 16-22.

Mitchell, V-W. and Gretorex, M. (1990). Measuring Perceived Risk and It's Components Across Product Categories. Proceedings 19th European Marketing Academy Conference Innsbruck.

Moser, C.A and Kalton, G. (1979). Survey Methods in Social Investigation. Heinemann Educational Books Ltd., England.

Moser, C.A. and Kalton, G. (1985). Survey Methods in Social Investigation (2nd edition). Gower Publishing Company Ltd., England.

Munson, M.J. (1979). Developing Practical Procedures for the Measurement of Personal Values in Cross - Cultural Marketing. **Journal of Marketing Research.** Vol. 15, pp. 48-52.

Murcott.A. (1983). The Sociology of Food and Eating. Gower Publishing Company Limited

Murctt.A. (1998). The Nations Diet. The Social Science of Food Choice. Addison Wesley Longman Limited

Mutsaers, B.R.H. and Shepherd, R. (1994). Food Poisoning and Other Food Hazards: Risk Perceptions and Implications for Risk Communication. **Proceedings of the XIV**Conference of Home Economics and Consumer Studies. pp. 356 - 363.

Nachmias, C. and Nachmias, D. (1981). Research Methods in the Social Sciences. Edward Arnold.

Nelson, E. (1986). New Product Development: The Role Of Social Change Analysis. The Food Consumer. Edited by C. Ritson, L. Gofton and McKenzie, J. John Wiley and Sons Ltd., pp.197 - 207.

Newman, J.W. (1957). Motivational Research and Marketing Management. Harvard University Press.

Nunnally, J.C. (1967). Psychometric Theory. McGraw - Hill, New Delhi.

O'Riodan, T. (1982). Risk Perception Studies and Policy Priorities. Risk Analysis. Vol. 2, No. 2, pp. 95-100.

O'Riodan, T. (1983). The Cognitive and Political Dimensions of Risk Analysis. Journal of Environmental Analysis. Vol. 3, pp. 345 - 354.

Oppenheim, A.N. (1992). Questionnaire Design Interviewing and Attitude Measurement. Punter Publisher, London.

Perry, M. (1990). Application of the List of Values Alternative Psychographic Assessment Scale. Psychological Reports. Vol. 66, pp. 403-406.

Perry, M and Hamm, B.C. (1969). Canonical Analysis of Relations Between Socio-economic Risk and Personal Influence in Purchase Decisions. **Journal of Market** Research. Vol. 11, pp. 351-354.

Peters, T.J. (1989). Thriving on Chaos Handbook for a Management Revolution.

Pan Books Ltd.

Powderly and MacNulty (1990). A Turbulent Time Ahead. Consumer Trends Marketing. (Oct 11), pp. 33 - 34.

R & D Sub Committee (1979). Qualitative research - a summary of the concepts involved. Journal of Market Research Society. Vol. 21, No. 2, p.110.

Riesman, D. (1950). **The Lonely Crowd.** Yale University Press, New Haven and London.

Rotter, J.B. (1966). Generalised expectations for Internal versus external control of reinforcement. **Psychological Monographs: General and Applied.** Vol. 80, No. 1, pp. 1 - 28.

Rotter, J.B. (1971). External Control and Internal Control. **Psychology Today.** (June) pp. 37 - 59.

Rokeach, M. (1960). The Open and the Closed Mind. Basic Books, New York cited in Nachmias, C. and D, Nachmias (1981). Research Methods in the Social Sciences. Edward Arnold.

Rokeach, M. (1973). The Nature of Human Values. New York: Free Press.

Rose, M.G., Shoham, A., Kahle, L.R and Batra, R. (1994). Social Values, Conformity and Dress. Journal of Applied Social Psychology. Vol. 24, No. 17, pp. 1501 - 1519.

Roselius, T. (1971). Consumer Rankings of Risk Reduction Methods. **Journal of Marketing.** Vol. 35, pp. 56 - 61.

Runciman, D. (1996). Do we really want to know the odds? **The Guardian.** (Thursday, October 3), p. 8.

Scott, J.E. and Lamont, L.H. (1970). Relating Consumer Values to Consumer Behaviour: A Model and Method for Investigation. In Increasing Marketing

Productivity. Editor Greer, T.V. Chicago American Marketing Association. pp. 283-288.

Shipman, M. (1981). The Limitations of Social Research. London.

Short, J.F. (1984). The Social Fabric at Risk: Toward The Social Transformation of Risk Analysis. American Sociological Review. Vol. 49, pp. 711 - 725.

Skelly, F.B. and Nelson, E. (1966). Market Segmentation and New Product Development. Scientific Business. Vol. 4, pp.13-22.

Slovic, P., Fischhoff, B. and Lichtenstein, S. (1979). Rating the Risks. Environment. Vol. 21, No. 3, pp. 14 - 39.

Slovic, P., Fischhoff, B. and Lichtenstein, S. (1982). Why Study Risk Perception. Risk Analysis. Vol. 2, No. 2, pp. 83 - 93.

Slovic, P. (1987). Perception of Risk. Science. Vol. 236, pp. 280-285.

Slovic, P. (1993). Perceived Risk, Trust and Democracy. Risk Analysis. Vol. 13, No. 6, pp. 675 - 682.

Soby, B.A, Simpson, A.C.D. and Ives, D.P. (1994). Managing Food Related Risks: Integrating Public and Scientific Judgements. Food Control. Vol. 5, No. 1, pp. 9 - 19.

Sparks, P., Shepherd, R. and Frewer, L.J. (1992). The Environmental, The Economy and The Ethical: Assessing Dimensions of Public Attitudes Toward the Application of Gene Technology to Food Production. Poster Presented at the 6th European Health Psychology Society Conference Leipzig, 25 - 28 August.

Tame, J. (1993). Life Planning for Executives. Long Range Planning. Vol. 26, No. 5, pp. 93 – 102.

Taylor, P.B. and Rao, C.P. (1982). An assessment of the interaction effects of brand and store reputation on consumer received risk and confidence. Akron Business and Economic Review. Vol. 13, No. 2, pp. 43 - 47.

Thurstone, L.L. (1928). Attitudes can be Measured. The American Journal of Sociology. Vol. 33, No. 4, (Jan).

Toh, R. and Heeren, S.G. (1982). Perceived Risks of Generic Grocery Products and Risk Reduction Strategies of Consumers. **Akron Business and Economic Review.** Vol. 13, pp. 43 – 48.

Veal, A.J. (1997). Research Methods for Leisure and Tourism. Longman ILAM UK.

Velzhoffer, O. and Ascheberg, C. (1999). Transnational Consumer Cultures and Social Milieus. Journal of the Market Research Society. Vol. 41, No. 1, pp. 47-57.

Vinson, D.E, Munson, J.M and Nakaniski, M. (1977a). An Investigation of the Rokeach Value Survey for Consumer Research Applications. In Advances in Consumer Research IV. Ed. W.D.Perrault, Provo Utah Association for Consumer Research.

Vinson, D.E, Scott, J.E. and Lamiont, L.M. (1977b). The Role of Personal Values in Marketing and Consumer Behaviour. **Journal of Marketing.** Vol. 41, pp. 44-50.

Walker, R. (1985). Applied Qualitative Research. Gower, England.

Weinstein, N.D., Klotz, M.L. and Sandman, P.M. (1988). Optimistic Biases in Public Perceptions of the Risk from Radon. American Journal of Public Health. Vol. 78, No. 7, pp. 796 - 800.

Weinstein, N.D. (1983). Reducing Unrealistic Optimism About Illness Susceptibility. **Health Psychology.** Vol. 2, No. 1, pp. 11 - 20.

Weinstein, N.D. (1984). Why It Won't Happen To Me: Perceptions of Risk Factors and Susceptibility. Health Psychology. Vol. 3, No. 5, pp. 431 - 457.

Weinstein, N.D. (1987). Unrealistic Optimism About Susceptibility to Health Problems: Conclusions from a Community - Wide Sample. **Journal of Behavioural Medicine.**Vol. 10, No. 4, pp. 41 - 60, No. 5, pp 481 - 500.

Wildavsky, A. (1979). No Risk is the Highest Risk of All. American Scientist. Vol. 67, pp. 32 - 37.

Wildavsky, A. and Dake, K. (1990). Theories of Risk Perception: Who Fears What and Why? **Daedalus.** Vol. 119, No. 4, pp. 41-60.

Wildavsky, A. and Dake, K. (1991). Individual Differences in Risk Perception and Risk.

- Taking Preferences. Cited in The Analysis, Communication and Perception of
Risk. Edited by B.J.Garrick and W.C.Gekler, Plenum Press, New York.

Wind, Y. (1978). Issues and advances in segmentation research. Journal of Marketing Research. Vol. 15, pp.153-165.

Wright, L. and Len Tiu. (2001) Food Taste Preferences and Cultural Influences on Consumption British Food Journal Vol 103 (5) pp 348 - 357.

Wonnacott, T.H and Wonnocott, R.J (1969). Introductory Statistics. John Wiley and Sons, Inc., New York.

Yankelovich, D. (1964). New Criteria for Market Segmentation. Harvard Business Review. Vol. 42, pp. 83 - 90.

Yeung, R.M.W and Morris, J. (2001). Food Safety Risk: Consumer Perception and Purchase Behaviour. **British Food Journal.** Vol. 103, No. 3, pp. 170-186.

APPENDICES

APPENDIX 9.

Quotes

APPENDIX 1.	Consumer Perceptions of Food Risk Questionnaire
	and Variable Codes
APPENDIX 2.	Correlation Matrix
	(Dominant Sample of 54)
APPENDIX 3.	Correlation Matrix
	(Total Homogeneous Sample of 215)
APPENDIX 4.	Interview One: Interview Guide / Schedule
APPENDIX 5.	Social Value Score Profiles for Total Homogeneous Sample of 215
APPENDIX 6.	Tables to Illustrate the Relationships between Social Value Variables and Food Risk Perception Variable for the Total Homogeneous Sample of 215
APPENDIX 7.	Tables to Illustrate the Relationships between Social Value Variables and Risk Relieving Strategy Variables for the Total Homogeneous Sample of 215
APPENDIX 8.	Sample for Qualitative Interviews

Qualitative Interview Transcripts: Selection of

APPENDIX 1

CONSUMER PERCEPTIONS OF FOOD RISKS QUESTIONNAIRE AND VARIABLE CODES

1.1 MAIN STUDY QUESTIONNAIRE

Part One. Homogeneous Group Questionnaire

Part Two. The Screening Instrument

Part Three. Food Risk Perception Questionnaire

Part Four. Risk Relieving Strategy Questionnaire

1.2 MAIN STUDY QUESTIONNAIRE VARIABLE CODES

Part Two. The Screening Instrument

Part Three. Food Risk Perception Questionnaire

Part Four. Risk Relieving Strategy Questionnaire

APPENDIX 1.1

MAIN STUDY QUESTIONNAIRE

Part One. Homogeneous Group Questionnaire

Part Two. The Screening Instrument

Part Three. Food Risk Perception Questionnaire

Part Four. Risk Relieving Strategy Questionnaire



Sheffield Hallam University

Questionnaire: Consumer Perceptions of Food Risks

Over 65

Do you work full time ?-	Yes	
	<u>N</u>	
Is your total household income <u>more than</u> £12,000 (pa)?	Yes	
	o N	
Do you have children 11 years old or younger living at home?	Yes	
	No	
Do you have children 12 to 16 years old (inclusive) living at home?	Yes	
	No No	
Would you be willing to participate further in this research at a later date?	Yes	
	No	

Please read the following statements carefully and respond as truthfully as possible, by ticking the appropriate box. the statement, or really don't know what your view is regarding the statement. All statements that show a * have a Please respond to all the statements, only answering Neither Agree or Disagree if you either don't understand further explanation/definition on page 5. Thank-You.

		Agree Agree Neither Dis a lot Agree or Disagree	Neither Disagree Disagree Agree or a lot Disagree	
1)	I am primarily motivated by a need to feel secure*.			
2)	I am primarily motivated by a need for self discovery*.			
3)	I am motivated by the desire to have the best.			
4	I am not keen to seek change.			
2)	I am largely unconcerned about the opinion of me held by the world at large.			
(9	I would describe myself as fairly unconventional*.			
2	Winning is a priority for me.			
8)	I tend to be set in my ways.			
6	I believe that the breakdown of tradition causes instability in society.			
10	10) I am strongly motivated to get ahead of others			

- 8) I tend to be set in my ways.
- 9) I believe that the breakdown of tradition causes
- 10) I am strongly motivated to get ahead of others.

	nsagree
11) I voice my opinions even if it means disagreeing with others or being unpopular.	
12) I tend to hold onto my existing lifestyle.	
13) I'm motivated by a desire to improve my status*.	
14) I work most effectively where there are few rules and little structure.	
15) I take pride in keeping up appearances.	
16) I believe groups with alternative values and lifestyles are a positive force in society * .	
17) I tend to make sure that other people know all about my successes.	
18) I invest relatively little time in making new contacts/friends.	

Neither Disagree Disagree

Agree

Agree a lot

Agree or

IMPORTANT:

Now you have completed this Part of the questionnaire, look back at your 18 responses and :-

- 1. Circle the three answers that you agree with the most.
- 2. Circle the three answers that you disagree with the most. Thank-You

Further explanations and definitions

1) Feel secure:

To feel safe, steady and reassured.

2) Self discovery:

To understand and realize yourself.

6) Unconventional:

Not conforming to societies established and accepted rules, attitudes, values and standards. To be individualistic, non-conformist.

13) Status:

Your social or professional position. Increased responsibility, privilege and esteem are attached to a higher level of status.

16) Alternative Lifestyle:

A lifestyle regarded by its members as preferable to that led by contemporary society because it is less conventional, materialistic, institutionalized and often regarded as more in harmony with nature.

arts

g risks.
od purchasin
\mathbf{f}_{0}
and f
risks
food
following
the f
about
1 worry
yor
w much
W.C

		Does not Worries me	ē
		worry me a great deal	al
		at all	
5	1) Genetically engineered foods.	1 2 3 4 5 6 7 8 9	
2)	Buying an expensive food brand.		
3)	Paying more for expensively and attractively packaged food.		
4	BSE (sometimes described as mad cow's disease).		
2)	Paying more for pre prepared vegetables e.g. Pre - packed, washed raw cauliflower florets.		
(9	Pesticides, fertilizers, veterinary drug residues in foods.		
12	Using a cheap food supermarket.		
8)	High salt diet.		
6	9) Artificial food preservatives.		
10)	10) High fat diet.		

11) Information on packaging that may be misleading.

17) The image of a food supermarket.

18) Artificial food coloring.

16) Using artificial sweeteners.

20) Limited information on food packaging.

(e.g. Heinz).

21) Artificial food flavorings.

22) Eating out as opposed to eating in.

23) Buying a cheap food brand.

24) Low fibre diet.

12) Food packaging which is not recyclable.

a great deal Worries me

worry те Does not at all

8	
7	
9	
5	
4	
3	
2	
1	

တ

-		

-			
-			
_			
	ı	L	لسسا

26) Using food grown or reared using non organic methods.

25) Trying a new or different supermarket to the one you usually use.

27) Using unfamiliar ingredients in food preparation.

28) Foods grown or/and produced by a company or country that operate controversially.

29) Using an expensive food supermarket.

If you are worried about a food risk or a food purchasing risk, to what extent would the following help to reduce your worry.

		Would not reduce Would	
		or relieve my worry	#
		at all my worry	
		1 2 3 4 5 6 7 8 9	
1	The food has been advertised a great deal.		
7)	A person similar to yourself, is not worried about the food risk.		•
3	Try a free sample of the product on a trial basis before buying.		
4	Buy the brand that I have always bought and am familiar with.		
5)	The food / ingredient / process has been approved by a government body.		
6	An expert on the food risk is not worried about it.		
2	Buy the brand (e.g. Heinz) that has a good reputation.		
8)	A family member is not worried about the food risk.		
6	Shop in the cheaper supermarket.		

10) Shop in the most expensive supermarket.

eliminate all mv worry Would or relieve my worry Would not reduce at all.

my wo		
=		
dii.		
ק ק		

9			
ဆ			
7			
6		•	
5			
4			
က			
7			
			

	-	

	L

13) A money back guarantee.

12) A respected celebrity is not worried about the food risk.

11) The food has clear and comprehensive labelling.

- - 15) Buy the cheaper brand, product.

16) Buy the most expensive brand, product.

17) The food / ingredient / process has been approved by an independent/private testing company e.g. Which? magazine.

In light of the recent BSE scare, have you done any of the following.

(Please tick yes or no for <u>all</u> the responses)

		Yes	No
2	None of the following are applicable to me because I never eat beef		
2)	Substituted all beef with another meat		
3)	Substituted all beef with vegetables		
4	Substituted some beef with another meat		
53	Substituted some beef with vegetables		
6)	Continued normal consumption levels of beef		
7)	Increased consumption level of beef when prices were reduced		
3)	Avoided cheaper beef products such as burgers, pies, sausages and continued to buy more expensive types of beef.		

APPENDIX 1.2

MAIN STUDY QUESTIONNAIRE VARIABLE CODES

Part Two. The Screening Instrument

Part Three. Food Risk Perception Questionnaire

Part Four. Risk Relieving Strategy Questionnaire

Part Two. Screening Instrument Variables and Codes

Ref	Variable	Code
1.	I am primarily motivated by a need to feel secure	Secure
2.	I am primarily motivated by a need for self discovery	Discovery
3.	I am motivated by the desire to have the best	Best
4.	I am not keen to seek change	Nochange
5.	I am largely unconcerned about the opinion of me held by the world at large	World
6.	I would describe myself as fairly unconventional	Unconven
7.	Winning is a priority for me	Winning
8.	I tend to be set in my ways	Setways
9.	I believe that the breakdown of tradition causes instability in society	Trad
10.	I am strongly motivated to get ahead of others	Ahead
11.	I voice my opinions even is it means disagreeing with others or being unpopular	Voiceops
12.	I tend to hold onto my existing lifestyle	Holdlife
13.	I'm motivated by a desire to improve my status	Status
14.	I work most effectively where there are few rules and little structure	Fewrules
15.	I take pride in keeping up appearances	Appear
16.	I believe groups with alternative values and lifestyles are a positive force in society	Alternat
17.	I tend to make sure that other people know all about my successes	Mysuccess
18.	I invest relatively little time in making new contacts / friends	Contacts

Part Three. Food Risk Perception Questionnaire Variables and Codes

Ref	Variable	Code
1.	Genetically engineered foods	geneng
2.	Buying an expensive food brand	expbrand
3.	Paying more for expensively and attractively packaged food	exppack
4.	BSE (sometimes described as mad cow's disease)	BSE
5.	Paying more for pre prepared vegetables e.g. Pre - packed, washed raw cauliflower florets	prepared
6.	Pesticides, fertilisers, veterinary drug residues in foods	pestfert
7.	Using a cheap food supermarket	cheapsup
8.	High salt diet	highsalt
9.	Artificial food preservatives	artipres
10.	High fat diet	highfat
11.	Information on packaging that may be misleading	informis
12.	Food packaging which is not recyclable	notrepac
13.	Having to buy a different or new brand because your usual brand is out of stock	difbrand
14.	Throwing away food that has past it's 'best before date'	throwbbd
16.	Using artificial sweeteners	artsweet
17.	The image of the food supermarket	imagesup
18.	Artificial food coloring	artcolor
19.	Buying a store's own label product (e.g. Tesco) as opposed to a manufacturers brand (e.g. Heinz)	ownlabel
20.	Limited information on food packaging	liminfo
21.	Artificial food flavourings	artflav
22.	Eating out as opposed to eating in	eatout
23.	Buying a cheap food brand	cheapbra
24.	Low fibre diet	lowfibre
25.	Trying a new or different supermarket to the one you usually use	difsup
26.	Using food grown or reared using non organic methods	nonorg
27.	Using unfamiliar ingredients in food preparation	unfaming
28.	Foods grown or / and produced by a company or country hat operate controversially	controv
29.	Using an expensive food supermarket	expsup

Part Four. Risk Relieving Strategy Questionnaire Variables and Codes

Ref	Variable	Code
1.	The food has been advertised a great deal	advert
2.	A person similar to yourself, is not worried about the food risk	similyou
3.	Try a free sample of the product on a trial basis before buying	freesamp
4.	Buy the brand that I have always bought and am familiar with	fambrand
5.	The food / ingredient / process has been approved by a Government body	govappro
6.		ovmost
	An expert on the food risk is not worried about it	expert
7.	Buy the brand (e.g. Heinz) that has a good reputation	brandrep
8.	A family member is not worried about the food risk	family
9.	Shop in the cheaper supermarket	cheapers
10.	Shop in the most expensive supermarket	expsuper
11.	The food has clear and comprehensive labelling	comlabel
12.	A respected celebrity is not worried about the food risk	celeb
13.	A money back guarantee	moneybac
14.	Shop in the supermarket that you believe has the best image	supimage
15.	Buy the cheaper brand / product	cheaperb
16.	Buy the most expensive brand / product	expbra
17.	The food / ingredient / process has been approved by an independent / private testing company e.g. Which ? magazine	indeapro

APPENDIX 2

CORRELATION MATRIX

(Results of Dominant Sample of 54, (18 Sustenance driven, 18 Outer directed and 18 Inner directed) (Using Pearson Correlation))

- 2.1 Correlation between the Social Value Variables within the Screening Instrument (Part Two of the Questionnaire)
- 2.2 Correlation between the Food Risk Perception Variables within Part Three of the Questionnaire
- 2.3 Correlation between the Risk Relieving Strategy Variables within Part Four of the Questionnaire
- 2.4 Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Food Risk Perception Variables(Part Three of the Questionnaire)
- 2.5 Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Risk Relieving Strategy Variables (Part Four of the Questionnaire)

Coefficients / significance levels are only detailed when <0.05.

Correlation between the Social Value Variables within the Screening Instrument (Part Two of the Questionnaire)

																						·														
																				-																
		contacts	4		•	-0.2867	0.0360	0.4106	0.0020	+				-0.2890	0.0340	0.3150	0.0200	+		-0.3091	0.0230	-0.5239	0.0000	+									-0.4335	0.0010		
	,	mysucces contacts										-0.3351	0.0130	0.3532	0.0000			0.3901	0.0040	0.4171	0.0020	1			2222	0.0170			0.2930	0.0320					-0.4335	0.0010
		alternat I	, '	0.4653	0.0000			-0.3748	0.0050			0.4303	0.0010	-0.3195	0.0190	-0.4448	0.0010	-0.3574	0.0080	-0.3947	0.0030	0.3709	0900'0		200.0	0.0290	0.4485	0.0010	-0.4306	0.0010						
		appear a	•	\dagger		0.3842	0.0040			+ 8905.0	0.000	-0.5944	0.0000	0.4692	0.0000	0.3528	0.0000	0.5800	0.0000	0.5791	0.0000			•	00000	0.0030	-0.4618	0.0000			-0.4306	0.0010	-0.2930 -	0.0320	_ '	1
		fewrules a	-0.2930 +	0.0320	0.0170			-0.3351 +	0.0130	0.4118	0.0020	0.5178	0.0000	-0.2828	0.0380	-0.4939	0.0000	-0.3259	0.0160			•		-0.3370	0.0130				-0.4618	0.0000	0.4485	0.0010			<u>.</u>	
		status fe		192.0	0.0420	-				-0.4309	0.0010	-0.3450	0.0110	0.6397	0.0000				0.6828	0.6828	0.0000	.+.			-			_	0.3909	0.0030	-0.2976	0.0290	0.3222	0.0170		
		I - I	+	+		+		0.5682 -	0.0000							0.4375 -	0.0010 +	+				+	-				-0.3370 -	0.0130		-					_•	-
		voiceops holdlife	+	85670	0.0010	+		-0.4479	0.0010		•	0.4854	0.0000	0.2838 -	0.0380	-0.5183	0.0000	-0.3276 +	0.0160			1				1					0.3709	0900.0	•		-0.5239 +	0.0000
				+	-	+				+ 9886.0-	0.0040	-0.3508	0.0000	0.8136	0.0000	•		0.2975	0.0290	+				•	0,000	0.0000	+		0.5791 -	0.0000	-0.3947	0.0030	0.4171	0.0020		0.0230
		ahead	1			+		0.4640 -	0.000.0	-0.2886	0.0340	-0.5632	0.0000		_	0.3690 -	0.0060			0.2975	0.0290	-0.3276 +	0.0160	1			-0.3259 -	0.0160	0.5800	0.0000	-0.3574	0.0030	0.3901	0.0040		4
•		ays trad	0.4059 +	0.0020	0.0080	+		0.7932	0.0000	-0.3535	0.0000	-0.5204 -(0.0000	+			_	0.3690	09000	_	_	-0.5183 -(0.0000	0.4375 +	0.0010	<u> </u>	-0.4939	0.0000	0.3528	0.0000	-0.4448	0.0010		_	0.3150 +	0.0200
		ing setways	0	5 6		<u> .</u>		0.2981 0	0.0290	o o	0	9	0				_	_	°	0.8136	0,0000	0.2838 -0	0.0380	-	0 6207	0.0000	-0.2828 -0	0.0380	0.4692 0	0.000.0	-0.3195 -0	0.0190	0.3532	0.0000		0.0340
•		unconver winning	-0.4952 -	0.0000	0.0050	-0.2795 +	0.0410	-0.4324 -0.	0.0010 0.	0.4569	0.0010				_	-0.5204 -	0.000	-0.5632 +	0.0000	-0.3508 0.	0.0090	0.4854 0.	0.0000	<u>.</u>	0.3450		0.5178 -0.	0.0000	-0.5944 0.	0.0000	0.4303 -0.	0.0010	-0.3351 0	0.0130	9	
		П					0.0040 0.0	70-	0.0	70).	0.4569	0.0010	•		-0.3535 -0.5	0.0000	O.	Ö	-0.3886 -0.3	0.0040 0.0	0.	Ö	!	0.4300		0.4118 0.2	0.0020	-0.5068 -0.	0.0000	Ö	ō	Ģ.	Ö		-
		nochange world		+		-0.3823	0.0							- 180	067				000	-0.3	0.0	+ 621	010	- 285		0.0			-0.5	0.0	748 +	050			0.4106 +	0.0020
		nochar	0.3678	37.0	<u> </u>	ļ.				23	40	95 -0.4324	10 0.0010	92 -0.2981	00 0.0290	0.7932	0.0000	0.4640	0.0000	- 16	8	-0.4479	0.0010	0.5682		. 8	-0.3351	0.0130	42 +	40	-0.3748	0.0050	,			
		best	+					-		-0.3823	0.0040	7 -0.2795	0.0410	0.5792	0.0000		0	+		0.6391	0.0000	+ 8	0	+	1 0 6061		-	0	0.3842	0.0040	-	00	+		0.2867	0.0360
		discover						- 3		+		2 0.3797	0.0050			9 -0.3558	0.0080					0.4258	0.0010	• -	0.2761	0.0420	0 0.3231	0.0170	,		0.4653	0.0000		_		
		secure				+		3 0.3678	0.0060	-0.4321	0.0010	-0.4952	0.0000			0.4059	0.0020	+						+	-		-0.2930	0.0320	+				+		+	
			secure	discourse		best		nochange		world		unconver		winning		setways		trad		ahead		voiceops		holdlife	ctature	c r	fewrules		арреаг		alternat		mysucces		contacts	

Correlation between the Food Risk Perception Variables within Part Three of the Questionnaire

	peneng ext	brantbxppack B	Г	prepared pestlert	rt Icheapsu	mehsalt	artipries		ifomis Inc	trepaddil	brandthre	throwb Hartsweel imagesu	eel image	esu artcolor		ownlabe liminfo	artilav	eatont Ic	leanbrillo.	cheanbrillowiibre diisun	aronoul an	org luntamin	nin controv	v expsup
geneng	3	0.3122 0.3735 + 0.0230 0.0060	1	0.0190	+ 77	+	+ + 0.5038 + 0.3044 0.3769 0.0000 0.0050	,		0.3644	0.3769 +	0.0	0.5360 +	0.3808		8 0.4626 0 0.0000		T	<u> </u>	+				+
expbrand	0.3122	0.0000	+	+	+		+	+		-	0.3856 +	<u> </u>	+	+		+	+	0.4335 +	+	0.0	0.3026 +	0.3298	+	0.0000
exppack	0.3735 0	0.0000	0.4956	56 0.5370 00 0.0000	70 + 00	+	0.4303	+	0.3023 +	+	+	0.0 4.0	0.4627 + 0.0010	0.4550	50 + 10	+	0.3070	0.4165 +	<u>.</u>		+	0.2864	164 0.2916 180 0.0340	
BSE	1+	+	1				+		<u> </u>		,	<u> -</u> -	<u> </u>	+	+		,		+	+	<u>, </u>	+		+
prepared	+	0.0000		0.3958	58 + 30	+	+	1		0.3332 +	+	0.0 9.0	0.4868 +	+	+	+	+	+	+	+	+	+	0.3214	0.3325 0.0150
pestiert	0.3177 +	0.5370 -	0.3958	30	+	0.2988	0.0000	+	0.3829	0.4157 + 0.0020	+	0.0 0.0	0.4110 - 0.0020	0.4827	27 -	+	0.3137 0.0210	+			0.0	0.3006 + 0.0270	0.0000	33 +
cheapsup	+	+	+	+			1+		ļ .	+		+	+	<u> </u>	+	+	+	+	0.0000	+	+	+	+	+
highsalt	+	+	+	0.2988	88.		0.5326	0.5530	0.4856	0.3199 +		<u> </u>	+	+	<u> </u>	0.0000	0.3057	<u> </u>	+	+	<u> </u>	<u> </u>	0.4378	- 01
artipres	0.5038 +	0.4303 +	+	0.4933	33 +	0.5326	L	0.3429	0.3646 +		0.2935 -	0.0	0.3847 +	0.6281	+ 00	<u> </u>	0.0000	+	<u>+</u>	+	+	+	0.3744	4 0
highfat	+	+	,	+		0.5530			0.0000	+	+	<u> </u>	<u> </u>	+	<u> </u>	0.3845	+	1	+	+	ļ		+	
infomis	+	0.3023 -	+	0.3829	- 62	0.4856	١	0.0000		0.5089 +	+	<u> </u>	+	+	+	0.0000	+	-	+	<u> </u>	<u>. </u>		0.4986	+ 98
notrepac	<u> </u>	+	0.3332		57 - 20	0.3199	1+	+	0.5089	+	0 0	0.3967 0.39 0.0030 0.00	0.3976 +	+		<u> </u>	0.0000	,	+		+		0.5511	+ 00
dilbrand	0.3769 0 0.0050 0	0.3856 + - 0.0040	+	+	+	+	0.2935	+	+		+	+	+	+	0.3854	4 0.3143 0 0.0210	+	+	+	o o	0.2926 + 0.0320	+	+	+
throwbbd	+)+	+	+				+			0.2422 0.0780	+		<u> . </u>	+	+	+	+	+	+			+	0.3082
artsweet	000000	0.4627 -	0.4868	68 0.4110 00 0.0020	10 +	+	0.3847	<u>+</u>		0.3976 +	+		0.4596	596 0.5782 010 0.0000	82 0.3243 00 0.0190	0.003564		+	•		+_	+	+	+
ımagesup	+	+	+	<u>. </u>	+	+	+	+	+	+	•	0.4 0.0	0.4596 0.0010			+		+	0.4214 + 0.0020	+	+	+	+	
artcolour	0.3808 +	0.4550 + 0.0010	+	0.4827	27 -	+	0.0000	+ +	+	+		0.5		170	+	+	0.0000	+	+	1	+	+	0.3484	+ 00
ownlabel	0.2778 -	+	+	<u>. </u>	+	,	+	-			0.3854 + 0.0040	0.3	0.3243 0.3525 0.0190 0.0100	525 + 00		+	+	+	0.0030	+	+	+		+
liminio	0.4626 +	+	+	+	+	0.4627	+	0.3845	0.5913		0.3143 + 0.0210	0.3	0.3564 + 0.0090	+-	+		0.4471		+		+	•	0.5045	+ 54
artilav	0.6366 +	i	+	0.3137	37 + 10	+	0.0000	+	##	0.0000	+	0.6245 0.0000	0.6245 0.2765 0.0000 0.0450	765 0.5742 150 0.0000	42 + 00	0.0010		+	+	+		0.3458 + 0.0100	0.2768	+
eatont		0.4335 0.4165 + 0.0010 0.0020	+	-	+	•	+			+	+	+	+	+	+		+	+		o o	0.4886	+	,	0.3163
cheapbra	+	+	+	+	0.6847	•	+	•	•	+	•	+	0.4214	214 +	0.3989	- 6. 0	+	+	+	0 0	+	0.2937	37 + 50	·
lowlibre	+			ļ	+	+	+	+	+	+	+	<u>. </u>	+	+	+	+	+	+		+	0.0	0.2856 +	+	+
disip	+	0.3026 - +		ļ. 	+	+	+	 ₊	-		0.2926 +	 	+	<u> </u>	+	<u> </u>	+	0.4886	0.3277 + 0.0180		+	0.4017	917 -	
попогтд	0.3560 +	+	+	0.3006	+ 90		+		+	+		+	+	+	+	+	0.3458	+	ı	0.0360		+	+	
unfammg	t	0.3298 0.2864 + 0.0160 0.0380	+	1+	+		+			+		+	+	+	+		+	+	0.2937 +	၁ ဝ	0.4017 + 0.0030		+	0.3244
controv	0.3105 +	0.2916	0.3214	.14 0.6603 80 0.0000	03 +	0.4378	0.3744	+	0.0000	0.5511 +	+	+	+	0.3484	900	0.5042	0.2768		+		+	+		+
dnsdxə	+	0.4775 0.5450 + 0.0000 0.0000	0.3325	25 + 50	+		•	-	+	+	0 0	0.3082 + 0.0250		+	+	+	+	0.3163 -	+			0.3244	+ 08	

Correlation between the Risk Relieving Strategy Variables within Part Four of the Questionnaire

•	similyon	freesamp	fambrand	govappro	expert	brandrep	family	cheapers	expsuper	comlabel	celeb	moneybac supimage		cheaperb	expbra	indeappr
	0.4959	+	+	0.5084	0.4674	0.4350	+_	+	0.2758	+	0.3682	0.3153	+	+	0.3291	0.2927
- 1	0.0000	-		0.0000	0.0000	0.0000			0.0430		0.0060	0.0200			0.0150	0.0320
4.	0.4959	0.3666	0.3308	0.3139	0.3231	0.2852	+	+	0.3230	+	0.2993	8998'0	+	+	0.2925	+
٥Į	0.0000	0900:0	0.0160	0.0210	0.0170	0.0370			0.0170		0.0280	0.0060			0.0320	
	0.3666	<u>~</u>	0.5902	0.5058	0.3684	0.4455	0.3946	0.5431	0.4065	0.6513	+_	0.7003	0.4221	0.4887	0.2895	+
	0.0060		0.0000	0.0000	0.0060	0.0010	0.0030	0.0000	0.0020	0.0000		0.0000	0.0010	0.0000	0.0340	,
	0.3308	8 0.5902		0.4191	0.3940	0.4767	0.5152	0.4621	0.2972	0.4902	0.2792	0.5395	0.4060	0.5291	0.3040	+
- 1	0.0160	0.0000		0.0020	0.0040	0.0000	0.0000	0.0000	0.0310	0.0000	0.0430	0.0000	0.0030	0.0000	0.0270	
5.5	0.5084 0.3139	9 0.5058	0.4191		0.7040	0.4459	0.4689	0.3651	0.5269	+	0.5407	0.5153	0.5502	+	0.5040	+
Õ	0.0000 0.0210	0.0000	0.0020		0.0000	0.0010	0.0000	0.0070	0.0000		0.0000	0.000	0.0000		0.0000	
<u>4</u> .	0.4674 0.3231	0.3684	0.3940	0.7040		0.4016	0.3512	0.4511	0.5022	+	0.5387	0.4610	0.5400	0.3297	0.4422	0.5302
0.0	0.0000 0.0170	0.0060	0.0040	0.0000		0.0030	0.0090	0.0010	0.0000		0.0000	0.0000	0.0000	0.0150	0.0010	0.0000
0.4350	350 0.2852	2 0.4455	0,4767	0.4459	0.4016		0.5181	0.2944	9806.0	0.3377	0.4104	0.3623	0.5691	0.2827	0.3612	0.4237
0.0	0.0010 0.0370	0.0010	0.0000	0.0010	0.0030		0.0000	0.0310	0.0230	0.0130	0.0020	0.0070	0.0000	0.0380	0.0070	0.0010
	+	0.3946	0.5152	0.4689	0.3512	0.5181		0.5451	0.3954	+	0.5006	0.6015	0.4569	0.3867		+
		0.0030	0.0000	0.0000	0.0090	0.0000		0.0000	0.0030		0.0000	0.0000	0.0010	0.0040	0.0400	
	_+	0.5431	0.4621	0.3651	0.4511	0.2944	0.5451		0.5146	0.4000	0.4256	0.7737	0.4092	0.8610	0.2854	+
		0,0000	0.0000	0.0070	0.0010	0.0310	0.0000		0.0000	0.0030	0.0010	0.0000	0.0020	0.000	0.0360	
0.2	0.2758 0.3230	0.4065	0.2972	0.5269	0.5022	0.3086	0.3954	0.5146		+	0.4502	0.3914	4143	0.3798	0.8828	+
0.0430	430 0.0170	0.0020	0.0310	0.0000	0.0000	0.0230	0:0030	0.0000			0.0010	0.0030	0.0020	0.0050	0.0000	
	+	0.6513	0.4902	+	+	0.3377	+	0.4000	+		+	0.4609	0.3692	0.4122	+	0.3488
		0,0000	0.0000			0.0130		0.0030				0.0000	0.0060	0.0020		0.0100
0.3	0.3682 0.2993	+	0.2792	0.5407	0.5387	0.4104	0.5006	0.4256	0.4502	+		0.4711	0.4623	0.4187	0.4552	+
0.0	0.0060 0.0280	6	0.0430	+	0.0000	0.0020	0.0000	0,0010	0.0010			0.0000	0.0000	0.0020	0.0010	
0.3	0.3153 0.3669		0.5395	0.5153	0.4610	0.3623	0.6015	0.7737	0.3914	0.4609	0.4711		0.4707	0.6350	+	
00	0.0200 0.0060	0.0000	0.0000	0.0000	0.0000	0.0007	0.0000	0.0000	0.0030	0.0000	0.0000		0.0000	0.0000		
	+	0.4221	0.4060	0.5502	0.5400	0.5691	0.4569	0.4092	0.4143	0.3692	0.4623	0.4707		0.3405	0.3614	0.3992
		0.0010	0.0030	0.0000	0.0000	0.0000	0.0010	0.0020	0.0020	0.0000	0.0000	0.0000		0.0120	0.0070	0.0030
	+	0.4887	0.5291	+	0.3297	0.2827	0.3867	0.8610	0.3798	0.4122	0.4187	0.6350	0.3405		0.3106	+
		0.0000	0.0000		0.0150	0.0380	0.0040	0.0000	0.0050	0.0020	0.0020	0.0000	0.0120		0.0220	
0.3	0.3291 0.2925	5 0.2895	0.3040	0.5040	0.4422	0.3612	0.2798	0.2854	0.8828	+	0.4552	+	0.3614	0.3106		+
0.0	0.0150 0.0320	0 0.0340	0.0270	0.0000	0.0010	0.0070	0.0400	0.0360	0.0000		0.0010		0.0070	0.0220		
0.2	0.2927 +	+	+_	+	0.5302	0.4237	+_	+	+	0.3488	+_	+_	0.3992	+_	+	
0.0	0.0320		,		0.0000	0.0010				0.0100			0.0030			

Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Food Risk Perception Variables (Part Three of the Questionnaire)

_		 	1						Г	Г						г						Г	r		
	geneng e.	expbrand exppack	ack BSE	epared	pestfert	cheapsup highsalt		artipires h	highfat ir	infomis no	notrepac difbrand	$\overline{}$	throwbbd arts	artsweet imag	imagesup artcolor	의	abel liminfo	artilav	catout	cheapbra	lowfibre di	difsup no	nonorg unfam	unfaming controv	expsup
secure		0.0370	•	+	+	•	1	+	<u>+ · </u>		+	+	•	•	•	-0.3	0.0190		•	,	<u></u>	+	+	<u> </u>	+
discover	0.3565	+	+	.+	0.3202	+	+	+	+	-	0.3825 -	"	0.2882 0	0.3761 -	0.5	0.3195	+	0.4223	+ +	+	+		0.3280 +	0.2847	+
	0.0080		_		0.0180						0.0040	٥	0.0350 0	0900:0	0.0	0.0190		0.0020	0				0.0150	0.0370	
pest		7.0-	-0.2945 -				0.2865	+	0.3205	+	+		+	0	0.2885 -	'	+	+		+	+	·			-0.3545
		0.0	0.0320				0.0360		0.0180	1	1		-	Ö	0.0360	-									0.0090
nochange	+		0.3056 +	+_	+_	0.3018	<u> </u>	_ <u>'</u>			-0.3573 +	+_	<u></u>	oʻ	-0.3591	+_		<u>.</u>	+_	+	+	1_	0.3464	- 2	0.3752
		ŏ	0.0260			0.0270					0.0080	1		Ö	0.0000	-							0.0100	00	0.0060
world	•	<u>+</u>	· ·	0.4025	+	-0.3392			+	+		-0.3481 +	+		+	+	+_	•	_+_		•		<u>,</u>	+	+
			\dashv	0.0030		0.0120						0.0100													
unconven	+	+	+	+	+	i	1	+	+	<u>+</u>		+	+	+	+	+			0.4498	+	+		•	+	+
winning		-0.3080	-0 4629	-0.3171	-0.3234	+					+	-		+		1		<u> </u>		0.2020	4			0.3767	0.4778
o			0.0000	0.0190						<u> </u>	·			-		-	<u> </u>	i	.	0.0350	_	<u> </u>	•	0.0160	
setways	Ţ	_+	+_		+	1			+		0.3364 +	+	-	Q	-0.3188	-	<u> </u>	_	+		+	'	+		l
				_							0.0130			0.	0.0200								_		0.0120
trad	+		-0.2832	2	+	0.3097	+	+	+			0.3083	+	+	_+_	+	+	<u>+</u>	-0.2922	+	+	<u>.</u>	0.2993	93 -	+
			0.0380	01		0.0230					-	0.0230				_			0.0320				0.0280	80	
ahead	•	· · · · ·	-0.4515	-0.4555		+	+		+		0.2915 +	<u>ٻ</u>	-0.2773	o <u> </u>	0.3201	•				0.4339	+		+	·	-0.4547
).O	0.0010	0.0010	0.0220				1		0.0320	1	0.0420	0	0.0190	-				0.0010					0.0010
voiceops	+ _	+	•	+	+	1	+	+	+	0.3126	0.3952 +		0.2956 +	+	+	+	+	+	+	+	0.3410 +		0.3696 +	+	
holdlife	+	0.3201 0.3	0.3676 +	+	+	_+		+	,	1 _	1	0 3438 +	4		-			<u> </u>			4				
			0.0070						·			0.0110	<u> </u>	•		-	1	<u>-</u>	F	<u> </u>			<u> </u>	•	-
status	-0.3622	, ,	-0.4192	ı	-0.3909		+		0.3062 +	+	+	,		4	Ö.	-0.3075		-0.3085	5 -0.2763	+	+			,	-0.3560
	0.0070	ŏ	0.0020		0.0030				0.0240	1			-	 	ő	0.0240	$\frac{1}{1}$	0.0230	0 0.0430						0.0090
fewrules	+	+	•	+	+		+	+	<u>+</u>	+		-0.3471 -0	-0.2743 +	+	·'O	0.4028 +	+	+	+_	+		+_	+	0.4113	
		-	,								-	0.0100	0.0450		0.0	0.0030				Ţ				0.0020	
appear	+	.0	-0.3506	-0.3726		+_	+	+	- '			0.3179	<u>'</u>	+		+_	+	+_	-0.3400	+	+	+_	+		
),()	0.0100	0.0060								0.0190		_					0.0120			1			
alternat		+	0.4358 -	0.4047		•	0.3136	0.3571	+	0.4550	0.6406 -	+		0.4356 +	ŏ	0.4360	+	0.3781	+		+	•	<u>.</u>	0.6008	+
	0.0080	0.0	0.0010	0.0020	0.0000		0.0210	0.0080		0.0010	0.0000	1		0.0010	ő	0.0010	-	0.0050	0					0.0000	
mysucces		·0·	-0.4775 -0.4080	80 -0.3048		+		-0.3947	+	<u>'</u>	+	+	-:-	<u> </u>	φ 	-0.4191	_+		-0.3729	+	+	_1_		_•	
		ïÖ	0.0000 0.0020	20 0.0250				0.0030							Ö	0.0020		-	0.0050				1	1	
contacts	<u> </u>	<u>+</u>	0.3265	55 +	+			+	-0.2689		-0.3573		•	<u> </u>	-0.2830 +		-0.3519	-61	+		<u></u>	+	+		+
			09100	30					0.0490	0.0240	0.0080			0	0.0400		0.0090	060					4	4	

Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Risk Relieving Strategy Variables (Part Four of the Questionnaire)

moneybac supimage cheaperb expbra indeappr	0.2693 + + + +	0.0490	0.2997		+ 0.3513 0.2711 0.4654 +	0.0090 0.0470 0.0000	+ + -0.2891 -0.3459	0.0340 0.0100	0.2697	0.0490	-0.3216 - + + - +	1,01000	0.0010	+	+		0.3121 + 0.3540 - 0.5540 +	0.00220 0.0090 0.0000	+ + + +	0.0410	0.3456	0.2825 + 0.2750 + 0.3619 +	0.0440	+	0.2939 + 0.2835 - +	0.0310 0.0380	-0.4518 - + -0.3193 +	0.0010	- 0.3843 - + +	
ej celeb	+		.!		+		-0.2854 +	0.0360	,			0.0140		+	+		0	0	0.3523 -0	0.0090	-0.2826 -		-		ľ	0	<u>م</u>	0	+_	_
er comlabel	•		+	1	0.4548 +	0.0010		0	+			0.3611	0.0070	<u>, </u>			0.4968 +	0.0000	0	0.0000	9 9	0.3736 -	0.0050	+	,		+		•	
expsuper	0.3084 -	0.0230	!		<i>-</i>	o .	<u> </u>				<u>-</u>	-		<u>,</u>	<u> </u>		0	0	+	0				+	+				+	_
cheapers	0.3	ö	<u>.</u>	-	+			_			<u></u>		<u>. </u>	<u> </u>	<u> </u> -	·	+		•			+		+			+			-
family	+	_	!		+ 68	30			53 -	30		71 7	. 08	+_	<u> </u>		39 +	30			•	17+	30	+	+		<u>,</u>			-
brandrep	+				0.3939	0.0030				0:0030	1	0.3571	0.0080		+		4 0.3939	0.0030	+			4 0.4017			+		7 -	0	7 +	_
expert	÷		1		0.2883	0.0350			-0.3150	0.0200	ì	1		+			0.3474	0.0100			•	0.3254		•	+		-0.4157		0.2697	
govappro			-0.3164	0.0200	0.4218	0.0010						7872 0	0900.0	+	+		0.3989	0.0030			+	0.4344	0.0010	,	+		-0.5030	0.0000	0.2834	_
fambrand g	+_										1				+							+		+		-	-0.2755	0.0460		
freesamp fa	+		•		+	į	+						•	+			+												Ť,	
similyou fr	+		1		+		•		+		+	+	-	-	+		0.3214 +	0.0180				+		+	-		-0.3287	0.0150	0.3105 +	
advert sin	,				0.3162	0.0200			-0.3317	0.0140	_'_	7 3763 0	0.0000	İ	0.2718 +	0.0470	0.4809	0.0000	•			0.3511	0.0000	+	+	· <u></u>	-0.4172	0.0020	0.3027	
ac	secure +		discover -		pest		nochange +	, !	world		nnconven -		gimin.	setways +	frad		ahead		voiceops +	1	+ holdlife +	status		fevrules -	annear +		alternat		mysucces	

APPENDIX 3

CORRELATION MATRIX

(Results of the Total Homogeneous Sample)

- 3.1 Correlation between the Social Value Variables within the Screening Instrument (Part Two of the Questionnaire)
- 3.2 Correlation between the Food Risk Perception Variables within Part Three of the Questionnaire
- 3.3 Correlation between the Risk Relieving Strategy Variables within Part Four of the Questionnaire
- 3.4 Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Food Risk Perception Variables (Part Three of the Questionnaire)
- Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Risk Relieving Strategy Variables (Part Four of the Questionnaire)

Coefficients / significance levels are only detailed when <0.05.

Correlation between the Social Value Variables within the Screening Instrument (Part Two of the Questionnaire)

winning setways trad ahead voiceops holdlife status fewrules appear alternat mysucces conta	5+ 0.2671 0.1756 0.2202 + - +	0.000.0	+ + 0.1345	0.0230	0.2106 0.4895 + + 0.4404 +	0.0000 0.0020 0.0000 0.0000	-0.1585 0.4985 0.3787 -0.1863 -0.1823 0.3448 -0.1435 -	0.0200 0.0000 0.0000 0.0060 0.0070 0.0000 0.0350	+	00000	+ -0.2211 + 0.3142 - 0.3752 -0.1734 0.3308 + +	0.0010 0.0000 0.0000 0.0000	+ + 0.6916 0.1498- 0.4229+ 0.2529- 0.2250-	0,000 0	+ 0.1772 - 0.2451 + +	00000 000000 000000 000000	11 + 0.1772 + -0.1362 + + -0.1362 + +	0.0000 0.0000 0.0000	. + + 0.3539 0.2673	0000'0 0000'0 0000'0 0000'0	0.14980.1362 + + + + + + + 0.1501 +	0.0280	+ + + + + + + + + + + + + + + + + + +	0.4229 + + 0.4783 + + 0.2518 -	000000 0000000 00000000	+	09000	0.2329 0.2633 0.2134	0.0000 0.00000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.00000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.00000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000	0.0000 0.00000 0.0280	0.2250 + + 0.2518 +	0.0010 0.0000 0.0000	
	+	-	0.2225 +	0.0010	0.5090	0.0000	-0.1585	0.0200	0.2728 -	0.0000	+		+		+			0.0010	0.6916				- 0.34	+	0.0000	0.3752 +		0.000			0.2250 +	0.0010	
nochange world un	3 -0.2157	0.0000					+		+		-0.1929 0.2728	0.0000 0.0000	+ -0.1585 - +	00 0.0200	0.4985 -	0.000	06 0.3787 +	0.0000 02	·		-0.1823 +	0.0070	0.3448 -	Ĺ	00 0.0350				0.0250	00000			
e discover best			+		+		0.2783 -	0.0000	-0.2157	0.0010	-0.2635 0.2225 -	0.0000 0.0010	0605:0 +	0.0000	0.2671 -	0.0000	0.1756 - 0.2106	0.0100 0.0020	+ 0.4895	0.0000	0.1545 +	0.0230	0.2202 - +	+ 0.4404	0.0000	_+		0.3207	0.0000	00000	0.1882 +	0.0060	
secure	secure	_	discover +		pest +		nochange 0	-	world -0	9	unconven -0	0	winning +		setways 0	ָם	trad 0	3	ahead -		voiceops -		holdlife 0	status +		fewrales -	_	appear +	alternat		mysucces +		

Correlation between the Food Risk Perception Variables within Part Three of the Questionnaire

															÷											
	geneng e	expbran exppack BSE	pack BSE	ргерат	red postic	rt chea	prepared postlert cheapsulhighsall artiprred highlat inforus notrepad d	alt artıprı	es highla	informs	notrepa	difbrand	throwbb		1,5		=	6	artilav eatout	1 7	cheapbrilowiibrd difsup	brodifsup			ırı contro	expsup
geneng	7	#	0.1473	40 +	0.5039)39 + 000	0.1512	12 0.3993 80 0.0000	93 +	0.0000	0.3285	0.0020	+	0.0000	0.13624	0.0000 0		0.00000	0.4944 +		0.1400 + 0.0440	+.	0.2343	+	0.2828	+
expbrand		0 0	0.6046 +	0.3489	+	+	,	+	+	0.1365	+	+	0.1401 -	+		ر ک	-0.1418 + 0.0390	+	0 0	0.3245 - 0.0000	+	+	ı	0.1919 0.0050	- 61	0.0000
exppack		0.6046	<u> </u>	0.5869	69 0.1775 00 0.0090	+ 577	+	+	+	0.1637	0.1613	0.1578	<u> </u>	+	+	-	+	0 0	0.1353 0. 0.0490 0.	0.2753 -	,	+	+	0.2157	+ 120	0.0000
BSE	0.1473+			+	0.2110	110 +	+	0.1828	28 + 30	+	ļ	1.	+	0.1390 -		0.1594 +	-		.	+	0.1716	716 +	+	+	0.1386	+
prepared		0.3489 0	0.0000		1	+ 06/	+	+	+	0.1385	0.2569	+	0.2317	0.2018 +	+	+	+	+	0 0	0.2595 -	,	+	+	0.1911	1	0.3507
pestlert	0.5039	1	0.0090 0.0020	0.0000	1	+	0.3089	089 0.4857	57 0.1971 00 0.0040	0.0000		+	0.1536	0.0000		0.5092 +		0.3264 0	0.0000	+	0.0	0.1567	0.2095]+		1+
cheapsup	 	+	<u>+</u>	+	+		+	+	1+	1.	<u> </u>	0.2033]+	.1	0.3307		0.3463 +	<u>]+</u>	+	0.0	0.0000	0.2942	<u> +</u>	0.1575	+	
highsalt	0.1512	+	+	+	0.3089	+ 680 000		0.5143	9		1	1	+	 +	<u> </u>	<u> </u>	<u> </u>	0.2826 0	0.2525 +	+		+	+	+		
artipres	0.3993	+	0.1828	\$28 + 80	0.4857	857 + 300	0.5143		0.3104	0.3843	0.2671	+	+	0.3792 +		0.5709 +	00	0.3206 0	0.5624 + 0.0000	0.0	0.1672 0.2 0.0150 0.00	0.2441 + 0.0000	0.2620	520 0.1598 000 0.0200	98 0.3315 00000 00000	+
highlat	<u> </u>	+	+	+	0.1971	971 + 340	0.5251	251 0.3104 000 0.0000	l	0.0000	0.1350	+	+		+	 	-	0.2453 +	-	+	<u> </u>	0.3639 -			+	
infomis	0.2649	0.1365 0	0.1637 + 0.0170	0.1385		518 - 300	0.4377		43 0.4530 00 0.0000	000	0.3210	+	+	0.1475 -		0.2839 -		L	0.2708 +	+	0.0	0.2517 -			0.2584	+
notrepac	0.3285	,	0.1613 -	0.2569	69 0.4370 000 0.0000	·	0.0050 0.0430 0.0050 0.0430	<u> </u>	<u> </u>	0.3210 0.0000		+	0.2776	0.2393 +		0.2115		<u> </u>	0.4033	<u> </u>	<u> -</u>	0.0080	310 0.1668 380 0.0150	568 - (50	0.0000	+
dilbrand	0.2148	-	0.1578 -	+	+		, ,	+	+	+	<u>+</u>		0.2065	0.0090	0.1927 +	i	0.3621 0	+	+	9 8	0.3125 +	0.3313	+	l	+	+
throwbbd	+	0.1401 + 0.0420	+	0.2317		0.1536 + 0.0250	+	+	+	+	0.2776			+	+	+		0.1913 +	0 0	0.2251 -	<u>.</u> 69	0.1441 + 0.0350	+	0.1422	22 0.1997 80 0.0030	0.1671
artsweet	0.0000	+	0.1390	390 0.2018 440 0.0030		0.3671 +	0.2234	234 0.3792 0.0000	92 -	0.1475	<u> + </u>	0.0000	+		0.2575	0.0000	0.1567 0 0.0220 0	0.3166 0 0.0000 0	0.4356 +	o	0.1455 + 0.0350	+	0.2150	+	0.2443	+
ımagesup	0.1362 +	+		+	+	0.0	0.3307 +	+			+	0.1927	+	0.2575		0.0000	0.2939 +		+		0.3509 +	0.1354	+	0.1444	44 + 09	
artcolour	0.0000	+	0.0	0.1594 + 0.0210	0.5092		0.1517 0.3479 0.0280 0.0000	479 0.5709 0000.0 000	+ 00 60	0.2839	9 0.2115 0 0.0020	 +	+	0.5152	0.2592	+	٦	0.2944 0 0.0000 0	0.00000	0.1363 0.7 0.0480 0.0	0.2229 0.2 0.0010 0.0	0.2061 +	0.2258	0.2258 + 0.0010	0.3702	- 0
ownlabel	0.1428	-0.1418 - 0.0390	+		+	0.0	0.3463 -	+	١,			1	+	0.1567	0.2939 +		- 	+	0 0	0.1399 0.4 0.0410 0.0	0.4298 +	0.2577	+ <i>LL</i> +	0.1382	+	
liminto	0.3278	+		+	0.3 0.0	0.3264 +	0.2826		06 0.2453 00 0.0000		4 0.3061 0 0.0000	0.1515	0.1913	0.3166 +	1	0.2944			0.4093 +	+	0.0	0.2280 - 0.0010	+			+
artilav	0.0000	<u></u>	0.1353 - 0.0490	+	L	0.3819 +	.25.25	0.5602	4 00 00 +	0.2708		4	+	0.0000	0.1460	0.0000		0.4093	+	+	0.0	0.1935 + 0.0050		0.2740 + 0.0000	0.2536	+ 8
eatont	+	0.3245 0	0.2753 -	0.2595	+ 95 +	+	+	+		+	,	+	0.2251	+		0.1363 (0.1399 + 0.0410	+		+	<u>. </u>	0.0060	- 688	4	+	0.2894
cheapbra	0.1400		+	<u> </u>	+	9:0	0.6741 +	1	72 + 50	+		0.3125	١٠	0.1455	0.3509	1	0.4298 +	+	+		0.0	0.1783 · 0.4307 0.0090 0.0000	307 + 300	0.2220	+	-0.1888
lowlibre	<u> </u>		0.0	0.1716 - 0.0130	+	+	0.2574	574 0.2441 000 0.0000	41 0.3639 00 0.0000	39 0.2517 30 0.0000	+ 0	+	0.1441	+		0.2061 +		0.2280 0.0010	0.1935	o 0	0.1783	+	+	+	0.0060	2 -
dısıp	-	+	+	+	ļ	l	0.2942 +	+	<u> </u>	ļ	0.1810	0.0000	+	+	0.1354 +	<u>.</u>	0.2577 -	+		0.1889 0.4 0.0060 0.0	0.4307 +		+	0.0000	+	,
попоп	0.2343	+	+	+	0.0	0.2095 + 0.0020	+	0.2620	- 00		0.1668	+	+	0.2150 +		0.2258 +	+		0.2740 -	+	+	+_		0.2063	63 0.2587 20 0.0000	•
unfamıng	+	0.1919 0	0.2157 + 0.0020	0.1911	+		0.1572 + 0.0220	0.1598	. 00			0.2267	+	+	0.1444 +	+	<u>. </u>	+	+	0.0	0.2220 +	0.4528		0.2063 0.0020	+	0.2035
controv	0.2828	+		0.1386 0.2043 0.0440 0.0030	, ,	0.4919 +	0.2797		115 + .	0.2584	0.0000	3 +	0.1997	0.2443 +		0.2702 +		0.3086 0	0.2536 +	+		0.1872 +	0.2 0.0	0.2587 + 0.0000		<u>. </u>
dusdxa	+	0.5856	0.4675 + 0.0000	0.0000	\$07 + 000			+	<u>, </u>	+	+	+	0.1671	+	-		<u>+</u>	+	+	o o	0900'0	<u> </u>	•	0.2035	35 - 30	

Correlation between the Risk Relieving Strategy Variables within Part Four of the Questionnaire

advert	similyou	n freesamp		fambrand g	govappro	expert	brandrep	family	cheapers	expsuper	comfabel	celcb	moneybac	supimage	cheaperb	expbra	indeappr
	0.50	0.5069	0.3508	0.3770	7	0.3563		0.4096			0.1644				0.2029	364	0.2015
	0.00	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0010	0.0000	0.0170	0.0000	0.0000	0.0000	0.0030	0.0000	0.0030
0	0.5069		0.3170	0.2208	0.3599	0.2517	0.2738	0.4249	0.2591	0.3227	+	0.4155	5 0.2957			0.3106	0.1569
0.0	0.000.0		0.0000	0.0010	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000		0.0000	0.0000	0.0010	0.0020	0.0000	0.0220
0	0.3508 0.31	0.3170	_	0.4621	0.2499	0.1753	0.3476	0.3312	0.3347	7 0.2843	0.3314	0.2332	0.4670			0.3274	
0.0	0.0000	0.0000		0.0000	0.0000	0.0100	0.0000	0.0000	0.0000	00000	0.0000	0.0010	0.0000			0.0000	
fambrrand 0.3	0.3770 0.22	0.2208 0	0.4621		0.3739	0.2767	0.5085	0.4477	0.3098	3 0.3593		0.2296			<u> </u>	0.3643	0.2024
0.0	0.0000	0.0010	0.000.0		0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000					0.0000	0.0030
govapprov 0.4	0.4897 0.33	0.3599 0	0.2499	0.3739		0,6253	9098:0	0.3435	0.3207	7 0,3650	0.1581					0.2964	0.2452
0.0	0.0000	0.0000	0.0000	0.0000		0.0000	0.0000	0.0000	0.0000	0.0000	0.0210	0.0000	0.0000	0.0000	0.0030	0.0000	0.0000
	0.3563 0.25	0.2517 0	0.1753	0.3767	0.6253	•	0.3134	0.2161	0.2208	8 0.2825	+	0.3041	0.1499	0.2513	+	0.2087	0.4035
).O	0.000	0.0000	0.0100	0.0000	0.0000		0.0000	0.0010	0.0010	0.0000		0.0000	0.0290	0.0000		0.0020	0.0000
brandrep 0.4	0.4739 0.27	0.2738 0	0.3348	0.5085	0.3606	0.3134		0.4380	0.2231	0.3508	0.3207	0.2705	0.2602	0.4835	0.1938	0.3741	0.3571
ö	0.0000	0.0000	0.000.0	0.0000	0.0000	0.0000		0.000	0.0010	0.0000	0.0000	0.0000	0.0000	0.0000	0.0040	0.0000	0.0000
,ò	0.4096 0.42	0.4249 0	0.3312	0.4477	0.3435	0.2161	0.4380		0.4987	7 0.4347	+	0.4652	0.4400	0.4058	0.0373	0.2309	0.1603
9	0.0000	0.0000	0.000.0	0.0000	0.0000	0.0010	0.0000		0.0000	0.0000		0.0000	0.0000	0.0000	0.0000	0.0000	0.0190
cheapers 0.2		0.2591 0	0.3347	0.3098	0.3207	0.2208	0.2231	0.4987		0.5304	0.2164	0.4110	0.4370	03030	0.6968	0.2426	+
اة اة	0.0010 0.00	0.000	0.000.0	0.000	0.0000	0.0010	0.0010	0.0000		0.0000	0.0020	0.0000	0.0000	0.0000	0.0000	0.0000	
expsuper 0.3	0.3485 0.3227		0.2843	0.3593	0.3650	0.2825	0.3508	0.4347	0.5304		0.2151	0.3410	0.2749	0.4947	0.3375	0.7597	0.1603
_	0.0000	0.0000	0.000.0	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000		0.0020	0,0000	0.0000	0.0000	0.0000	0.0000	0.0190
comlabel 0.	0.1644 +	_	0.3314	0.3277	0.1581	+	0.3207	_+_	0.2164	1 0.2151		+	0,3174	0.3010	0.2344	0.1941	0.3920
9	0.0170		0.000.0	0.0000	0.0210		0.0000		0.0020	0.0020			0.0000	0.0000	0.0010	0.0050	0.0000
Ġ		0.4155 0	0.2332	0.2296	0.3942	0.3041	0.2705	0.4652	0.4110	0.3410	+		0.4525	0.3852	0.4273	0.3797	+
ö	0.0000	0.0000	0.0010	0.0010	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000			0.0000	0.0000	0.0000	0.0000	
moneybac 0.2	0.2958 0.25	0.2977 0	0.4670	0.3890	0.2865	0.1499	0.2602	0.4400	0.4370	0.2749	0.3174	0.4525		0.4542	0.5215	0.3402	0.2080
ö	0.0000	0.000.0	0.0000	0.0000	0.0000	0.0290	0.0000	0.0000	0,0000	0.0000	0.0000	0.0000	,	0.0000	0.0000	0.0000	0.0020
supimage 0.3			0.3159	0.4783	0.3678	. 0.2513	0.4835	0.4058	0.3090	0.4947	0.3010	0.3852	0.4542		0.3110	0.5553	0.2763
ö	0.0000 0.00	0.0010	0.0000	0.0000	0.000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000		0.0000	0.0000	0.0000
cheaperb 0.2	0.2029 0.20	0.2099 0	0.4061	0.3111	0.2007	+	0.1938	0.3734	0.6968	8 0.3375	0.2344	0.4273	0.5215	0.3110		0.3939	+
0.0	0:0030	0.0020	0.0000	0.0000	0.0030		0.0040	0.0000	0.0000	0.0000	0.0010	0.0000	0.0000	0.0000		0.0000	
0	0.3364 0.31	0.3106 0	0.3274	0.3643	0.2964	0.2087	0.3741	0.2909	0.2426	5 0.7597	0.1941	0.3797	0.3402	0.5553	0.3939		0.1800
ŏ		0.0000	0.0000	0.0000	0.0000	0.0020	0.0000	0.0000	0.0000	0.0000	0.0050	0.0000	0.0000	0.0000	0.0000		0.0080
indeapprov 0.2		0.1569 +		0.2024	0.2452	0.4035	0.3571	0.1603	+	0.1603	0.3920	<u>+</u>	0.2080	0.2763	+	0.1800	
_	0000	_						_	_	_		_	!				

•

Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Food Risk Perception Variables (Part Three of the Questionnaire)

	geneng expbr	geneng expbran exppack BSE	preparedp	prepared pestfert cheapsu highsalt artiprres highfat infomis notrepad difbrand throwbb artsweet imagesu artcolor ownlabd liminfo	psu highsa	lt artiprres	highfat in	omis notre	pad difbrand	throwbb ar	tsweet ima	gesu artcolor	ownlabe	iminfo a	artflav eatout	_	cheapbralowfibre difsup		nonorg un	unfamin controv expsup	rov exps
secure	+	+	+		+	ı	+		+	+					+		·	+		+	+
discover	+	0.1395 +	+	0.1608 -		+	+	0.2515	515	+	<u> .</u>	+	+	 	0.1520 +	ļ.	+		0.1763 +	0	0.1659 +
		0.0420		0,0190				0.0000	000						0.0260				0.0100	0.	0.0150
pest	<u></u>			-0.1563 +	+	+	0.1478 +	•	+		0.1	0.1692		+	1	0.2046	46 +	0.1460	+	Ġ.	-0.1398 -0.2263
			0.0460	0.0220			0.0310			_).0	0.0130			1	0.0030	30	0.0330		0	0.0410 0.0010
nochange	-0.1415 +	0.1463	+	<u>+</u> _	_ 1		<u>.</u>	-0.2655	+ 559	<u>·</u>	٩̈́	-0.1364 -0.1649	<u>`</u>		-0.2107	<u>.</u>	-0.1379	+	0	0.2486 -0.	-0.1684 0.2040
	0.0400	0.0330			-			0.0000	80		0.0	0.0470 0.0160		1	0.0020		0.0440		١	0.0000	0.0140 0.0030
world	+_	0.1902 -	0.1936 +		+		+	+	,	+		+		-	+	-0.1435	35 -	-0.1380	•		0.1727
		0.0050	0.0050								-					0.0370	70	0.0440	-		0.0120
unconven	+ +	0.1731	0.2053 +	+	_•	+	+	0.1;	0.1552 +	0.1832 +	_+_	+	+	+		0.2612 +		+	+	+	0.1500
		0.0110	0.0030					0.0230	230	0.0070			·		0	0.000.0		,			0.0290
winning	<u> </u>	-0.1796	-0.1505	+			+ +		+		0.1	0.1716	+			0.1901	+ 10	+	+	•	-0.1602
		0.0000	0.0280								0.	0.0120				0.0050	50				0.0190
setways	- 0.2014	114 0.1562 -	+	+		-0.1399		-0.2212	212 0.2032	<u> </u>			+	<u>·</u>	+	+	-0.1683	+		0.2385	0.2210
	0.0030	30 0.0230				0.0410		0.0010	- 1								0.0140			0.0000	0.0010
trad	+	. +	+	+	0.1454 +	+	+ +	•	0.2070	_+_	_+_	•		+	,	+_		+		0.1608 -0.	-0.1776 +
				0.0	0.0340			-	0.0020											0.0190	0.0090
ahead	<u>.</u>	-0.1790	-0.1673 -	···	0.2232 +	•	+	<u>.</u>	0.1783	<u>+</u>	.o.	0.0189 +	+	+ +	-!-	0.2390	+ 06	0.1453	+	<u> </u>	-0.1703
		0.0000	0.0150	- - - -	0.0010			1	0.0090		ŏ	0.0060			+	0.0000		0.0340		1	0.0130
voiceops	+	+	+	<u>. </u>	•	+	+	0.2	0.2705 +	+	+_	+	+	+	+	+	0.1487	+	+	o	0.1752 +
			1					0.0	0.0000	1	+	+		\dagger	+	+	0.0300		+	Ö	0.0100
holdlife	+	+	+	+	+	,	+	•	0.1514	+ -		+	+	+	+	•	1	+	+	+	+
status	-0.1547			-0.1905 +	,	,	+	,	+		+	_,	+	+	-0.1380 -	+_	+_	+		<u>. </u>	-0.1394
3	0.020.0	, 1383		0.000	<u> </u>	1.	-	<u> </u>			<u> </u>	1		+	0.0440	+	+			+	0.0420
lewinies	+	0.0440	+	·	+	+	+	+		+	+	+	+	+ +	•	<u>.l.</u>	<u>.</u>	1	+	+	+
appear	1	-0.1508	<u> </u>	+	0.1557		+	-0.2259	259 +		+			+	9	-0.1647 +	ļ	+	-0.1687 +	Ó.	-0.2559 -
		0.0280			0.0230	30		0.0	0.0010						0	0.0160			0.0130	· o	0.0000
alternat	+ 0.2067	+	+	<u>.</u>	<u>.</u>	0.1683		0.1501 0.3	0.3700 -	+	0.1744	0.1879	-	0.2029	0.2537	•		,	0.2002	oʻ	0.3615 +
	0.0030					0.0140		0.0280 0.00	0.0000		0.0110	0.0060		0.0030	0.0000				0.0030	O)	0.000.0
mysucces	<u>.</u>	-0.1387	<u>, </u>		-0.1359			+	+	+	•	-0.1394		+ +		<u></u>	<u>+</u>		<u></u>	+	
		0.0430			0.0470	70 0.0160						0.0430		1	+	-			+	-	+
contacts	·	+	·	•	<u>. t</u>	,	<u>.</u>	-0.1	-0.1616 -		•	•		0.0170	0.0290	1	•	+	+	ợ ơ	-0.1702 -
												-		-1							

Correlation between the Social Value Variables (Part Two of the Questionnaire) and the Risk Relieving Strategy Variables (Part Four of the Questionnaire)

	advert	similvon	Greesamn	Sambrand	oovannro	pansa	brandren	family	cheaners	Pynciiner	comlahel	heleh	moneyhac	sunimage	cheanerb	exnbra	indeappr
secure	16360	+		0.15680					980	+	8	+	Τg	0.22060	+		
	0.01700			0.02200					0.04100		0.02900		0.02000				
discover	,	+	+				+	,		+	. +	+	+		+	+	
best	0.15850		+	+	0.16900	+	0.15460	+	+	0.22010		+	0.14810	0.22570	+	0.24710 +	
_	0.02100				0.01300		0.02400			0.00100			0.03100			0.00000	
nochange	0.16520	0.17590	+	+	+	1	+	0.15890	+			+	+	+	+		
	0.01000						0.0							3,00			
world	1		+			•	0.00400		+	<u>. </u>	+		1	0.00800	+	1	
unconven	,		4		4	+	-	,	+		+	+			+		
winning	0.19600	+	+	+	+	+_	0.20560	.+	+	0.23700	+	0.17420 +	+	0.23360	+	0.28320	
	0.00400						0.00300			0.00000		0.01100		0.00100		0.00000	
setways	0.22370		+	0.18540		+	+	0.15780	+	+		0.27450	+	0.14500		+	
	0.00100	0.00200		0.00700	0.03200			0.02100				0.0000		0.03400	0.04900		
trad	+	+	+	+	+		+	+	4	+	•	+	+	+	+		
ahead	0.21590	+	+	0.19960		+.	0.21960		+	+ 0.25690 +	+	0.19000	+	0.20490	+	0.27480	
	0.00200			0.00300	0.04300		0.00100	0.03600		0.00000		0.00500		0.00300		0.0000	
voiceops	,	-				1	1	-0.15590	-	_	+	•			,	+	
holdlife	+	+	1	+		4.	+	+	+	+			+		+		i
status	+	+	+		+ _	+	+	+	+	+		+	+	+	-	0.13830	
fewrules.	+	+	0.19990	+	,			+	+	+	+	+	+	ı	0.15290	+	
			0.00300												0.02500		
арреаг	+	+	0.12420		+	+	0.15190		+	0.14050		0.18630	+	0.25550	+	+	
1	0 24440		0.07000	0.00400			0070.0	0.01000		0.04100	0.18770			-0.16600		-0.14500+	
dilcanar ,	0.0000	<u> </u>	0.02700				· -	0.00400		0.02700				0.01500		0.03400	
mysucces	+	0.01000	+		+	+	+	+	+	+	+	0.22110 +	+	+	-	+	
contacts		+			+					-0.20550					-0.17430		-0.20560
										0.00300					0.01100	0.02000	0.00300

Interview One: Interview Guide / Schedule

FAMILY

Can you tell me about your family, and what the family unit symbolises to you.

How important is your family to you.

To what degree do you rely on your family.

To what degree do they influence your decisions.

What does the family provide for you.

Are you concerned about what other people think of your family.

How important is it to you that other people think favourably about your family.

How important are friends in comparison to your family.

WORK AND LEISURE

Can you tell me about your work and leisure time and how they relate to each other.

How does your work time and leisure time relate, are they very separate things in the your life or do they have common features.

Is your leisure time and you work time of equal importance to you.

How much leisure time to you have, is it enough.

How do you spend your leisure time.

What kinds activities are you involved.

ATTITUDES TOWARDS AUTHORITY (THOUGH AN EXAMINATION OF HEALTH

Can you tell me about the type of health care you use.

What kind of health care do you use, where do you turn when you are unwell.

How do you feel about going to the doctors.

How much confidence and trust do you have in doctors.

Would you consider private health care. Why

What kind of treatment/medication do you feel is best.

Would you consider alternative treatment or would you rather stick to more conventional treatment and medication.

What are your views on alternative medicine.

PREFERRED ORGANISATIONAL STRUCTURE

Can you describe to me the type of organisation you prefer to work in.

What are the most important features in an organisation that allow you to work effectively and happily.

Do you like to work were things are very structured were rules and procedures are set clearly.

How do you think communication should occur.

Do you like to be managed or left to get on with things.

Do you think it is important to have managers on several different level in an organisation.

If an opportunity came up for promotion with an increase in salary, but there was a possibility that you may not enjoy the job as much as the one you are currently doing. Would you take the promotion with the belief that the extra money is compensatory.

What motivates you within this organisation.

HOW COULD YOU IMPROVE YOUR LIFESTYLE?

WHAT KINDS OF THINGS MAKE YOU FEEL GOOD ABOUT YOURSELF?

Social Value Score Profiles for the Total Homogeneous Sample of 215

Social Value Score Profiles for Homogeneous Sample of 215

ID	Sust	Inner	Outer	ID	Sust	Inner	Outer	ID	Sust	Inner	Outer
	S/c	S/c	S/c		S/c	S/c	S/c	İ	S/c	S/c	S/c
1	-3	2	5	73	-3	3	1	145 .	-6	-1	-8
Outer											
2	-2	-1	0	74	-5	-2	-7	146	2	-2	2
3	0	1	4	75	-2	2	6	147	-6	2	-3
Outer				Outer							
4	-5	0	-3	76	-5	3	-8	148	-1	4	-2
5	-5	-1	0	77	-3	-1	-5	149	4	0	7
6	-4	4	2	78	2	-1	-2	150	5	-2	-5
								Sust			
7	3	4	-8	79	-4	-1	-6	151	-5	4	-3
8	-3	-1	-3	80	2	-1	2	152	-2	0	-10
9	-4	3	-4	81	-7	10	-7	153	-2	7	-6
				Inner				Inner			<u> </u>
10	0	3	6	82	-3	1	3	154	4	-1	11
Outer								Outer			
11	-2	3	-1	83	-7	7	-3	155	3	-4	-1
				Inner							
12	0	1	2	84	-8	5	-1	156	0	-1	2
13	-4	4	5	85	-5	5	-2	157	-2	0	3
				Inner							
14	-2	2	-3	86	-3	2	4	158	-4	7	-4
ļ						1	ľ	Inner			
15	-3	4	-1	87	-2	2	0	159	2	1	-2
16	-3	7	6	88	-1	3	-3	160	-3	5	0
17	-2	1	-2	89	-3	-2	-10	161	3	2	-2
18	-3	4	4	90	-3	5	0	162	-6	6	-4
								Inner			
19	7	-1	0	91	-3	1	-4	163	-4	9	-4
Sust								Inner			
20	-7	1	-3	92	-2	4	-3	164	5	-1	-1
								Sust			
21	0	4	-7	93	-3	3	-2	165	-4	0	-6
22	-4	-2	-6	94 .	3	0	-1	166	-1	6	-9
					}			Inner]	
23	1	5	2	95	1	0	7	167	-1	2	ı
		 		Outer							
24	-4	6	0	96	-2	3	-4	168	-7	4	-4
1		ľ	l Į		L	L	I	l	L	<u> </u>	1

Inner	T		1	1	T	-			1	T	
25	2	-1	5	97	3	3	1	169	-6	-2	-3
Outer					ļ						
26	-4	5	1	98	2	-1	1	170	-4	9	-5
								Inner			
27	-2	2	-6	9 9	5	-3	-4	171	-2	0	-2
			·	Sust							
28	1	1	6	100	6	5	3	172	-3	5	-6
Outer											
29	-1	-2	2	101	-1	0	-6	173	-2	4	-3
30	-3	2	-1	102	1	1	2	174	0	-2	-2
31	0	0	0	103	3	4	2	175	6	-4	2
				•				Sust			
32	-4	5	6	104	-4	-2	0	176	-2	-3	-4
33	4	5	-4	105	3	1	-7	177	2	0	-2
34	3	-3	5	106	-4	-4	-8	178	-4	-1	2
35	-4	6	-6	107	-1	2	3	179	-2	0	0
Inner			-								
36	-3	0	-6	108	-4	4	3	180	2	-6	-4
37	-1	4	-5	109	-2	-4	-4	181	-9	4	5
38	-2	-3	-4	110	-1	0	1.	182	-1	-1	-1
39	-2	-1	-2	111	-3	3	-9	183	0	1	-4
40	-4	4	-2	112	-1	0	-4	184	1	-1	5
								Outer			
41	5	2	-1	113	-7	-1	-1	185	2	1	2
Sust						ŀ					
42	-3	9	5	114	2	3	-2	186	-3	6	-2
43	8	-7	-5	115	-2	1	3	187	4	0	-2
Sust					`			Sust			
44	1	-2	1	116	-3	2	0	188	3	-2	4
45	-1	2	1	117	0	1	1.	189	3	4	2
46	-4	9	-4	118	-2	3	7	190	0	4	-4
Inner		Ì		Outer							
47	-6	9-	8	119	-7	-3	1	191	-6	5	1
48	-6	0	1	120	1	-2	-3	192	-4	4	-4
49	-3	1	-5	121	-2	6	6	193	0	6	-2
								Inner			
50	-6	2	8	122	2	0	-1	194	-8	2	-3
Outer											
51	1	3	0 ·	123	-7	3	1	195	-1	1	-2

<u> </u>	1 2	T 4	10	104			10	100	10	1 1	2
52	-2	4	2	124	-2	-1	2	196	0	-1	
53	-2	5	3	125	0	3	-8	197	5	-6	-4
		1						Sust			
54	7	3	-2	126	0 .	3	-4	198	-3	8	-6
Sust								Inner			
55	-1	4	-4	127	-1	-2	-3	199	-2	2	0
56	-2	1	2	128	-1	-1	4	200	-2	8	-7
					!			Inner			
57	2	-2	-3	129	-7	7	-7	201	5	1	-2
				Inner				Sust	ļ		
58	4	-1	-2	130	-2	-1	-2	202	6	2	1
Sust								Sust			
59	-4	5	-6	131	2	1	-8	203	5	-1	-1
			ļ					Sust			
60	-2	0	-2	132	-4	-5	1	204	0	-1	-3
61	-4	1	0	133	-4	5	-7	205	-6	0	9
								Outer			
62	-3	-3	6	134	-2	1	4	206	4	0	-2
Outer				Outer			Ì	Sust			
63	-2	9	-7	135	-2	1	6	207	-3	2	5
Inner				Outer				Outer			
64	-3	2	-1	136	2	1	-8	208	-3	-1	-3
65	-2	2	-2	137	6	9	1	209	7	3	-2
								Sust	į.		
66	-3	-4	-3	138	4	-5	1	210	0	-4	0
				Sust							
67	4	4	-6	139	-6	10	6	211	-4	1	7
								Outer			
68	-5	6	-4	140	-1	3	3	212	7	-1	0
Inner								Sust			
69	-1	3	-1	141	6	-3	6	2 13	-7	1	-3
70	1	0	-4	142	0	2	1	214	0	0	0
71	0	-1	1	143	-2	-1	0	215	-1	3	0
72	-6	4	5	144	-2	3	6				
				Outer							
	<u> </u>	<u> </u>		L	<u> </u>	<u> </u>	ш			I	

Outer	Outer Directed
Sust	Sustenance Driven
Inner	Inner Directed

Tables to Illustrate the Relationships between Social Value Variables and Food Risk Perception Variables for the Total Homogeneous Sample of 215

Table 4.26 Factor 1 Technology (man made additions to food stuffs) Risks

	Social	Genetic	Artificial	Artificial	Artificial	Artificial
	Value	Engineering	Preservative	Sweeteners	colours	Flavours
	Group					
Contacts	S		· · · · · · · · · · · · · · · · · · ·			_
Setways	S					
Nochange	S	- "				
Status	О	-			·	_
Mysucess	О		-			
Discover	I					+
Alternat	I	+	+	+	+	+

 Table 4.27
 Factor 2 High Cost as a Risk

	Social	Expensive	Expensive	Paying more	Eating out	Expensive
	Value	Brand	Packaging	for Pre		Store
	Group			prepared food		
Nochange	S		+			+
Setways	S	+	+			+
Appear	0		-			
Status	0					_
Best	0	·				-
Winning	0			_		
Mysuccess	0		_			
Ahead	0					_
Discover	I		+			
World	I		+	+		+
Fewrules	I		+			
Unconve	I		+	+	+	+

 Table 4.28
 Factor 3
 Nutritional Risks

	Social	High Salt	High Fat	*Misleading
	Value			Information
	Group			
Best	О		+	
Mysuccess	0	_		
Appear	0.	+		
Alternat	I			+

Table 4.29 Factor 4 Cheap (Image Associated) Risks

	Social Value Group	Cheap Store	Image of Store	Own label	Cheap Brand
Tradition	S	+		·	
Nochange	S				
Best	0		+		+
Ahead	0	+	+		+
Winning	0		+		+
World	I				_

Table 4.30 Factor 5 Ecological (conservationist / environmental) Risks

	Social	Pesticides /	Non / recycle	Controversial
	Value	Fertilisers	Packaging	operations
	Group			
Contacts	S		_	_
Setways	S		_	
Trad	S			_
Nochange	S		· —	•
Best	0	-		
Appear	0		_	. —
Status	0	_		
Voiceops	I		+	+
Discover	I	+	+	+
Unconve	I		+	
Alternat	I		+	+

 Table 4.31
 Factor 6/7
 Unfamiliarity as Risk

	Social	Different	Non Organic	Unfamiliar	Different
İ	Value	Super		ingredients	Brand
·	Group				
Holdlife	S				+-
Trad	S			+ .	+
Setways	S			+	+
Best	S	+			
Nochange	S			+	
Ahead	0	+			+
Appear	0		_	·	
World	I	_			•
Altern	I		+ .		
Discover	I		+-		

Factor 8 BSE Risk

Positive correlation were identified between BSE and both Inner directed value variable 'discover' and between Sustenance driven value variables 'holdlife' and 'contacts' however these correlations were not significant.

Tables to Illustrate the Relationships between Social Value Variables and Risk Relieving Strategy Variables for the Total Homogeneous Sample of 215

 Table 4.39
 Factor 1 Societal Acceptance

	Social Value Group	Advertised	Someone similar to you approves	Family approves	Celebrity approves
Secure	S	+			
Setways	S	+	+	+	+
Nochange	S	+	+	+	
Best	0	+			
Winning	0	+			
Status	0				+
Ahead	0	+		+	+
Mysuccess	0	-	+		
Appear	0			+	+
Voiceops	I				
Alternat	I			-	·

Table 4.40 Factor 2 Low Cost

	Social	Cheaper Store	Money back	Cheaper
	Value	Value		Brand
	Group			
Setways	S			+
Secure	S	+	+	
Fewrules	S			
Best	0		+	+

Table 4.41 Factor 3 Branding

	Social Value Group	Free Sample	Familiar brand	Brand reputation	*Comprehens ive labelling
Setways	S		+		
Secure	S		+		_
Best	0			+	
Winning	0			+	
Ahead	0			+	
Appear	0	+	+	+	
Alternat	I	-	_		+
World	I	·		_	
Fewrules	I	+			

Table 4.42 Factor 4 High Cost (Image)

	Social Value Group	Expensive Store	Store Image	Expensive Brand
Secure	S		+	
Setways	S		+	
Contacts	S	_		_
Status	0			+
Ahead	0	+	+	+
Winning	0	+	+	+
Best	0	+	+	+
Appear	0	+	+	
World	I		_	
Alternat	I		_	_

Table 4.43 Factor 5 Authoritarian Reassurance

	Social	Government	Independent	Expert
	Value	Approval	Approval	Approval
	Group			
Contacts	S		_	
Setways	S	+		
Best	0	+		
Ahead	О	+		

Sample for Qualitative Interviews

Interviewee	Sample	Id	Interview	Interview	Interview Three
		Code	One	Two	Food Risk Perception
			Social Values	Food Risk	and Risk Relieving
·				Perception	Strategies
DP	Pilot	S1	*	*	
JW	Pilot	S2	*	*	
LB	Pilot	S3	*	*	
JM	Main	S10			*
DC	Main	S11	·	1	*
CJ	Main	S12			*
SW	Main	S13			*
RJ	Main	S14			*
MM	Main	S15			*
LP	Pilot	11	*	*	
AT	Pilot	12	*	*	
CW	Pilot	13	*	*	
LK	Main	I10 .	-		*
JF	Main	I11			*
HP	Main	I12			*
VM	Main	I13			*
VS	Pilot	01	*	*	
ND	Pilot	O2	*	*	
SH	Pilot	O3	*	*	
CS	Main	O10			*
AR	Main	O11			*
SF	Main	O12			*
AK	Main	O13			*
ТВ	Pilot	ND1	*		
SM	Pilot	ND2	*		
RB	Pilot	ND3	*		
VD	Pilot	ND4	*		
PB	Pilot	ND5	*		
DM	Pilot	ND6	*		
JН	Pilot	ND7	*		_
LD	Pilot	ND8	* '		
LC	Pilot	MD9	*		
PH	Pilot	ND10	*		
TOTAL			19	9	14

^{*} identified that the interviewee was interviewed at this stage

Qualitative Interview Transcripts: Selection of Quotes

Interview One

Question area 1 - Family; Question area 2 - Work and Leisure; Question area 3 - Attitudes toward Authority; Question area 4 - Organisational Structures; Question area 5 - Improve Lifestyle; Question area 6 - Feel Good

Interviewee	Sample	Id	Quest	Quest	Quest	Quest	Quest	Quest
		Code	area	area	area	area	area	area
			1	2	3	4	5	6
DP	Pilot	S1	0	*	*	*	*	*
JW	Pilot	S2	*	*	*	*		
LB	Pilot	S3	*	*	*	*	*	*
JM	Main	S10						!
DC	Main	S11						
CJ	Main	S12				-		
SW	Main	S13	-					
RJ	Main	S14						
MM	Main	S15						
LP	Pilot	I1	*	*	*	*		*
AT	Pilot	12	*	*	*	*	*	*
CW	Pilot	13	*	*	*	*	*	*
LK	Main	I10						-
JF	Main	I11						÷.
HP	Main	I12						
VM	Main	I13		,				
VS	Pilot	Ol	*	*	*	*	*	*
ND	Pilot	O2	*	*	*	*		
SH	Pilot	O3	*	*	*	*		
CS	Main	O10						
AR	Main	011						
SF	Main	O12						
AK	Main	O13						

^{*} indicates that a response was made in this area and as such included in the analysis

Interview Two

Question area 1 - Technology; Question area 2 - High Cost; Question area 3 - Nutrition; Question area 4 - Misleading Information; Question area 5 - Cheap / Poor Image; Question area 6 - Ecological / ethical; Question area 7 - Unfamiliar; Question area 8 - Non Organic

Interviewee	Sample	Id	Quest	Quest	Quest	Quest	Quest	Quest	Quest	Quest
		Code	area	area	area	area	area	area	area	area
			1	2	3	4	5	6	7	8
JW	Pilot	S2	*	*	*	* ,	*			*
JM	Main	S10	*	*			*	* .		
DC	Main	S11	*	*	*		*	*	*	*
CJ	Main	S12		*	*	*	*	*	*	*
SW	Main	S13	*	*			*	*	*	*
RJ	Main	S14	*	*	*	*	*	*	*	
MM	Main	S15	*	*			*	*	*	*
AT	Pilot	I2	*	*	*	*		*		
CW	Pilot	13	*	·	*	*		*	*	*
LK	Main	I10	*			*	*	*	*	
JF	Main	I11	*			*	*	*	*	*
HP	Main	I12	*	*		*	*	*	*	*
VM	Main	I13	*		*		*	*	*	*
VS	Pilot	01	*				*	*	*	
CS	Main	O10		*						
AR	Main	O11	*	*	*		*	*	*	*
SF	Main	O12		*	*			*	*	*
AK	Main	O13	*	*		*	*	*		*

^{*} indicates that a response was made in this area and as such included in the analysis

Interview Three

Question area 1 - Societal acceptance / endorsement; Question area 2 - Low Cost; Question area 3 -Branding; Question area 4 - High Cost; Question area 5 - Authoritarian reassurance

Interviewee	Sample	Id	Quest	Quest	Quest	Quest	Quest
		Code	area	area	area	area	area
			1	2	3	4	5
JM	Main	S10		*	*		*
DC	Main	S11		*	*		
CJ	Main	S12	*		*		*
sw	Main	S13	*	*	*	_	*
RJ	Main	S14	*	*	*		*
MM	Main	S15	*		*		*
LK	Main	I10	*		*		*
JF	Main	I11	*	*	*		*
HP	Main	I12					
VM	Main	I13			*		*
CS	Main	O10	*		*	*	*
AR	Main	O11	*		*	*	*
SF	Main	O12	*			*	*
AK	Main	O13	*		*	*	*

^{*} indicates that a response was made in this area and as such included in the analysis